Published for SISSA by 🖉 Springer

RECEIVED: October 10, 2013 REVISED: January 23, 2014 ACCEPTED: February 18, 2014 PUBLISHED: April 2, 2014

Chiral four-dimensional F-theory compactifications with SU(5) and multiple U(1)-factors

Mirjam Cvetič,^{*a,b*} Antonella Grassi,^{*c*} Denis Klevers^{*a*} and Hernan Piragua^{*a*}

^aDepartment of Physics and Astronomy, University of Pennsylvania, Philadelphia, PA 19104-6396, U.S.A.

^cDepartment of Mathematics, University of Pennsylvania,

Philadelphia, PA 19104-6396, U.S.A.

E-mail: cvetic@cvetic.hep.upenn.edu, grassi@math.upenn.edu, klevers@sas.upenn.edu, hpiragua@sas.upenn.edu

ABSTRACT: We develop geometric techniques to determine the spectrum and the chiral indices of matter multiplets for four-dimensional F-theory compactifications on elliptic Calabi-Yau fourfolds with rank two Mordell-Weil group. The general elliptic fiber is the Calabi-Yau onefold in dP_2 . We classify its resolved elliptic fibrations over a general base B. The study of singularities of these fibrations leads to explicit matter representations, that we determine both for $U(1) \times U(1)$ and $SU(5) \times U(1) \times U(1)$ constructions. We determine for the first time certain matter curves and surfaces using techniques involving prime ideals. The vertical cohomology ring of these fourfolds is calculated for both cases and general formulas for the Euler numbers are derived. Explicit calculations are presented for a specific base $B = \mathbb{P}^3$. We determine the general G_4 -flux that belongs to $H_V^{(2,2)}$ of the resolved Calabi-Yau fourfolds. As a by-product, we derive for the first time all conditions on G_4 -flux in general F-theory compactifications with a non-holomorphic zero section. These conditions have to be formulated after a circle reduction in terms of Chern-Simons terms on the 3D Coulomb branch and invoke M-theory/F-theory duality. New Chern-Simons terms are generated by Kaluza-Klein states of the circle compactification. We explicitly perform the relevant field theory computations, that yield non-vanishing results precisely for fourfolds with a non-holomorphic zero section. Taking into account the new Chern-Simons terms, all 4D matter chiralities are determined via 3D M-theory/F-theory duality. We independently check these chiralities using the subset of matter surfaces we determined. The presented techniques are general and do not rely on toric data.

KEYWORDS: Flux compactifications, F-Theory

ARXIV EPRINT: 1306.3987





^bCenter for Applied Mathematics and Theoretical Physics, University of Maribor, Maribor, Slovenia

Contents

1	Intr	oducti	on and summary of results	1
2	The	e ellipti	c curve in dP_2 and its fibrations	6
	2.1	The el	liptic curve with rank two Mordell-Weil group	6
	2.2	Genera	al Calabi-Yau fibrations with dP_2 -elliptic fiber	8
3	Cal	abi-Ya	u fourfolds with rank two Mordell-Weil	13
	3.1	Singul	arities of the fibration: matter Surfaces & Yukawa points	14
		3.1.1	Matter: codimension two	14
		3.1.2	Yukawa couplings: codimension three	20
	3.2	The co	bhomology ring and the Chern classes of \hat{X}	20
		3.2.1	Second Chern class and Euler number of \hat{X} : general formulas	22
		3.2.2	The full cohomology ring of \hat{X} : base $B = \mathbb{P}^3$	24
4	G_4 -	flux co	onditions in F-theory from CS-terms: Kaluza-Klein states or	n
	\mathbf{the}	3D Co	oulomb branch	26
	4.1	A brie	f portrait of G_4 -flux in M-theory	27
	4.2	Derivit	ng conditions on G_4 -flux in F-theory	28
		4.2.1	F-theory conditions from KK-states corrected CS-terms	28
		4.2.2	KK-corrected 3D CS-terms: field theory computations	30
5	G_4 -	flux &	chiralities on fourfolds with two U(1)s	33
	5.1	G_4 -flux	x on fourfolds with two rational sections	34
	5.2	4D chi	ralities from matter surfaces & 3D CS-terms	37
		5.2.1	General strategy to determine 4D chiralities	37
		5.2.2	Chiralities on \hat{X} with $B = \mathbb{P}^3$: matter surfaces & CS-terms	38
	5.3	4D and	omaly cancellation: F-theory with multiple $U(1)s$	41
		5.3.1	4D anomaly cancellation: general discussion	41
		5.3.2	Anomaly cancellation in 4D F-theory with a U(1) ² -sector: $B = \mathbb{P}^3$	42
	5.4	A torio	c example	43
6	Cal	abi-Ya	u fourfolds with $\mathrm{SU}(5)\! imes\!\mathrm{U}(1) imes\mathrm{U}(1)$	44
	6.1	Singul	arities of the fibration: gauge group, matter & Yukawa Couplings	45
		6.1.1	Explicit resolution of a codimension one $SU(5)$ -singularity	45
		6.1.2	Matter: codimension two	50
		6.1.3	Yukawa couplings: codimension three	55
	6.2	The co	bhomology ring and the Chern classes of $\hat{X}_{\mathrm{SU}(5)}$	57
		6.2.1	Second Chern classes and Euler number of $X_{SU(5)}$: general formulas	58

7	G_4 -flux & chiralities on fourfolds with $SU(5) \times U(1)^2$ 7.1 G_4 -flux on fourfolds with two rational sections & $SU(5)$ 7.24D chiralities from matter surfaces & 3D CS-terms7.2.1Anomaly cancellation7.3A toric example	60 60 61 64 65				
8	Conclusions and future directions	67				
A	Chern classes of dP_2 -elliptic fibrations	69				
в	Intersection ring on \hat{X} with $B = \mathbb{P}^3$ 73					
\mathbf{C}	Total Chern class of $\widehat{dP}^B_2(\mathcal{S}_7,\mathcal{S}_9)$	76				
D	Intersection ring for $\hat{X}_{\mathrm{SU}(5)}$ with $B = \mathbb{P}^3$ 77					
\mathbf{E}	Basis of $H^{(2,2)}_V(\hat{X}_{\mathrm{SU}(5)})$ with $B=\mathbb{P}^3$	79				
F	CS-levels & chiralities for $\hat{X}_{SU(5)}$ with $B = \mathbb{P}^3$ F.1 Chern-Simons levels F.2 Chiralities	80 80 85				
G	Toric tuning of \mathcal{S}_7 and \mathcal{S}_9 for $B = \mathbb{P}^3$	86				

1 Introduction and summary of results

Four-dimensional F-theory compactifications provide a broad domain of the string theory landscape with the potential to derive promising particle physics consequences of string theory. It has a number of advantages, such as encoding geometric structures of string theory compactifications at a finite string couplings. The main focus in recent years has been on studies of F-theory GUT models with SU(5) as well as SO(10) gauge groups, initiated in [1–4] within local model building. Techniques for constructions of global models were developed in [5–9], and efforts to embed local models into global ones have been pursued, see e.g. [10–12] for reviews.

The origin of non-Abelian gauge symmetries in four-dimensional F-theory compactifications is well understood since the origins of F-theory [13–15], and it is due to the full classification of codimension one singularities of elliptically fibered Calabi-Yau fourfolds with a section in the Weierstrass or Tate model [16–18]. Matter multiplets appear at codimension two singularities and were studied originally in [13, 14, 19] and more recently in [9–12].¹ Constructions with chiral matter in four-dimensions require the addition of

¹Recent efforts clarified and filled the gaps in classifications singularities at higher co-dimensions [20-22]. For most recent complementary advances, employing deformations of singularities, see [23].

 G_4 -flux, which belongs to the middle (vertical) cohomology on the resolved Calabi-Yau fourfold. Specific fluxes of this type were constructed in [21, 24].

On the other hand Abelian gauge symmetries in F-theory are less understood and studied. This is primarily due to the fact that the classification of Abelian gauge symmetries and their matter spectrum depends crucially on the global geometry of the elliptic fibration. Nevertheless, some aspects of Abelian gauge symmetries have been studied in local F-theory models employing spectral cover methods [6, 25–31]. The study of higher rank Abelian sectors in four-dimensional F-theory compactifications is also of phenomenological interest, since it can play an important rôle in model building beyond the Standard Model model.

Abelian gauge theory sectors appear in compactifications of F-theory on elliptic Calabi-Yau varieties with a general fiber being an elliptic curve with rational points. Elliptic curves with a so-called non-trivial Mordell-Weil group of rational points are a classical subject in mathematics [32–36]. These rational points lift to rational sections of the elliptically fibered Calabi-Yau manifold, which contribute new harmonic two-forms to the cohomology that support Abelian gauge fields in the F- theory effective action. The number of Abelian gauge fields is set by the rank of the Mordell-Weil group of the elliptic curve and its torsion subgroup gives rise to non-simply connected groups [15, 37, 38]. Rank one Mordell-Weil groups in compact elliptic fibrations have been studied recently in the F-theory literature in a variety of contexts [39–44, 44–47]. Elliptic fibrations with elliptic fiber of D_5 -type have been constructed in the context of counting BPS states in [48]. The resolutions of their most general fibrations, which have a Mordell-Weil group of rank three, as well as the complete induced F-theory matter spectrum have been analyzed in [49].

While a classification of possible Abelian gauge sectors in F-theory, analogous to the well-studied non-Abelian sector, is lacking (see, however [50] for a systematic study of rational sections on toric K3-surfaces), significant progress has been made recently for the systematic study of U(1) × U(1) gauge symmetry on elliptically fibered Calabi-Yau manifolds with a rank two Mordell-Weil group [46, 51].² The analysis found that the natural presentation of an elliptic curve with two rational points and a zero point is the generic Calabi-Yau one-fold in dP_2 and the birational map to its Tate and Weierstrass form was derived [51]. While the discussion of its resolved elliptic fibrations was done for a Calabi-Yau variety over a general base B, their classification was first performed for the base $B = \mathbb{P}^2$ and later for any two-dimensional base B in [54]. One key finding of this classification, which we further elaborate on also in this work, was the identification of all topological degrees of freedom in the construction of a fibrations of a fixed elliptic curve

²The elliptic curves in $\mathbb{P}^2(1, 1, 2)$ and \mathbb{P}^2 have also appeared in [52] and independently in [53], cf. also [48], as F-theory duals of heterotic backgrounds with U(1)-Wilson lines. In the former, elliptic threefolds with these elliptic fibers over \mathbb{F}_n are constructed and the non-Abelian gauge groups are matched between the dual theories in dependence on n as in [15]. In the latter, the Kaehler classes of these elliptic threefolds, which are dual to the Wilson lines breaking $E_8 \to E_{8-k} \times \mathrm{U}(1)^k$, k = 0, 1, 2, in the heterotic string, are identified by matching certain Gromov-Witten invariants with BPS-states of non-critical strings from E_{8-k} small instantons computed on the heterotic side. However, the detailed analysis and classification of the Abelian sectors on the F-theory side, by studying the singularities of these elliptic fibrations, has first been carried out in [51] and in this work using the resolved dP_2 -elliptic fibrations.

over a fixed base *B*. These are encoded in the choice of two divisor classes S_7 , S_9 [51]. Allowing for all their possible values yields elliptic fibrations with novel properties, most notably with a non-holomorphic zero section, that were not constructed before because the existence of a holomorphic zero section was enforced. Next, a thorough analysis of the generic codimension two singularities of this most general family of elliptic Calabi-Yau threefolds was given. This determines geometrically all the matter representations under U(1) × U(1) and their multiplicities, that were shown to be consistent with anomaly cancellations in the six-dimensional compactified theory. Explicit toric examples were constructed, both with U(1) × U(1) and SU(5) × U(1) × U(1) gauge symmetries.

It is the purpose of this paper to develop explicit techniques for the calculation of the matter spectrum and their chiralities for four-dimensional F-theory compactifications with two U(1)-factors. This involves now F-theory compactifications on elliptically fibered Calabi-Yau fourfolds with rank two Mordell-Weil groups. Furthermore the appearance of the chiral matter requires an explicit construction of G_4 -flux. We would like to emphasize that our calculations are performed for general globally defined elliptic Calabi-Yau fourfolds and are *not* restricted to geometries described by toric reflexive polytopes. In particular, we find closed formulas for the basis of the vertical cohomology groups, the consistent G_4 -flux and the chiral indices for entire discrete families of Calabi-Yau fourfolds which depend explicitly on the aforementioned degrees of freedom S_7 , S_9 of these families. We underline that even if each member of this family could be realized torically, it would be hard to obtain this dependence on \mathcal{S}_7 , \mathcal{S}_9 by considering reflexive polytopes. In recent works [45, 47, 55] a classification of toric Calabi-Yau fourfolds with SU(5) and allowed U(1) factors was given (see also [46]). Note however that techniques employed have, at least to the knowledge of the authors, not culminated in the determination of the matter representations and their 4D chiralities yet which are the main results of this paper.

We advance the program in several important ways:

- The general elliptic fiber is the generic Calabi-Yau one-fold in dP_2 with two rational points and a zero point (as analyzed in detail in [51]). The chosen fiber determines the representations of the matter multiplets under U(1) × U(1) over any base B. The spectrum for U(1) × U(1) and SU(5)× U(1) × U(1) (for a specific SU(5)) is derived and summarized in table 1. However, the analysis is performed now over a general three-dimensional base and the matter appears over codimension two Riemann surfaces in the base, denoted as matter curves. We employ explicit algebraic geometry techniques using prime ideals³ to represent these matter curves and classify the singularities of the elliptic fibration over them using the Calabi-Yau fourfold with the resolved elliptic fibration (see sections 2, 3.1 and 6.1). At present these techniques allow us, however, to determine only a subset of matter surfaces (three out of six in the U(1) × U(1) and six out of twelve in the SU(5) × U(1) × U(1) case).
- As a preparation for the construction of G_4 -flux on these Calabi-Yau fourfolds, we find for the first time consistent conditions on the G_4 -flux in F-theory compactifications

 $^{^{3}}$ Note that in the case of Calabi-Yau threefolds these geometric techniques were sufficient to determine multiplicities of all the matter multiplets. [51].

with a non-holomorphic zero section. These conditions have to be formulated in three dimensions after the compactification on a circle and require the use of M-theory/F-theory duality. We show the connections between holomorphicity of the zero section and Kaluza Klein-states via Chern-Simons (CS) terms on the 3D Coulomb branch. In particular, there are new CS-terms for a non-holomorphic zero section.

• We develop techniques to calculate explicitly the most general G_4 -flux that belongs to the vertical cohomology $H_V^{(2,2)}$ resolved Calabi-Yau fourfolds (see section 3.2 and 4). To this end we algebraically calculate the full vertical cohomology ring of these Calabi-Yau fourfolds. These cohomology calculations allow us to compute the general expression for the Euler number and the Chern classes of these fourfolds for an arbitrary base B, both in the U(1)×U(1) and SU(5)×U(1)×U(1) cases. As an application of these techniques we derive an explicit basis of the cohomology group for all elliptically fibered Calabi-Yau fourfolds with fiber in dP_2 and base $B = \mathbb{P}^3$.⁴ Again these techniques are general and not restricted to toric examples. In particular the dependence on the divisors S_7 , S_9 is manifest. When the G_4 -flux is integrated over matter surfaces (determined via the geometric techniques mentioned above) we obtain the chiralities of three matter representations (second set in table 1). Chiralities of the remaining matter representations are determined by a subset of the 3D CS-terms of the dual M-theory invoking M-/F-theory duality. 4D anomalies are found to be cancelled. We note that the rest of the 3D CS-terms, in particular those for the Kaluza-Klein vector, provide an independent check for chiralities of matter multiplets obtained via geometric techniques. It is important to note that, given the list of representations that are realized, all CS-terms taken together are sufficient to determine chiralities of all the matter multiplets. Our geometric techniques allow us to have an independent determination for a subset of them (see section 5). We also perform an independent check that with the obtained spectrum the four-dimensional anomalies are cancelled. Explicit results are presented for the most general G_4 -flux for all generic resolved elliptic Calabi-Yau fourfolds over the base $B = \mathbb{P}^3$, both for $U(1) \times U(1)$ and $SU(5) \times U(1) \times U(1)$ (for a specific embedding of SU(5)).

We note that the fibrations over the chosen base $B = \mathbb{P}^3$ are generally non-flat at a single codimension three locus. This can be circumvented in two ways. Either, one can forbid the existence of the non-flat fiber geometrically, or restrict the allowed G_4 -flux by requiring a vanishing integral of it over the non-flat fiber. Both approaches yield an anomaly-free 4D spectrum with no chiral excess of additional light states.

The paper is organized in the following way: in section 2 we summarize the geometry of the general elliptic curve with rank two Mordell Weil group in dP_2 and classify its fibrations. Section 3.1 is devoted to the analysis of the codimension two and three singularities of the fibrations specifying, respectively, the matter content, the matter curves and surfaces as well as the Yukawa couplings. In section 3.2 we determine the cohomology ring and Chern classes for resolved elliptic Calabi-Yau fourfolds with rank two Mordell-Weil group.

⁴These techniques have been used in the context of mirror symmetry on Calabi-Yau fourfolds in [56–58].

$U(1) \times U(1)$	$SU(5) \times U(1) \times U(1)$
(1,0) (0,1) (1,1)	$(5, -\frac{2}{5}, 0) \ (5, \frac{3}{5}, 0) \ (5, -\frac{2}{5}, -1)$
(-1,1) (0,2) (-1,-2)	$(5, -\frac{2}{5}, 1) \ (5, \frac{3}{5}, 1) \ (\overline{10}, -\frac{1}{5}, 0)$

Table 1. Matter representation for F-theory compactifications with rank- two Mordell Weil group. While $U(1) \times U(1)$ charges for SU(5) singlets are general, $U(1) \times U(1)$ charges for the additional non-singlet matter representations of SU(5) depend on a specific realization of the SU(5) gauge symmetry.

Section 4 is devoted to a detailed discussion of G_4 -flux and conditions imposed on them in general F-theory compactifications with a non-holomorphic zero section. This study is based on relations between 3D Chern-Simons terms under F-theory/M-theory duality. A special emphasis is on quantum loop corrections due to Kaluza-Klein states, that are computed explicitly. In section 5 the general G_4 -flux is explicitly calculated (section 5.1), and matter chiralities both geometrically and via 3D Chern-Simons terms of dual the M-theory are derived (section 5.2). In section 5.3 anomaly cancellation of the four-dimensional field theory is checked and in section 5.4 an explicit toric example is presented. A generalization to F-theory compactifications with an additional SU(5) non-Abelian gauge symmetry is spelled out in sections 6 and 7. In section 6.1 the singularities at codimension one, two (matter representations) and three (Yukawa points) are presented. A non-flat fiber at codimension three is discovered. The cohomology ring and the general Euler number are calculated in 6.2. In section 7 the G_4 -flux and the 4D chiralities are computed and anomaly cancellation is checked in detail for all elliptic fibrations with dP_2 -fiber over \mathbb{P}^3 . Section 7.1 demonstrates the general construction of G_4 -flux in F-theory compactifications with rank two Mordell-Weil group and a resolved SU(5)-singularity with $B = \mathbf{P}^3$. In section 7.2 the 4D chiralities are evaluated and anomaly cancellation is checked. We finish the section with one concrete toric example in section 7.3. Conclusions and future directions can be found in section 8.

This work has seven appendices. Appendix A contains formulas for all Chern classes of dP_2 -elliptic fibrations over an arbitrary base B along with the Chern classes and Euler numbers for their resolved Calabi-Yau two-, three- and fourfolds. In appendix B we present the cohomology ring of the generic fourfolds with $B = \mathbb{P}^3$ and dP_2 -elliptic fiber. Appendix C contains the Chern classes of dP_2 -fibrations with resolved SU(5)-singularities, in appendices D and E the intersections and vertical cohomology ring of fourfolds with $B = \mathbb{P}^3$ are computed. Appendix F contains all 3D CS-terms and chiralities. Appendix G concludes by describing how to systematically construct the toric polytopes for the example of Calabi-Yau fourfolds with base $B = \mathbb{P}^3$.

2 The elliptic curve in dP_2 and its fibrations

In this section we review the construction of the elliptic curve \mathcal{E} in dP_2 and its Calabi-Yau elliptic fibrations over a general B. These Calabi-Yau manifolds have a rank two Mordell-Weil group, that gives rise to U(1)×U(1) gauge symmetry in F-theory.

In section 2.1 we review the geometry of \mathcal{E} as the generic Calabi-Yau onefold in the del Pezzo surface dP_2 following the conventions and notations of [51], to which we also refer for more details. The reader familiar with the geometry of this elliptic curve \mathcal{E} can safely skip the first part of the section. Then, in section 2.2 we construct resolved elliptically fibered Calabi-Yau manifolds $\hat{\pi} : \hat{X} \to X$ over an arbitrary base B with this elliptic curve \mathcal{E} as the general fiber. The singular Calabi-Yau manifold is denoted by X We show that these Calabi-Yau manifolds \hat{X} are classified by the choice of two divisors \mathcal{S}_7 , \mathcal{S}_9 in the base B. In particular, we work out all the line bundles that are relevant to formulate the Calabi-Yau constraint of \hat{X} , which is the analog of the Tate model for elliptic fibrations with dP_2 -elliptic fiber.

The content of section 2.2 is a direct extension of the discussion in [51], where the possibility of a full classification of all Calabi-Yau elliptic fibrations with general fiber \mathcal{E} was pointed out, but demonstrated explicitly only for $B = \mathbb{P}^2$.

2.1 The elliptic curve with rank two Mordell-Weil group

The hypersurface description of the elliptic curve \mathcal{E} with a zero point P and two rational points Q and R has been derived in [51] from the existence of a degree three ample line bundle $M = \mathcal{O}(P + Q + R)$ on \mathcal{E} . The result of this analysis is that such an elliptic curve is naturally represented as the generic Calabi-Yau hypersurface in the del Pezzo dP_2 . The Calabi-Yau constraint in dP_2 takes the form

$$p = u(s_1u^2e_1^2e_2^2 + s_2uve_1e_2^2 + s_3v^2e_2^2 + s_5uwe_1^2e_2 + s_6vwe_1e_2 + s_8w^2e_1^2) + s_7v^2we_2 + s_9vw^2e_1,$$
(2.1)

where we introduced the homogeneous coordinates $[u : v : w : e_1 : e_2]$ on dP_2 .⁵ One readily checks that p is the most general section of the anti-canonical bundle $K_{dP_2}^{-1} = \mathcal{O}(3H - E_1 - E_2)$, by noting the following divisor classes of the homogeneous coordinates,

	divisor class	\mathbb{C}^*	'-act	ions
u	$D_u = H - E_1 - E_2$	1	1	1
v	$D_v = H - E_2$	1	0	1
w	$D_w = H - E_1$	1	1	0
e_1	E_1	0 ·	-1	0
e_2	E_2	0	0	-1

Here we also introduced the divisors D_u , D_v and D_w obtained by setting u, v, w to zero, respectively. The group of divisors is generated by H, E_1 and E_2 whose geometric

⁵We deviate here from the conventions of [51] by denoting coordinates on dP_2 by $[u:v:w:e_1:e_2]$ and those on \mathbb{P}^2 by [u':v':w'].



Figure 1. Fan of dP_2 . The coordinates corresponding to its rays are indicated.

interpretation we give momentarily. We note that dP_2 is a toric variety with the $(\mathbb{C}^*)^3$ action on the homogeneous coordinates⁶ specified by the last three columns of (2.2). Its reflexive two-dimensional polytope is given in figure 1.

It is useful for our purposes to recall that a general del Pezzo surface dP_n is obtained from \mathbb{P}^2 by a blow-up at *n* generic points. Thus, the del Pezzo surface dP_2 is the blow-up of \mathbb{P}^2 at two generic points. In terms of the homogeneous coordinates [u': v': w'] on \mathbb{P}^2 , this blow-up is performed in our case at u' = w' = 0 and u' = v' = 0, so that the blow-down map takes the form

$$u' = ue_1e_2, \qquad v' = ve_2, \qquad w' = we_1,$$
 (2.3)

with the two sections e_i associated to the exceptional divisors E_i . We note that one can use this map to represent the elliptic curve \mathcal{E} in (2.1) as a *non-generic* Calabi-Yau onefold in \mathbb{P}^2 . As discussed in [51] this presentation of \mathcal{E} suffices to obtain its Weierstrass model and discriminant. However, for the understanding of elliptic fibrations the completely resolved curve \mathcal{E} in dP_2 with is inevitable.

The divisors classes in (2.2) are the classes of the e_i , i.e. the exceptional divisors E_i of the blow-up (2.3), and the divisor class H, that is the pullback of the hyperplane class on \mathbb{P}^2 . We note the intersections on dP_2 are given as

$$H^{2} = 1, \qquad H \cdot E_{i} = 0, \qquad E_{i} \cdot E_{j} = -\delta_{ij}.$$
 (2.4)

These intersections immediately follow on the one hand from the rule that two divisors corresponding to rays of the same two-dimensional cone have intersection number one and on the other hand from the exceptional set, the Stanley-Reissner ideal SR. The latter encodes all rays which do not share a two dimensional cone and have intersections number zero. From figure 1 one readily obtains

$$SR = \{uv, uw, e_1e_2, e_1v, e_2w\}.$$
(2.5)

⁶Denoting one column vector in the last three columns of (2.2) by ℓ and the homogeneous coordinates collectively as x_i , than the corresponding \mathbb{C}^* -action is defined as $x_i \mapsto \lambda^{\ell_i} x_i$ with $\lambda \in \mathbb{C}^*$.

The rational points Q and R as well as the zero point P on \mathcal{E} are simply given by the intersections of the three independent divisors in (2.2) with the elliptic curve (2.1). We have chosen the three different points on \mathcal{E} that are obtained by intersecting the three divisors D_u , E_1 and E_2 on dP_2 with p = 0 in (2.1). Upon setting all coordinates to one that can not vanish simultaneously with the divisor under consideration due to the exceptional set (2.5) we obtain

$$P: E_2 \cap p = [-s_9: s_8: 1: 1: 0], \qquad Q: E_1 \cap p = [-s_7: 1: s_3: 0: 1],$$

$$R: D_u \cap p = [0: 1: 1: -s_7: s_9].$$
(2.6)

The point P is considered as the zero point on \mathcal{E} and the points Q, R are the rational points. They are the two generators of the rank two Mordell-Weil group of rational points, with the group law given by the addition of points on \mathcal{E} . We note that these three points are generically distinct. P is always distinct from Q, however, we observe P = R for $s_9 = 0$ and Q = R for $s_7 = 0$.

When considering elliptically fibered Calabi-Yau fourfolds \hat{X} with the curve (2.1) as the general elliptic fiber, the points P, Q and R lift to rational sections of the fibration. We note that the coefficients s_i in (2.1) are then base-dependent functions, that can vanish on B. In particular, we see from (2.6) that the points P, Q, and R, respectively, are ill-defined when $s_8 = s_9 = 0$, $s_3 = s_8 = 0$ and $s_7 = s_9 = 0$, respectively. This behavior is typical for rational sections and further discussed in section 3.1. In F-theory compactifications on such a fourfold \hat{X} each of the rational sections gives rise to an Abelian gauge symmetry. Thus, Calabi-Yau fourfolds with general elliptic fiber in dP_2 generically have a rank two Abelian gauge group, i.e. an U(1)×U(1) gauge symmetry.

2.2 General Calabi-Yau fibrations with dP₂-elliptic fiber

In this section we discuss the construction of resolved elliptically fibered Calabi-Yau manifolds \hat{X} with general elliptic fiber in dP_2 . The following results hold for general complex dimension of \hat{X} , in particular for Calabi-Yau three- and fourfolds. We end this section with the concrete example of $B = \mathbb{P}^3$.

Classifying dP_2 -fibrations and their Calabi-Yau hypersurfaces \hat{X}

In general an elliptically fibered Calabi-Yau manifold $\mathcal{E} \to \hat{X} \xrightarrow{\pi} B$ with π denoting the projection to the base B is constructed by first considering the defining equation for the desired elliptic curve \mathcal{E} alone and then by lifting the coefficients in this equation to sections over the base B. In the case at hand, the elliptic curve is described by (2.1). Thus, all we have to do to obtain an elliptic fibration is to promote the coefficients s_i to sections of line bundles on the base B. Finally, the Calabi-Yau condition for (2.1) fixes the respective line bundles for the sections s_i .

The procedure of lifting the s_i to sections of B is described as follows. First, we have to define the ambient space in which the elliptically fibered manifold $\hat{X} \to B$ is embedded. Since the constraint (2.1) merely cuts the elliptic curve \mathcal{E} out of dP_2 , the ambient space is simply a dP_2 -fibration over the base B of \hat{X} . It takes the form

which can be viewed as a generalization of a projective bundle. Here S_7 and S_9 are two divisors on B associated to the vanishing loci of the sections s_7 and s_9 in (2.1). The total space is denoted $dP_2^B(S_7, S_9)$ since it is uniquely determined by these divisors S_7 and S_9 if we demand that the constraint (2.1) defines a Calabi-Yau manifold \hat{X} . In fact, we first note that any the dP_2 -fibration is specified by only two divisors on B. This can be seen by noting that in a general such fibrations the homogeneous coordinates $[u : v : w : e_1 : e_2]$ on dP_2 are sections of five different line bundles on the base B, respectively. However, we can always use the three \mathbb{C}^* -actions in (2.2) to eliminate three of these line bundles, so that only two of the five coordinates on dP_2 take values in non-trivial line bundles. We make the following assignment of line bundles on B to the coordinates,

$$u \in \mathcal{O}_B(\mathcal{S}_9 + [K_B]), \qquad v \in \mathcal{O}_B(\mathcal{S}_9 - \mathcal{S}_7),$$

$$(2.8)$$

where K_B denotes the canonical bundle on B and $[K_B]$ the associated divisor. All other coordinates on dP_2 transform as the trivial bundle on B. We note that this parametrization of the two line bundles for u and v is completely general, because S_7 and S_9 are completely general divisors on B at the moment.

Next, we use these results to readily calculate the total Chern class of $dP_2^B(S_7, S_9)$ from adjunction, see (A.2) in appendix A, from which we obtain its anti-canonical bundle

$$K_{dP_2^B}^{-1} = \mathcal{O}(3H - E_1 - E_2 + 2\mathcal{S}_9 - \mathcal{S}_7), \qquad (2.9)$$

where we suppressed the dependence on S_7 , S_9 for brevity of our notation. Then the Calabi-Yau condition implies that the constraint (2.1) has to be a section of $K_{dP_2}^{-1}$. This immediately fixes the line bundles of all the sections s_i on B. We summarize the sections defining the elliptically fibered Calabi-Yau manifold \hat{X} as follows

section	bundle	section	bundle	
u	$\mathcal{O}(H - E_1 - E_2 + \mathcal{S}_9 + [K_B])$	s_1	$\mathcal{O}(3[K_B^{-1}] - \mathcal{S}_7 - \mathcal{S}_9)$	
v	$\mathcal{O}(H - E_2 + \mathcal{S}_9 - \mathcal{S}_7)$	s_2	$\mathcal{O}(2[K_B^{-1}] - \mathcal{S}_9)$	
w	$\mathcal{O}(H-E_1)$	s_3	$\mathcal{O}([K_B^{-1}] + \mathcal{S}_7 - \mathcal{S}_9)$	
e_1	$\mathcal{O}(E_1)$	s_5	$\mathcal{O}(2[K_B^{-1}] - \mathcal{S}_7)$	(2.10)
e_2	$\mathcal{O}(E_2)$	s_6	$\mathcal{O}([K_B^{-1}])$	
		s_7	$\mathcal{O}(\mathcal{S}_7)$	
		s_8	$\mathcal{O}([K_B^{-1}] + \mathcal{S}_9 - \mathcal{S}_7)$	
		s_9	$\mathcal{O}(\mathcal{S}_9)$	

In particular we see that with the parametrization (2.8) the divisors S_7 and S_9 are indeed associated to s_7 and s_9 as claimed at the beginning.

Basic geometry of Calabi-Yau manifolds with dP₂-elliptic fiber

Having constructed the general elliptically fibered Calabi-Yau manifolds \hat{X} over B, we discuss next the group of divisors on \hat{X} . By construction, the basis of divisors on a generic⁷ \hat{X} is induced by a basis of divisors on the ambient space $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$, which consists of divisors of the base B and the fiber dP_2 . The divisors induced from a basis of divisors D_{α}^b of the base B are the vertical divisors $D_{\alpha} = \pi^*(D_{\alpha}^b)$ of the elliptic fibration $\pi : \hat{X} \to B$. Similarly, the classes H, E_1, E_2 of the fiber dP_2 in (2.2) become divisors on \hat{X} . Then, the points P, Q and R in (2.6) lift to, in general, rational sections of the fibration of $\pi : \hat{X} \to B$, denoted \hat{s}_P , \hat{s}_Q and \hat{s}_R , with \hat{s}_P the zero section. We denote the homology classes of the associated divisors by capital letters,

$$S_P = E_2, \qquad S_Q = E_1, \qquad S_R = H - E_1 - E_2 + S_9 + [K_B].$$
 (2.11)

In general, a rational section is a non-holomorphic map of the base B into \hat{X} , such as $\hat{s}_P : B \to \hat{X}$ for example. A rational section $B \to \hat{X}$ is ill-defined over codimension two loci to the effect that it wraps entire fiber components over these loci. From a given rational section, one can easily obtain a *holomorphic section*, i.e. a holomorphic map $\hat{B} \to$ \hat{X} , by a birational transformation, namely a blow-up $\hat{B} \to B$ at those codimension two loci of B. Usually the zero section \hat{s}_P has been assumed to be holomorphic in F-theory. Only lately, the possibility of a non-holomorphic zero section \hat{s}_P in F-theory has been studied [45, 51, 55]. The group of sections excluding the zero section \hat{s}_P is the *Mordell-Weil group* of rational sections on \hat{X} , which in the case at hand is rank two and generated by \hat{s}_Q , \hat{s}_R . For brevity of our notation, we will occasionally denote the generators of the Mordell-Weil group and their divisor classes collectively as

$$\hat{s}_m = (\hat{s}_Q, \hat{s}_R), \qquad S_m = (S_Q, S_R).$$
 (2.12)

There are some characteristic intersections involving the divisors S_P , S_Q and S_R in (2.11) that immediately follow from the defining properties of a section. We list them in the following and refer to [43, 44, 51, 59] for a more thorough discussion. We also give a simple criterion to distinguish between rational and holomorphic sections. A more detailed account on intersections in the presence of a rational zero section can be found in [55]. Here, we content ourselves with noting that \hat{s}_P is holomorphic if $S_9 = 0$ or $S_8 = 0$, \hat{s}_Q is holomorphic if $S_3 = 0$ or $S_7 = 0$ and \hat{s}_R is holomorphic if $S_7 = 0$ or $S_9 = 0$, cf. the paragraph following (2.6) and [51].

⁷By generic we mean the absence of Cartan divisors D_i from resolutions of codimension one singularities of the fibration of \hat{X} . We will briefly discuss the geometry of \hat{X} in the presence of D_i at the end of this section. We refer to section 6 for more details.

г	ha	fal	11.	omina	in	torgoationa	and	daf	<u>.</u>	tiona	:11	ha	ampaial	in	the	noat	$-\mathbf{f}$	thia	monler
T.	пе	IOI	ш	OWING.	ш	lersections.	and	uer	ш	JUDHS	WIII	De	cruciai	ш	une	rest	OL	UIIIS	WOLK:
				··· ()															

Universal intersection:	$S_P \cdot F = S_m \cdot F = 1$ with general fiber $F \cong \mathcal{E}$, (2.13)
Rational sections:	$\pi(S_P^2 + [K_B^{-1}] \cdot S_P) = \pi(S_m^2 + [K_B^{-1}] \cdot S_m) = 0, (2.14)$ $\mathcal{S}_7 = \pi(S_P \cdot S_R), \qquad \mathcal{S}_9 = \pi(S_Q \cdot S_R), (2.15)$
Holomorphic sections:	$S_P^2 + [K_B^{-1}] \cdot S_P = S_m^2 + [K_B^{-1}] \cdot S_m = 0, \qquad (2.16)$
Shioda maps:	$\sigma(\hat{s}_Q) = S_Q - S_P - [K_B^{-1}], \qquad (2.17)$ $\sigma(\hat{s}_R) = S_R - S_P - [K_B^{-1}] - S_9,$
Height pairing:	$\pi(\sigma(\hat{s}_m) \cdot \sigma(\hat{s}_n)) = \begin{pmatrix} 2[K_B] & [K_B] - \mathcal{S}_7 + \mathcal{S}_9 \\ [K_B] + \mathcal{S}_7 - \mathcal{S}_9 & 2[K_B] - 2\mathcal{S}_9 \end{pmatrix}_{\substack{mn \\ (2.18)}}$

Let us briefly comment on these intersections in the order of their appearance. The intersection (2.13) is an immediate consequence of the definition of a section: its divisor class intersects the general class of the fiber $F \cong \mathcal{E}$ at a point. The relation (2.14) can be shown by an adjunction argument, see section 3.2 for direct cohomology computations. Here we have defined the a projection onto the homology $H^4(B)$ of the base as

$$\pi(\mathcal{C}) = (\mathcal{C} \cdot \Sigma^{\alpha}) D^b_{\alpha}, \qquad \Sigma^{\alpha}_b \cdot D^b_{\beta} = \delta^{\alpha}_{\beta}$$
(2.19)

for every complex surface C in \hat{X} . The intersection pairings on \hat{X} , respectively, B are denoted \cdot and the $\Sigma^{\alpha} = \pi^*(\Sigma_b^{\alpha})$ arise from a basis of curves Σ_b^{α} dual to the divisors D_{α}^b on B as indicated in the last equation in (2.19). We emphasize that in the case of a holomorphic section, the relations (2.14) hold in the full homology of \hat{X} as indicated in (2.16). The divisors S_7 , S_9 are the codimension one loci where the sections collide in the fiber \mathcal{E} , as discussed below (2.6). They are encoded in the intersections (2.15). Next, we introduce the divisors $\sigma(\hat{s}_Q)$, $\sigma(\hat{s}_R)$ in (2.17). The map σ is the Shioda map that takes here the form

$$\sigma(\hat{s}_m) := S_m - \tilde{S}_P - \pi(S_m \cdot \tilde{S}_P), \qquad (2.20)$$

where we introduced the combination [60, 61]

$$\tilde{S}_P = S_P + \frac{1}{2} [K_B^{-1}]. \qquad (2.21)$$

We refer to [33, 43, 44, 51, 59] for more details on the Shioda map and to section 6 for the inclusion of an SU(5)-sector. We note that the divisors (2.17) support U(1)-gauge fields in F-theory due to their vanishing intersections with vertical divisors D_{α} and the zero-section, as well as potential Cartan divisors D_i of non-Abelian groups. Finally, we have calculated the intersection matrix of the Shioda map of \hat{s}_Q , \hat{s}_R in (2.18).

We finish this section by some concluding definitions and remarks on the general structure of the fibrations (2.7) and \hat{X} . First, we summarize the basis of divisors on \hat{X} as

$$D_A = (\tilde{S}_P, D_\alpha, D_i, \sigma(\hat{s}_m)), \quad A = 0, 1, \dots, h^{(1,1)}(\hat{B}) + \operatorname{rk}(G) + 3, \qquad (2.22)$$

where we have collectively denoted the basis (2.17) as $\sigma(\hat{s}_m)$. We have also introduced one set of Cartan divisors D_i with $i = 1, ..., \operatorname{rk}(G)$ in order to prepare for the presence of a non-Abelian group G, as in section 6 with $G = \operatorname{SU}(5)$. These divisors D_i are present for nongeneric \hat{X} with a resolved singularity of type G of the elliptic fibration over codimension one in B. The D_i admit a fibration

 $\begin{array}{ccc} c_{-\alpha_i} & \longrightarrow & D_i \\ & & & \downarrow \\ & & & \mathcal{S}^b_C \end{array} \tag{2.23}$

where the general fiber is a rational curve $c_{-\alpha_i} \cong \mathbb{P}^1$ that corresponds to the simple root $-\alpha_i$ of G. The divisor \mathcal{S}_G^b in B physically supports 7-branes that give rise to the non-Abelian gauge symmetry G in F-theory [14, 15, 18].

Next, we expand the canonical bundle K_B of the base B in terms of the vertical divisors D_{α} as

$$[K_B] = K^{\alpha} D_{\alpha} \tag{2.24}$$

with coefficients K^{α} . Similarly, we expand the divisors

$$\mathcal{S}_7 = n_7^{\alpha} D_{\alpha}^b, \qquad \mathcal{S}_9 = n_9^{\alpha} D_{\alpha}^b, \qquad (2.25)$$

with general positive integral coefficients n_7^{α} , n_9^{α} , $\alpha = 1, \ldots, h^{(1,1)}(B)$. It is important to emphasize that the coefficients n_7^{α} , n_9^{α} are in general further bounded from above by the requirement that all sections s_i in (2.10) are generic, i.e. that the line bundle of s_i admits sufficiently many holomorphic sections. If this is not the case we expect additional singularities in \hat{X} , potentially corresponding to a minimal (non-Abelian) gauge symmetry in F-theory. For this reason, we will in the rest of this work assume that \hat{X} can be constructed with generic s_i .

Despite these restrictions on the integers n_7^{α} and n_9^{α} we would like to point out that the constructions of the fibration (2.7) and of \hat{X} hold in general for an arbitrary base Band arbitrary complex dimension. In particular this analysis applies to an arbitrary choice of divisors S_7 and S_9 within these bounds. In particular the general construction here reproduce immediately the classification in [51] with $B = \mathbb{P}^2$ as a special case.

dP_2 -fibrations over $B = \mathbb{P}^3$ with generic Calabi-Yau hypersurfaces \hat{X}

We conclude with the discussion of the special case $B = \mathbb{P}^3$, which will be considered in later sections of this work. In this case, there is only one divisor in the base, the hyperplane H_B , so that the dP_2 -fibration (2.7) is specified only by two integers $n_7 \equiv n_7^1$, $n_9 \equiv n_9^1$. In



Figure 2. Each dot corresponds to a dP_2 -fibration over \mathbb{P}^3 with generic Calabi-Yau \hat{X} .

this case we use the notation

where we suppress the base $B = \mathbb{P}^3$ when denoting the total space (2.7) of the fibration if the context is clear.

We note that $K_{\mathbb{P}^3}^{-1} = \mathcal{O}_{\mathbb{P}}(4)$. In this case all sections s_i exist iff all bundles in the second table in (2.10) have non-negative degree. This puts the following conditions on the integers n_7 , n_9 ,

$$0 \le n_7, n_9 \le 8, \quad n_7 + n_9 \le 12, \quad 0 \le 4 + n_7 - n_9, \quad 0 \le 4 + n_9 - n_7.$$
 (2.27)

The domain of allowed valued for n_7 and n_9 are displayed in figure 2. As we will see in section 5.4 we can torically construct the Calabi-Yau fourfolds \hat{X} for a wide range of values of n_7 , n_9 . The general strategy to build the corresponding reflexive polytopes is outlined in appendix G. It is satisfying, that in the toric context the conditions (2.27) are enforced by reflexivity of the toric polytope, i.e. for values n_7 , n_9 exceeding the bounds (2.27) the toric polytope is no longer reflexive.

3 Calabi-Yau fourfolds with rank two Mordell-Weil

In this section we analyze F-theory compactifications to four dimensions on a generic elliptically fibered Calabi-Yau fourfold \hat{X} over a base B with general fiber \mathcal{E} in dP_2 . These compactifications have a gauge theory with U(1)×U(1) gauge group and a number of chiral matter fields in representations $\mathbf{1}_{(q_1,q_2)}$. The possible U(1)×U(1)-charges (q_1,q_2) have been determined recently in [46, 51] and the full 6D anomaly-free spectrum including matter multiplicities has been derived for $B = \mathbb{P}^2$ in [51].

Here we extend this geometric analysis to fourfolds. The main difference to the 6D case is that matter is not localized anymore at points in B, but on in general rather complicated matter curves. The determination of these matter curves and some of their associated matter surfaces, along with the Yukawa points, is presented in section 3.1. Then, in section 3.2 we present a method to determine the cohomology ring of the fourfold \hat{X} . We

use these techniques to derive general expressions for the Euler number of \hat{X} and its second Chern class. For the example of $B = \mathbb{P}^3$ we finally compute the full vertical cohomology group. These calculations serve as a preparation for the computation of 4D chiralities in section 5, which requires both the knowledge of matter surfaces and the construction of G_4 -flux.

3.1 Singularities of the fibration: matter Surfaces & Yukawa points

We organize this section into a detailed discussion of codimension two singularities in section 3.1.1 and a very brief account on codimension three singularities in section 3.1.2.

3.1.1 Matter: codimension two

In general, the determination of the matter sector in F-theory vacua with general gauge group requires a detailed analysis of singularities of the elliptic fibration of the Calabi-Yau fourfold at codimension two in the base B, where the elliptic fiber \mathcal{E} becomes reducible. Then one has to identify the isolated rational curve $c_{\mathbf{w}}$ in the fiber over these loci, since these correspond in F-theory to matter in a representation \mathbf{R} from wrapped M2-brane states. These curves are in one-to-one correspondence to the weights \mathbf{w} of the representations \mathbf{R} and accordingly labeled. In the case of elliptically fibered Calabi-Yau fourfolds, the codimension two matter loci are Riemann surfaces of genus g, the so-called matter curves $\Sigma_{\mathbf{R}}$ in B conveniently labeled by the corresponding matter representation \mathbb{R} . In addition, for the determination of four-dimensional chirality, compare section 5, we have to know the homology classes of the associated matter surfaces

which are constructed as the fibration of the rational curve $c_{\mathbf{w}}$ corresponding to a given weight \mathbf{w} of the representation \mathbf{R} fibered over $\Sigma_{\mathbf{R}}$.

In this section we determine the matter curves $\Sigma_{\mathbf{R}}$ and the matter surfaces $C_{\mathbf{R}}^{\mathbf{w}}$ for the six representations occurring in the Calabi-Yau fourfold \hat{X} . As we demonstrate, their determination is complicated by the fact that three of the six the codimension two loci in the base B where the elliptic fiber \mathcal{E} becomes reducible are themselves reducible curves. Their irreducible components are multiple different matter curves $\Sigma_{\mathbf{R}}$. Some of these matter curves, denoted $\Sigma_{\mathbf{R}'}$, fail to be complete intersection and can only be described in terms of their prime ideals. These prime ideals are straightforwardly constructed from the two equations of the original reducible codimension two locus. However, the isolation of rational curves $c_{\mathbf{w}}$ over those matter curves $\Sigma_{\mathbf{R}'}$ is very involved. Thus, in these cases we can not determine the corresponding matter surfaces (3.1) explicitly. Fortunately, we can obtain the other three matter surfaces straightforwardly, and are still able to determine the full F-theory matter spectrum for the fourfold \hat{X} , as outlined in section 5. It would be desirable, however, to reproduce the results obtained there invoking M-/F-theory duality by direct geometric computation based on a better understanding of the matter surfaces $C_{\mathbf{R}'}$ in general.

In any case, we can qualitatively describe all the matter surfaces $C_{\mathbf{R}}^{\mathbf{w}}$ by recalling the construction of the resolved fourfold \hat{X} . The smooth fourfold \hat{X} is formed by two consecutive blow-ups of a singular Weierstrass model X. We depict this schematically as



where the full blow-down map $\hat{\pi} : \hat{X} \to X$ is consequently a composition $\hat{\pi} = \pi_1 \circ \pi_2$. On the left we have the smooth geometry with elliptic fiber constructed in section 2. It can be understood as a toric blow-up $\pi_2 : \hat{X} \to \tilde{X}$ from a non- generic cubic in \mathbb{P}^2 , with corresponding fourfold denoted by \hat{X} . A final blow-down π_1 yields the singular Weierstrass form (WSF) X with $\mathbb{P}^2(1,2,3)$ -fiber. The birational map π_1 is derived in detail in [51], see its defining equations eqs. (3.18) and (3.20) therein.

Having the diagram (3.2) in mind, the three matter surfaces $C_{\mathbf{R}}$ which have a simple description are those generated in the blow-up π_2 . There are three simple codimension two singularities in \tilde{X} , which are precisely the three simple matter curves $\Sigma_{\mathbf{R}}$. Their pullbacks under π_2 are precisely the matter surfaces $C_{\mathbf{R}} = \pi_2^*(\Sigma_{\mathbf{R}})$. Because of the simplicity of both $\Sigma_{\mathbf{R}}$ and the blow-up π_2 , these surfaces have a description as a simple complete intersection in the ambient space $dP_2(\mathcal{S}_7, \mathcal{S}_9)$. In contrast, the other three matter curves $\Sigma_{\mathbf{R}'}$ are the loci of codimension two singularities in the WSF X, which are resolved by the map π_1 . However, these curves $\Sigma'_{\mathbf{R}}$ have a description only in terms of prime ideals and the map π_1 is not a simple toric blow-up but a fully-fledged birational map [51]. These two complications make an explicit determination of the surfaces $\mathcal{C}_{\mathbf{\Sigma}'_{\mathbf{R}}}$ hard. Nevertheless, the matter surfaces are again abstractly given by $\mathcal{C}_{\mathbf{R}'} = \pi_1^*(\Sigma'_{\mathbf{R}})$, which are ruled surfaces over $\Sigma'_{\mathbf{R}}$. Thus, the determination of the exceptional loci of the map π_1 might be a first step towards an understanding of these matter surfaces.

Summary of matter representations & their matter curves

Before going into technical calculations of matter curves and surfaces, let us briefly summarize the matter content as it has been determined in [46, 51].

There are six different matter representations $\mathbf{R} = \mathbf{1}_{(q_1,q_2)}$ in the F-theory compactification on the fourfold \hat{X} . The list of realized U(1)×U(1)-charges, together with the



Figure 3. I_2 -fiber from resolving a codimension two singularity of the fibration of \hat{X} .

Matter	Homology class of $\Sigma_{\mathbf{R}}$ in B	
$1_{(1,0)}$	$6[K_B^{-1}]^2 + 4[K_B^{-1}] \cdot S_7 - 5[K_B^{-1}] \cdot S_9 + S_9^2 + S_7 \cdot S_9 - 2S_7^2$	
$1_{(0,1)}$	$6[K_B^{-1}]^2 + 4[K_B^{-1}] \cdot (\mathcal{S}_7 + \mathcal{S}_9) - 2\mathcal{S}_7^2 - 2\mathcal{S}_9^2$	
$1_{(1,1)}$	$6[K_B^{-1}]^2 + 4[K_B^{-1}] \cdot S_9 - 5[K_B^{-1}] \cdot S_7 + S_7^2 + S_7 \cdot S_9 - 2S_9^2$	(3.3)
$1_{(-1,1)}$	$\left([K_B^{-1}] + \mathcal{S}_7 - \mathcal{S}_9 ight) \cdot \mathcal{S}_7$	
$1_{(0,2)}$	$\mathcal{S}_7\cdot\mathcal{S}_9$	
${f 1}_{(-1,-2)}$	$\mathcal{S}_9 \cdot \left([K_B^{-1}] + \mathcal{S}_9 - \mathcal{S}_7 \right)$	

cohomology class of the corresponding matter curves $\Sigma_{\mathbf{R}}$ determined below, reads

Here we used as before the notation $[K_B^{-1}]$ for the anti-canonical divisor of the base and denoted the intersection on B as '.'. These representations of matter fields are modelindependent and in particular do not depend on the choice of base B. The last three matter representations arise from rational curves created in the blow-up π_2^{-1} in (3.2). Their matter curves are simply described by $s_3 = s_7 = 0$, $s_7 = s_9 = 0$ and $s_8 = s_9 = 0$ in the order of their appearance in (3.3). The first three representations arise from rational curves from the blow-up π_1^{-1} in (3.2). The determination of their matter curves is more involved and presented below.

All the matter representations in (3.2) arise from M2-branes on rational curves $c_{\mathbf{w}}$ with wight $\mathbf{w} = (q_1, q_2)$. These charges are calculated by the intersection of the curve $c_{\mathbf{w}}$ with the Shioda maps $\sigma(\hat{s}_Q)$, $\sigma(\hat{s}_R)$ defined in (2.17) as

$$q_m \equiv \sigma(\hat{s}_m) \cdot c_{\mathbf{w}} = (S_m \cdot c_{\mathbf{w}}) - (S_P \cdot c_{\mathbf{w}}), \qquad (3.4)$$

All curves $c_{\mathbf{w}}$ are part of an I_2 -fiber. Along the matter surfaces in (3.3) the general elliptic fiber \mathcal{E} splits into two rational curves $c_1, c_2 \cong \mathbb{P}^1$ intersecting in two points with one curve, say c_1 , the original singular fiber and the other curve $c_2 \equiv c_{\mathbf{w}}$. We write this as

$$I_2$$
-fiber : $\mathcal{E} = c_1 + c_2, \quad c_1 \cdot c_2 = 2.$ (3.5)

A cartoon of such a reducible fiber together with possible locations of the points P, Q and R is depicted in figure 3. In terms of the Calabi-Yau constraint (2.1) the split of \mathcal{E} into an I_2 -fiber is visible as a factorization at a point $pt \in \Sigma_{\mathbf{R}}$ as

$$p|_{pt} = p_1 \cdot p_2 \,. \tag{3.6}$$

Here the two rational curves in (3.5) are described by one of these two factors, for example $c_1 = \{p_1 = 0, pt \in \Sigma_{\mathbf{R}}\}$ and $c_2 = \{p_2 = 0, pt \in \Sigma_{\mathbf{R}}\}.$

Matter surfaces $C_{(-1,1)}$, $C_{(0,2)}$, $C_{(-1,-2)}$ and their homology classes

Next, we determine the matter surfaces $C_{\mathbf{R}}^{\mathbf{w}}$ for the last three representations in (3.3). As we will see, this is straightforward since the matter curves $\Sigma_{\mathbf{R}}$ in these cases are irreducible varieties of the simple form $s_i = s_j = 0$ for appropriate i, j. This implies that the factorization of the elliptic fiber $\mathcal{E} = c_1 + c_2$ is manifest over the entire matter curve $\Sigma_{\mathbf{R}}$ and the matter surfaces $C_{\mathbf{R}}^{\mathbf{w}}$ can be described by a complete intersection of three constraints in the ambient space $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$. Then its homology class is given simply by the product of the divisor classes of each of these constraints.

The resulting homology classes of matter surfaces read

Matter surface	Homology class	
$\mathcal{C}_{1_{(-1,1)}}$	$([K_B^{-1}] + \mathcal{S}_7 - \mathcal{S}_9) \cdot \mathcal{S}_7 \cdot E_1$	(3.7)
$\mathcal{C}_{1_{(0,2)}}$	$\mathcal{S}_7 \cdot \mathcal{S}_9 \cdot ([K_B^{-1}] + \mathcal{S}_9 - \mathcal{S}_7 + 2H)$	(0.1)
$\mathcal{C}_{1_{(-1,-2)}}$	$([K_B^{-1}] + \mathcal{S}_9 - \mathcal{S}_7) \cdot \mathcal{S}_9 \cdot (3H - E_1 - 2E_2 + 2\mathcal{S}_9 - \mathcal{S}_7)$	

Here we suppressed the weight **w** since it is identical to the charges (q_1, q_2) . We obtain the homology class of the first matter surface $C_{\mathbf{1}_{(-1,1)}}$ by noting its description as the complete intersection

$$\mathcal{C}_{\mathbf{1}_{(-1,1)}} = \{ s_3 = s_7 = 0 \,, \ e_1 = 0 \} \tag{3.8}$$

in the ambient space $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$. Here the first two equations describe the matter curve $\Sigma_{\mathbf{1}_{(-1,1)}} = \{s_3 = s_7 = 0\}$ over which the Calabi-Yau constraint (2.1) factorizes as $p = p_1 p_2$ with one factor given by e_1 , cf. section 4.3 of [51]. Thus, the rational isolated curve is described as $c_{(-1,1)} = \{e_1 = 0, pt \in \Sigma_{(-1,1)}\}$ over all the points of $\Sigma_{(-1,1)}$. The homology class in the first line of (3.7) for $\mathcal{C}_{\mathbf{1}_{(-1,1)}}$ in the ambient space $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$ is then easily obtained from (3.8) employing the assignments (2.10) of line bundles to s_3 , s_7 and e_1 .

Similarly we obtain the homology classes of the matter surfaces $C_{\mathbf{1}_{(0,2)}}$ and $C_{\mathbf{1}_{(-1,-2)}}$. In the former case the matter curve is $\Sigma_{\mathbf{1}_{(0,2)}} = \{s_7 = s_9 = 0\}$ over which the Calabi-Yau constraint (2.1) factorizes globally with the isolated rational curve given by [51]

$$c_{(0,2)} = \left\{ s_1 e_1^2 e_2^2 u^2 + s_2 e_1 e_2^2 u v + s_3 e_2^2 v^2 + s_5 e_1^2 e_2 u w + s_6 e_1 e_2 v w + s_8 e_1^2 w^2 = 0 \right\}.$$
 (3.9)

Here it is understood that the sections s_i are evaluated on $\Sigma_{\mathbf{1}_{(0,2)}}$. The homology class of this complete intersection is the product of the class of $c_{(0,2)}$ and of $\Sigma_{\mathbf{1}_{(0,2)}}$ and we immediately reproduce the second line in (3.7) using (2.10). Finally, the matter curve for $\mathbf{1}_{(-1,-2)}$ is given by $\Sigma_{\mathbf{1}_{(-1,-2)}} = \{s_8 = s_9 = 0\}$ and the isolated rational curve is [51]

$$c_{(-1,-2)} = \left\{ s_1 e_1^2 e_2 u^3 + s_2 e_1 e_2 u^2 v + s_3 e_2 u v^2 + s_5 e_1^2 u^2 w + s_6 e_1 u v w + s_7 v^2 w = 0 \right\}, \quad (3.10)$$

where as before the s_i are evaluated on $\Sigma_{\mathbf{1}_{(-1,-2)}}$. Then, the matter surface is again a complete intersection in the ambient space (2.7) and its homology class, employing the line bundles (2.10), is indeed given by the third line in (3.7).

Matter curves $\Sigma_{(1,0)}$, $\Sigma_{(0,1)}$, $\Sigma_{(1,1)}$ and their prime ideals

As mentioned before, the three remaining matter curves $\Sigma_{\mathbf{R}'}$ are themselves no simple complete intersections, but contained in a reducible codimension two subvarieties in Bthat are complete intersections. To isolate the component $\Sigma_{\mathbf{R}'}$ of interest we have to determine its prime ideal. This prime ideal is generated by more than two constraints, but still describes a codimension two variety in B. In addition, the factorization (3.6) describing the split $\mathcal{E} = c_1 + c_2$ of the elliptic curve does not occur globally over the matter curves $\Sigma_{\mathbf{R}'}$, but is manifest only at generic points of $\Sigma_{\mathbf{R}'}$. These combined effects render the determination of the homology class of the matter surfaces $C_{\mathbf{R}'}$ unfeasible. However, we can obtain the homology class of the matter curves $\Sigma_{\mathbf{R}'}$ as shown next. For completeness we will also present the prime ideal for one illustrative example. In general the prime ideals are needed for a thorough analysis of codimension three singularities presented in section 3.1.2.

We begin with the determination of the homology classes of the matter curves $\Sigma_{\mathbf{R}'}$. They can be obtained by first determining the homology class of the two equations for the reducible codimension two locus in B and by then subtracting the classes of those components $\Sigma_{\mathbf{R}}$ we are not interested in. As in the six-dimensional case [51] we have to subtract the components $\Sigma_{\mathbf{R}}$ with the right multiplicity, which is computed by the resultant⁸ of the two equations at the root corresponding to $\Sigma_{\mathbf{R}}$. The resulting homology classes of this computations give the first three lines of (3.3). We work out these homology classes in detail in the remainder of this section.

First we present the equations for the reducible codimension two loci in B that contain the three matter curves $\Sigma_{\mathbf{R}}$ that we are interested in as irreducible components. These codimension two loci read [51]

$$\begin{aligned} \log_{1} &= \left\{ s_{7}s_{8}^{2} + s_{9}(s_{5}s_{9} - s_{6}s_{8}) = s_{3}s_{8}^{2} - s_{2}s_{8}s_{9} + s_{1}s_{9}^{2} = 0 \right\}, \end{aligned} \tag{3.11} \\ \log_{2} &= \left\{ s_{3}s_{6}s_{8} - s_{2}s_{7}s_{8} - s_{3}s_{5}s_{9} + s_{1}s_{7}s_{9} = s_{3}^{2}s_{8}^{2} + s_{7}\left(s_{1}s_{7}s_{8} + s_{2}s_{5}s_{9} - s_{2}s_{6}s_{8}\right) \\ &+ s_{3}\left(s_{6}^{2}s_{8} - s_{5}s_{7}s_{8} - s_{5}s_{6}s_{9} - s_{2}s_{8}s_{9} + s_{1}s_{9}^{2}\right) = 0 \right\}, \end{aligned} \\ \log_{3} &= \left\{ 2s_{7}^{3}s_{8}^{3} + s_{3}s_{9}^{3}\left(s_{5}s_{9} - s_{6}s_{8}\right) + s_{7}^{2}s_{8}s_{9}\left(2s_{5}s_{9} - 3s_{6}s_{8}\right) - s_{7}s_{9}^{2}\left(s_{5}s_{6}s_{9} + s_{2}s_{8}s_{9} - s_{6}^{2}s_{8} - 2s_{3}s_{8}^{2} - s_{1}s_{9}^{2}\right) = s_{7}^{4}s_{8}^{4} + 2s_{7}^{3}s_{8}^{2}s_{9}\left(s_{5}s_{9} - s_{6}s_{8}\right) + s_{7}s_{9}^{3}\left(2s_{3}s_{8} - s_{2}s_{9}\right)\left(s_{5}s_{9} - s_{6}s_{8}\right) \\ &- s_{7}^{2}s_{9}^{2}\left(2s_{5}s_{6}s_{8}s_{9} + s_{2}s_{8}^{2}s_{9} - s_{6}^{2}s_{8}^{2} - 2s_{3}s_{8}^{3} - s_{5}^{2}s_{9}^{2}\right) + s_{3}s_{9}^{4}\left(s_{3}s_{8}^{2} + s_{9}\left(s_{1}s_{9} - s_{2}s_{8}\right)\right) = 0 \right\} \end{aligned}$$

By calculating the associated prime ideals⁹ of loc₁ we see that it has two irreducible components. One is obviously the matter curve $\Sigma_{\mathbf{1}_{(-1,-2)}} = \{s_8 = s_9 = 0\}$ and the other one is the matter curve $\Sigma_{\mathbf{1}_{(1,1)}}$. Then we determine the homology class of the reducible variety loc₁. We further recall from [51] that the order of the root $(s_8, s_9) = (0, 0)$ of the two polynomials in loc₁ is 4. Thus, decompose the class of loc₁ as

$$[\operatorname{loc}_{1}] = (2[K_{B}^{-1}] + 2\mathcal{S}_{9} - \mathcal{S}_{7}) \cdot (3[K_{B}^{-1}] + \mathcal{S}_{9} - \mathcal{S}_{7}) \cong \Sigma_{(1,1)} + 4\Sigma_{(-1,-2)} \Rightarrow \Sigma_{(1,1)} \cong 6[K_{B}^{-1}]^{2} + 4[K_{B}^{-1}] \cdot \mathcal{S}_{9} - 5[K_{B}^{-1}] \cdot \mathcal{S}_{7} + \mathcal{S}_{7}^{2} + \mathcal{S}_{7} \cdot \mathcal{S}_{9} - 2\mathcal{S}_{9}^{2},$$
(3.12)

⁸In general, the resultant gives the order of a root of two polynomials in two variables.

⁹The ideal generated by loc₁ is the intersection of its associated primary ideals. An ideal I is primary ideal if $ab \in I$ implies $a \in I$ or $b^n \in I$ for some n > 0. If n = 1, the ideal I is a prime ideal.

where we used (2.10) and denote the equivalence relation in homology as ' \cong '. Here we also used that the homology class of $\Sigma_{\mathbf{1}_{(-1,-2)}}$ as given in the third line of (3.7) by the first two factors. Thus, we have obtained the homology class of $\Sigma_{\mathbf{1}_{(1,1)}}$ as in (3.3).

Similarly, we obtain the homology class of $\Sigma_{\mathbf{1}_{(1,0)}}$. We calculate four associated prime ideals of loc₂ in (3.11) that correspond to four different irreducible components. These components are the curves $\Sigma_{\mathbf{1}_{(-1,1)}} = \{s_3 = s_7 = 0\}, \Sigma_{\mathbf{1}_{(-1,-2)}} = \{s_8 = s_9 = 0\}, \Sigma_{\mathbf{1}_{(1,1)}}$ and finally $\Sigma_{\mathbf{1}_{(1,0)}}$. By calculating the resultants of loc₂ at the relevant roots, we obtain multiplicities one for all irreducible components. Thus, we obtain the homology class of $\Sigma_{(1,0)}$ from decomposition of the class of loc₂ as

$$[loc_2] = 12[K_B^{-1}]^2 = \Sigma_{(-1,1)} + \Sigma_{(-1,-2)} + \Sigma_{(1,1)} + \Sigma_{(1,0)} \Rightarrow \Sigma_{(1,0)} \cong 6[K_B^{-1}]^2 + 4[K_B^{-1}] \cdot S_7 - 5[K_B^{-1}] \cdot S_9 + S_9^2 + S_7 \cdot S_9 - 2S_7^2,$$
 (3.13)

where we have used (2.10), the homology classes of matter curves in (3.7) as well as in (3.12). This is the result in (3.3).

Finally, we determine the homology class of the matter curve $\Sigma_{\mathbf{1}_{(0,1)}}$. The ideal loc₃ in (3.11) has five prime ideals corresponding to the matter curves $\Sigma_{\mathbf{1}_{(-1,1)}}$, $\Sigma_{\mathbf{1}_{(0,2)}}$, $\Sigma_{\mathbf{1}_{(-1,-2)}}$, $\Sigma_{\mathbf{1}_{(1,1)}}$ and the matter curve $\Sigma_{\mathbf{1}_{(0,1)}}$ we are interested in. The multiplicities of the irreducible components we are not interested are calculated as one, 16, 16, one, respectively, by the corresponding resultants of loc₃. Thus, we calculate the homology class of the curve $\Sigma_{\mathbf{1}_{(0,1)}}$ from the homology class of loc₃ as

$$\log_3 = 12[K_B^{-1}]^2 = \Sigma_{(-1,1)} + 16 \cdot \Sigma_{(0,2)} + 16 \cdot \Sigma_{(-1,-2)} + \Sigma_{(1,1)} + \Sigma_{(0,1)} \Rightarrow \Sigma_{(0,1)} = 6[K_B^{-1}]^2 + 4[K_B^{-1}] \cdot (\mathcal{S}_7 + \mathcal{S}_9) - 2\mathcal{S}_7^2 - 2\mathcal{S}_9^2,$$
 (3.14)

where we used the homology class of the matter curves in (3.7) and (3.12). This is the homology class in (3.3).

We conclude this discussion by presenting the associated prime ideal of selected matter surfaces as an instructive preparation of section 3.1.2. The prime ideal of $\Sigma_{\mathbf{1}_{(1,1)}}$ reads

$$\mathcal{P} = \left\{ s_3^2 s_5^2 + s_7 \left(s_2^2 s_5 - s_1 s_2 s_6 + s_1^2 s_7 \right) + s_3 \left(-s_2 s_5 s_6 + s_1 \left(s_6^2 - 2 s_5 s_7 \right) \right), \\ s_3 s_5^2 s_9 + s_2 s_5 \left(s_7 s_8 - s_6 s_9 \right) + s_1 \left(-s_6 s_7 s_8 + s_6^2 s_9 - s_5 s_7 s_9 \right), \\ s_3 s_5 s_8 - s_1 s_7 s_8 - s_2 s_5 s_9 + s_1 s_6 s_9, \\ s_3 s_8^2 + s_9 \left(-s_2 s_8 + s_1 s_9 \right), \\ s_7 s_8^2 + s_9 \left(-s_2 s_8 + s_1 s_9 \right), \\ s_7 s_8^2 + s_9 \left(-s_6 s_8 + s_5 s_9 \right) \right\}.$$
(3.15)

The dimension of \mathcal{P} is calculated to be six in the ring generated by the s_i which confirms that the irreducible variety described by it is codimension two in B as expected. It is evident from (3.15) that the two irreducible components $\Sigma_{\mathbf{1}_{(1,1)}}$ and $\Sigma_{\mathbf{1}_{(-1,-2)}} = \{s_8 = s_9 = 0\}$ of loc₁ intersect at points in B. These points may correspond to Yukawa points in F-theory since the fiber in the resolved space \hat{X} splits further into three components, as can be seen by a prime ideal analysis. We also determine the prime ideal of the matter curve $\Sigma_{\mathbf{1}_{(1,0)}}$ as

$$\mathcal{P} = \left\{ -s_2^2 s_8^2 + s_2 \left(s_5 s_6 s_8 - s_5^2 s_9 + 2 s_1 s_8 s_9 \right) - s_1 \left(s_6^2 s_8 - s_5 s_6 s_9 + s_1 s_9^2 \right), \\ s_2 s_5 s_7 - s_1 s_6 s_7 - s_2 s_3 s_8 + s_1 s_3 s_9, s_3 s_5 s_7 - s_1 s_7^2 - s_3^2 s_8, \\ s_3 s_6 s_7 - s_2 s_7^2 - s_3^2 s_9, s_3 s_6 s_8 - s_2 s_7 s_8 - s_3 s_5 s_9 + s_1 s_7 s_9 \right\}.$$
(3.16)

From this ideal we see that $\Sigma_{\mathbf{1}_{(1,0)}}$ intersects the matter curve $\Sigma_{\mathbf{1}_{(-1,1)}}$, where again there is an additional split of the fiber.

3.1.2 Yukawa couplings: codimension three

At codimension three the singularities of the fibration enhance further signaling the presence of a Yukawa point. In the case at hand we find an enhancement to an I_3 -singularity, which is resolved into three intersecting \mathbb{P}^1 's in \hat{X} . The loci of I_3 -fibers are determined by looking at zeros of higher order of the discriminant and by checking whether the fiber in the resolution \hat{X} splits further. We find the following loci,

Loci	Yukawa	
$s_8 = s_9 = s_7 = 0$	$1_{(-1,-2)} imes 1_{(0,2)} imes 1_{(1,0)}$	
$s_3 = s_7 = s_9 = 0$	$1_{(0,2)} imes \overline{1}_{(1,-1)} imes \overline{1}_{(-1,-1)}$	(3.17)
$\Sigma_{1_{(1,0)}} \cap \Sigma_{1_{(1,0)}} \cap \Sigma_{1_{(1,0)}}$	$\overline{1}_{(-1,-1)} imes 1_{(1,0)} imes 1_{(0,1)}$	(0.11)
$\Sigma_{1_{(1,0)}} \cap \{s_3 = s_7 = 0\}$	$\overline{1}_{(1,-1)} imes \overline{1}_{(-1,0)} imes 1_{(0,1)}$	
$\Sigma_{1_{(1,1)}} \cap \{s_8 = s_9 = 0\}$	$1_{(1,1)} imes 1_{(-1,-2)} imes 1_{(0,1)}$	

We note that the first three agree with earlier results, see [46]. The last two loci also produce reducible fibers with three irreducible components that can be described in terms of the prime ideals. The study of these new Yukawa points, along with a more thorough discussion of the use of prime ideals, will be postponed to future work.

3.2 The cohomology ring and the Chern classes of \hat{X}

In this section we abstractly calculate the cohomology ring of the fourfold \hat{X} . The central result of these computations is the basis of surfaces or dual (2, 2)-forms in $H^{(2,2)}(\hat{X})$, which is relevant for the construction of G_4 -flux, see section 5.1. Furthermore, for the calculation of the D3-brane tadpole and the quantization of the G_4 -flux, we use these techniques to calculate the general expression for the Euler number and the second Chern class of \hat{X} for a general base B. In addition, we derive the full cohomology ring explicitly for $B = \mathbb{P}^3$, leaving the straightforward generalization to other bases for future works.

We note that the presentation of the cohomology of \hat{X} used here has been employed in the context of toric mirror symmetry for a long time and is in this sense not new. For an F-theory context see e.g. [58, 62] and references therein. We refer also to [21] for cohomology calculations in the same spirit. However, we emphasize that, except for the language that we borrow from toric geometry, the following discussion is based only on reasonable assumptions on the intersections of \hat{X} . Thus, we expect the following procedure to work also in the non-toric case. In particular, not all fourfolds considered here have, to our knowledge, a description in terms of a reflexive polytope, which does, however, not keep us from using them for F-theory and computing their full chiral 4D spectrum.

The basic idea to calculate the cohomology ring $H_V^{(*,*)}(\hat{X})$ of a general elliptically fibered Calabi-Yau fourfold \hat{X} over a base B with general fiber the elliptic curve in dP_2 is to exploit the Stanley-Reissner (SR) ideal¹⁰ SR of the ambient space $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$ together with the linear equivalences of divisors. After dividing out the linear equivalences, the cohomology ring $H_V^{(*,*)}(\hat{X})$ can be represented as the quotient ring R of the form¹¹

$$H_V^{(*,*)}(\hat{X}) \cong \frac{\mathbb{C}[D_\alpha, S_P, S_Q, S_R] \cdot \left[\hat{X}\right]}{SR}, \qquad (3.18)$$

where the basis (2.11) together with the vertical divisors D_{α} are the variables of the free polynomial ring $\mathbb{C}[D_{\alpha}, S_P, S_Q, S_R]$ and SR is considered as an ideal in this ring. For this purpose, the ideal SR has to be translated into intersection relations of those divisors. Note that we have to multiply by the homology class of \hat{X} in $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$ in (2.9) to restrict the intersections on the ambient space $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$ to \hat{X} .¹² By the Calabi-Yau condition this class is precisely given as $\mathcal{O}(\hat{X}) = K_{dP_2}^{-1}$.

The quotient ring (3.18) is graded with each graded piece being finitely generated by monomials in the divisors D_{α} , S_P and S_m of appropriate degree. We denote this ring by R. The k-th graded piece is then identified with

$$H_V^{(k,k)}(\hat{X}) = R^{(k)}, \qquad (3.19)$$

after restriction to \hat{X} , i.e. after dropping the overall factor $K_{dP_2}^{-1}$ in (3.18). More precisely, at grade zero we obtain $H^{(0,0)}(\hat{X}) = \langle 1 \rangle$, at grade one we have $H^{(1,1)}(\hat{X}) = \langle D_{\alpha}, S_P, S_Q, S_R \rangle$. At higher grade we obtain naively as many generators as homogeneous monomials of appropriate degree in the divisors in (3.18). However, due to equivalence relation in R the number of independent monomials is in general smaller. In fact, by Poincaré duality the rings $R^{(3)}$, and $R^{(4)}$ are fixed, i.e. the corresponding Hodge numbers are related as $h^{(3,3)}(\hat{X}) \stackrel{!}{=} h^{(1,1)}(\hat{X}) = 3 + h^{(1,1)}(B)$ and $h^{(4,4)}(\hat{X}) \stackrel{!}{=} h^{(0,0)}(\hat{X}) = 1$. At degree $k \geq 5$ we trivially have $R^{(k)} = \{0\}$ by reasons of dimensionality. In this sense the only non-trivial piece is $R^{(2)} \cong H_V^{(2,2)}(\hat{X})$, and the corresponding Hodge-number $h_V^{(2,2)}(\hat{X})$. Furthermore, it is precisely the elements $H_V^{(2,2)}(\hat{X})$ of independent surfaces on \hat{X} into which the general G_4 -flux on \hat{X} has to be expanded, as discussed in section 5.1.

The main advantage of the representation (3.18) compared to concrete toric models is that it allows us to determine the cohomology ring for all fibrations $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$ in (2.7) over a given base B with general divisors \mathcal{S}_7 , \mathcal{S}_9 . Of course the relevant computations depend on the geometry of the base B since the ideal SR in (3.18) in general is generated by the SR-ideal (2.5) of the fiber dP_2 and of the base B, which differs from case to case. Nevertheless, as we demonstrate next, it is possible to calculate the total Chern class $c(\hat{X})$ and Euler number $\chi(\hat{X})$ of \hat{X} for any base B using minimal assumptions.

¹⁰We merely borrow this term from toric geometry. In general, SR can be any ideal containing all vanishing intersections of divisors on $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$, which not necessarily has to be a toric variety.

¹¹We note that this polynomial ring is only the primary vertical cohomology $H_V^{(*,*)}(\hat{X})$ [63]. This is the subspace of $H^{(*,*)}(\hat{X})$ relevant for G_4 -flux inducing chirality in F-theory. Its complement in $H^{(2,2)}(\hat{X})$ is the horizontal cohomology $H_H^{(2,2)}(\hat{X})$ that encodes complex structure moduli of \hat{X} . See e.g. [58] for an analysis of G_4 -flux in $H_H^{(2,2)}(\hat{X})$ in F-theory.

¹²Generally, not all divisors on \hat{X} arise as restrictions of divisors on the ambient space. However, for generic \hat{X} with elliptic fiber in dP_2 , only divisors in B can potentially miss this assumption. We exclude those B in the following. Note that non-generic \hat{X} can have additional divisors, see the footnote 13.

3.2.1 Second Chern class and Euler number of \hat{X} : general formulas

For the purpose of finding the general expression for $c(\hat{X})$ and, thus, the Euler number $\chi(\hat{X})$, it suffices to know, that the intersections of more than three vertical divisors D_{α} in both $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$ and \hat{X} are zero. The latter is true because of the properties of fibrations. Thus, we are working in the following with the ideal of vanishing intersections on $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$ generated by the ideal (2.2) of the fiber dP_2 supplemented by the vanishing of quartic intersection of vertical divisors,

$$SR' = \{S_R \cdot (S_R + S_Q - S_7 - [K_B]), S_R \cdot (S_R + S_P - S_9 - [K_B]), S_Q \cdot S_P,$$
(3.20)
$$S_Q \cdot (S_R + S_Q - S_7 - [K_B]), S_P \cdot (S_R + S_P - S_9 - [K_B]), D_\alpha \cdot D_\beta \cdot D_\gamma \cdot D_\delta\}.$$

Here we have employed (2.10) in combination with (2.11) to translate (2.5) into intersection relations. The prime in SR' reminds us that we are not working with the full SR-ideal of the base B, but just assume vanishing quartic intersections. As before '.' denotes the intersections product in $dP_2^B(S_7, S_9)$.

Next we can perform the calculation of the total Chern class of \hat{X} . For this purpose we first compute the formal expression of the Chern class $c(\hat{X})$ by adjunction,

$$c(\hat{X}) = \frac{c(dP_2^B)}{1 + c_1(\mathcal{O}(\hat{X}))}.$$
(3.21)

The numerator denotes the total Chern class of $dP_2^B(S_7, S_9)$ and the denominator is the Chern class of its anti-canonical bundle (2.9), which is the class of \hat{X} as mentioned above.

Then, we reduce this expression in the quotient ring (3.18) with SR replaced by the reduced ideal SR' in (3.20). We refer to appendix A for the detailed calculations leading to the following results, as well as for the general expression of the total Chern classes $c(dP_2^B)$ and $c(\hat{X})$ for Calabi-Yau two-, three- and fourfolds. We obtain for the second Chern class $c_2(\hat{X})$ of \hat{X} the expression

$$c_{2}(\hat{X}) = 3c_{1}^{2} + c_{2} - 2S_{Q}^{2} - 3S_{P}^{2} + c_{1}(2S_{Q} + S_{P} + 4S_{R} - 2(S_{7} + S_{9})) + 2S_{7}(S_{Q} - S_{P}) + S_{9}(3S_{P} - 2S_{Q} - S_{R} + S_{7}), \qquad (3.22)$$

where we have expressed all cohomology classes in terms of the basis of divisors (2.11) on the fiber dP_2 and the first and second Chern classes $c_1 \equiv c_1(B)$, respectively, $c_2 \equiv c_2(B)$ of the base *B*. By abuse of notation we denote a divisor and its Poincaré dual (1, 1)-form by the same symbol.

As a first sanity check we note that (3.22) is consistent with the formula for the second Chern class of a fourfold with a generic E_6 -elliptic fiber, i.e. with the elliptic curve in \mathbb{P}^2 . In fact, in the limit $S_7 = S_9 = 0$, the total space $dP_2^B(S_7, S_9)$ formally turns into $\mathbb{P}(\mathcal{O}_B \oplus K_B^{-1} \oplus K_B^{-1})$, the sections S_P , S_Q and S_R become indistinguishable and fuse into a single holomorphic three-section σ , following conventions in the literature. Then the second line in (3.22) vanishes and we use the relation (2.16) for σ^2 to rewrite the first line as

$$c_2(\tilde{X}) \rightarrow c_2(X_{E_6}) = 3c_1^2 + c_2 + 12\sigma c_1,$$
 (3.23)

where we denote by X_{E_6} the fourfold with E_6 -elliptic fiber. This expression is in line with the results obtained in [53].

Similarly, the Euler number of \hat{X} is calculated from the integration of the fourth Chern class $c_4(\hat{X})$ as

$$\chi(\hat{X}) = 3 \int_{B} \left[24c_1^3 + 4c_1c_2 - 16c_1^2(\mathcal{S}_7 + \mathcal{S}_9) + c_1(8\mathcal{S}_7^2 + \mathcal{S}_7\mathcal{S}_9 + 8\mathcal{S}_9^2) - \mathcal{S}_7\mathcal{S}_9(\mathcal{S}_7 + \mathcal{S}_9) \right] .$$
(3.24)

Here the integral over \hat{X} has been reduced to an integral over the base B by first consecutive application of the relation (2.14) and then by employing (2.13), which can be rewritten for Calabi-Yau fourfolds as the intersection relation

$$S_P \cdot D_{\alpha} \cdot D_{\beta} \cdot D_{\gamma} = S_m \cdot D_{\alpha} \cdot D_{\beta} \cdot D_{\gamma} = (D_{\alpha} \cdot D_{\beta} \cdot D_{\gamma})|_B.$$
(3.25)

Here S_m collectively denotes the divisors S_Q , S_R of the sections \hat{s}_Q , \hat{s}_R and D_{α} , D_{β} , D_{γ} are general vertical divisors. We emphasize that our expression of the Euler number (3.24) reproduces the Euler number of [53] as the special case $S_7 = S_9 = 0$. As before S_P , S_Q and S_R become homologous and we obtain

$$\chi(\hat{X}_{E_6}) = 72 \int_B c_1^3 + 12 \int_B c_1 c_2 \,. \tag{3.26}$$

As another consistency check, and also for the sake of the discussion of general flux quantization and the D3-brane tadpole in section 5.1, we calculate the arithmetic genus $\chi_0(\hat{X})$ on \hat{X} . It is calculated from the Todd class $\mathrm{Td}_4(\hat{X})$ by the Hirzebruch-Riemann-Roch index theorem. Since \hat{X} is a simply-connected Calabi-Yau fourfold, its arithmetic genus has to be two,

$$\chi_0(\hat{X}) := \sum_p (-1)^p h^{(p,0)}(\hat{X}) \stackrel{!}{=} 2.$$
(3.27)

This immediately follows from $h^{(0,0)}(\hat{X}) = h^{(4,0)}(\hat{X}) = 1$ and $h^{(p,0)}(\hat{X}) = 0$ otherwise. From index theory, however, we obtain

$$\chi_0(\hat{X}) = \int_{\hat{X}} \mathrm{Td}_4(\hat{X}) = \frac{1}{720} \int_{\hat{X}} (3c_2(\hat{X})^2 - c_4(\hat{X})) = \frac{1}{720} \left(3 \int_{\hat{X}} c_2(\hat{X})^2 - \chi(\hat{X}) \right) . \quad (3.28)$$

Evaluating this integral using our expressions (3.22), (3.24) for the second Chern class and Euler number on \hat{X} , cf. appendix A for details, we obtain

$$\chi_0(\hat{X}) = \frac{1}{12} \int_B c_1 c_2 = 2\chi_0(B) \stackrel{!}{=} 2.$$
(3.29)

Here the first equality is due to a remarkable cancellation of all terms containing the divisors S_7 , S_9 in the second Chern class and Euler number of \hat{X} . In the second equality we used the index theorem for the arithmetic genus of the base

$$\chi_0(B) = \frac{1}{24} \int_B c_1 c_2 \,, \tag{3.30}$$

and the last equality follows from the constraint (3.27). Thus we see, that the arithmetic genus $\chi_0(\hat{X}) = 2$ precisely iff $\chi_0(B) = 1$. In general, one demands the stronger conditions $h^{(1,0)}(B) = h^{(2,0)}(B) = h^{(3,0)}(B) = 0$ since non-trivial (p, 0)-forms of the base B would pull back to (p, 0)-forms on \hat{X} under the projection $\pi : \hat{X} \to B$, which we excluded by assumption.

We note that our result (3.29) for the arithmetic genus is in line with the computations in [57, 64], whose analysis we followed. We also refer to [21] for an application of these techniques to F-theory with SU(5) gauge group.

3.2.2 The full cohomology ring of \hat{X} : base $B = \mathbb{P}^3$

As we demonstrate next, the representation (3.18) for a concrete base B allows us to calculate the full cohomology ring for a general Calabi-Yau fourfold \hat{X} in $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$ with general divisors \mathcal{S}_7 , \mathcal{S}_9 . We exemplify this in the following for the base $B = \mathbb{P}^3$, but note that this analysis can be generalized to other bases. For all details of the intersection calculations as well as the quartic intersections, we refer the reader to appendix B.

In the case $B = \mathbb{P}^3$ the cohomology $H^{(1,1)}(\hat{X})$ is generated according to (2.22) by the divisors D_A . We choose the following basis,

$$H^{(1,1)}(\hat{X}) = \langle H_B, S_P, S_Q, S_R \rangle, \qquad (3.31)$$

where H_B is the only vertical divisor of the fibration, which is pullback of the hyperplane of \mathbb{P}^3 to \hat{X} . The three other divisors are related to the in general rational sections \hat{s}_P , \hat{s}_Q and \hat{s}_R . Employing (2.10) and (2.11), their associated divisor classes are

$$S_P = E_2$$
, $S_Q = E_1$, $S_R = H - E_1 - E_2 + (n_9 - 4)H_B$, (3.32)

where we have used $c_1(K_{\mathbb{P}^3}) = -c_1(\mathbb{P}^3) = -4H_B$. We recall that the divisors \mathcal{S}_7 , \mathcal{S}_9 on \mathbb{P}^3 are specified by integers n_7 , n_9 in the region in figure 2 specifying the total space (2.26) of the dP_2 -fibration over \mathbb{P}^3 .

We set up the construction of the cohomology ring of $\pi : \hat{X} \to \mathbb{P}^3$ via (3.18) by specifying the ideal SR. We note that the SR-ideal in the case of $B = \mathbb{P}^3$ is generated by the Stanley-Reissner ideal (2.5) of the fiber dP_2 and the base, which is just $H_B^4 = 0$. Thus, using the divisor classes (3.31), the resulting ideal is identical to (3.20) with all vertical divisors equal to H_B and with $K_{\mathbb{P}^3}^{-1} = \mathcal{O}_{\mathbb{P}^3}(4)$. Then, we need the anti-canonical bundle $K_{dP_2(n_7,n_9)}^{-1}$ of $dP_2(n_7,n_9)$. It is given in general in (2.9) and easily specialized to $B = \mathbb{P}^3$ using $S_7 = n_7 H_B$ and $S_9 = n_9 H_B$ as well as expressed in the basis (3.32). Now we are equipped with all the necessary quantities to construct the quotient ring representation (3.18) of the cohomology ring $H_V^{(*,*)}(\hat{X})$.

We begin by summarizing the Hodge numbers of the vertical cohomology of \hat{X} as

$$h^{(0,0)}(\hat{X}) = h^{(4,4)}(\hat{X}) = 1, \quad h^{(1,1)}(\hat{X}) = h^{(3,3)}(\hat{X}) = 4, \quad h^{(2,2)}_V(\hat{X}) = 5 \ (4),$$
 (3.33)

where the subscript V indicates that we are considering the vertical subspace, and the number in the bracket denotes the non-generic case with (n_7, n_9) on the boundary¹³ of the allowed region in figure 2. These are the lines $n_7 = 0$, $n_9 = 0$, $n_9 = 4 + n_7$ for $n_7 \le 4$, $n_9 = n_7 - 4$ and $n_9 = 12 - n_7$, both of the latter two for $4 < n_7$.

Indeed, we obtain these Hodge numbers as follows from the ring (3.18). At degree zero, which is $H^{(0,0)}(\hat{X})$, the only generator is the trivial element 1. The graded piece $R^{(1)} \cong H^{(1,1)}(\hat{X})$ is generated by the four divisors D_A . At degree two, i.e. $H_V^{(2,2)}(\hat{X})$, there are ten different combinations $D_A \cdot_{\hat{X}} D_B$, of which, however, only five are generically inequivalent. As outlined in appendix B a choice of basis, denoted in general by C_r with $r = 1, \ldots, h_V^{(2,2)}(\hat{X})$, for $H_V^{(2,2)}(\hat{X})$ is given by

$$H_V^{(2,2)}(\hat{X}) = \langle H_B^2, H_B \cdot S_P, H_B \cdot \sigma(\hat{s}_Q), H_B \cdot \sigma(\hat{s}_R), S_P^2 \rangle.$$
(3.34)

Here $\sigma(\hat{s}_Q)$, respectively, $\sigma(\hat{s}_R)$ are the Shioda maps (2.17) of the sections \hat{s}_Q , \hat{s}_R . In the case at hand these take the form

$$\sigma(\hat{s}_Q) = S_Q - S_P - 4H_B, \quad \sigma(\hat{s}_R) = S_R - S_P - (4 + n_9)H_B$$
(3.35)

We can evaluate the 5 × 5-intersection matrix $\eta^{(2)}$ in the basis (3.34) using the quartic intersections in (B.4) as

$$\eta^{(2)} = \begin{pmatrix} 0 & 1 & 0 & 0 & -4 \\ 1 & -4 & 0 & 0 & 16 + (n_7 - n_9 - 4) n_9 \\ 0 & 0 & -8 & n_7 - n_9 - 4 & n_9 (4 - n_7 + n_9) \\ 0 & 0 & \eta_{34}^{(2)} - 2 (4 + n_9) & 2n_9 (4 - n_7 + n_9) \\ -4 & \eta_{25}^{(2)} & \eta_{35}^{(2)} & \eta_{45}^{(2)} & -64 - (8 + n_7 - 2n_9) (n_7 - n_9 - 4) n_9 \end{pmatrix}.$$
 (3.36)

Here entries $\eta_{rs}^{(2)}$ that are determined by symmetry are omitted and denoted by $\eta_{sr}^{(2)}$. We note that for values of (n_7, n_9) on the boundary of figure 2, there are only four inequivalent such surfaces. A quick way to see this is by calculating the rank of the matrix (3.36) which is generically five, but decreases to four in these cases. In all these cases we can drop the basis element S_P^2 in (3.34) since it becomes homologous to the other four basis elements, cf. (B.8), (B.10), (B.11) and (B.12) in appendix B. The corresponding intersection matrix $\eta^{(2)}$ is then obtained from (3.36) by deleting the last row and column. We note that both the knowledge of the basis (3.34) as well as of the intersections (3.36) is essential for the construction of G_4 -flux in section 5.1.

At degree three, there are 20 combinations of three divisors D_A , however, there are only four inequivalent ones, which is expected by duality of $H^{(3,3)}(\hat{X})$ and $H^{(1,1)}(\hat{X})$. Finally, at degree four, which is $H^{(4,4)}(\hat{X})$, there are 35 different quartic monomials in the D_A ,

¹³We note that for the two special values $(n_7, n_9) = (4, 8)$, (8, 4) there is one additional divisor on \hat{X} that is not induced from the ambient space. For these special values we see from (2.10) that \hat{X} is not generic since s_1, s_2, s_3 , respectively, s_1, s_5, s_8 are constants. Then, we can perform a variable transformation on the fiber coordinates to achieve $s_1 = 0$, i.e. the elliptic curve will have an additional section at u = 1, v = w = 0. The elliptic fiber can then be embedded into dP_3 with all sections toric. We thank Jan Keitel for pointing out the existence of a non-toric divisor for non-generic $\hat{X} \to \mathbb{P}^3$.

of which there is only one inequivalent combination. This combination is precisely the quartic intersections on \hat{X} . The higher graded pieces of $H^{(k,k)}(\hat{X})$, k > 4 vanish, which is intuitively clear since there are at most quartic intersections on a Calabi-Yau fourfold.

We conclude by summarizing some key intersections on \hat{X} which are discussed in section 2.2 as general properties of the fibrations, that can, however, be proven explicitly using the representation (3.18). Of the complete quartic intersections summarized in B.4 of appendix B we highlight the following intersections,

$$S_P \cdot S_R \cdot H_B^2 = n_9 S_* \cdot H_B^3, \qquad S_Q \cdot S_R \cdot H_B^2 = n_7 S_* \cdot H_B^3, \qquad S_* \cdot H_B^3 = 1, \qquad S_*^2 \cdot H_B^2 = -4,$$
(3.37)

where S_* collectively denotes all the divisor classes S_P , S_Q and S_R of the sections. Here the first two relations are the versions of (2.15), respectively, on $B = \mathbb{P}^3$. The third relation implies that a section of the elliptic fibration of \hat{X} intersects the generic fiber $F = \pi^*(pt)$ for a generic point pt in B precisely at one point, cf. (2.13). Finally, the last relation is the analog of (2.14) on $B = \mathbb{P}^3$. We note that for a holomorphic section, this relation holds without intersection with H_B^2 . Indeed, this is confirmed by the concrete cohomology calculation in appendix B for the zero-section S_P in (B.12).

4 G_4 -flux conditions in F-theory from CS-terms: Kaluza-Klein states on the 3D Coulomb branch

In this section we discuss the construction of G_4 -flux in F-theory compactifications on general elliptically fibered Calabi-Yau fourfolds \hat{X} with a non-trivial Mordell-Weil group and a non-holomorphic zero section.

We define G_4 -flux in F-theory through the *M*-theory compactification on the resolved fourfold \hat{X} , that is dual to F-theory reduced on a circle to 3D. The general constraints on G_4 -flux in M-theory compactifications are reviewed in section 4.1. In addition to these conditions, G_4 -flux that is admissible for an *F*-theory compactification has to obey additional constraints. The form of G_4 -flux that yields a consistent F-theory has been derived first in [65] by requiring a Lorentz-invariant uplift to four dimensions. Here we discuss a different logic to obtain constraints on the G_4 -flux. As we point out in section 4.2, these constraints are appropriately formulated as the requirement of the vanishing of certain Chern-Simons (CS) terms on the Coulomb branch of the effective three-dimensional theory.¹⁴ In particular, consistent conditions on the G_4 -flux are obtained only if one-loop corrections of both massive states on the 3D Coulomb branch as well as *Kaluza-Klein* (KK) states are taken into account. Most importantly, the presence of a non-holomorphic zero section is linked to the existence of new CS-terms for the KK-vector, that are generated by KK-states, whereas other CS-terms receive additional shifts.

We present for the first time a consistent set of conditions on G_4 -flux in F-theory compactifications with a non-holomorphic zero section. We also evaluate explicitly the

 $^{^{14}}$ See also [66, 67] for recent related studies of connections between CS-terms and contact terms in 3D effective field theories with background fields in the context of F-maximization.

corrections of massive states to 3D CS-levels for the F-theory/M-theory compactification on the fourfold \hat{X} with dP_2 -elliptic fiber.

The following discussion is an extension of [44], where KK-states have first been discussed in the context of 4D anomaly cancellation, and inspired by the analogous sixdimensional analysis in [55],¹⁵ see also [68] for the relevance of KK-states in the description of self-dual two-forms in 6D/5D.

4.1 A brief portrait of G_4 -flux in M-theory

Let \hat{X} denote an arbitrary smooth Calabi-Yau fourfold. In general, G_4 -flux in M-theory can only be defined on such a smooth manifold, that in the context of F-theory typically arises from resolutions of both codimension one singularities from non-Abelian gauge groups or, as in the case considered here, from higher codimension singularities in the presence of a non-trivial Mordell-Weil group.

Then, G_4 -flux in F-theory is defined as G_4 -flux in M-theory with as set of additional F-theoretic restrictions discussed in the next section 4.2. G_4 -flux in M-theory is consistent if it obeys two basic conditions. First, G_4 has to be quantized as [69]

$$G_4 + \frac{c_2(\hat{X})}{2} \in H^4(\hat{X}, \mathbb{Z}),$$
 (4.1)

which depends on the second Chern class $c_2(\hat{X})$ of \hat{X} . In addition, the M2-brane tadpole has to be cancelled [64, 70],¹⁶

$$\frac{\chi(\hat{X})}{24} = n_3 + \frac{1}{2} \int_{\hat{X}} G_4 \wedge G_4 \,, \tag{4.2}$$

where $\chi(\hat{X})$ is the Euler characteristic of \hat{X} and n_3 the number of spacetime-filling M2branes. This tadpole lifts to the D3-brane tadpole in F-theory with n_3 denoting the number of D3-branes.

In addition, one can distinguish G_4 -flux further by decomposing $H^4(\hat{X})$ into its primary vertical and horizontal subspaces $H^4(\hat{X}) = H_V^4(\hat{X}) \oplus H_H^4(\hat{X})$ [63]. As mentioned before, only G_4 -flux in the vertical homology induces 4D chirality as well as gaugings of axions in F-theory and is considered here. A general G_4 -flux in the vertical cohomology $H_V^{(2,2)}(\hat{X})$ than has an expansion as

$$G_4 = m^r \mathcal{C}_r \,, \tag{4.3}$$

where C_r with $r = 1, \ldots, h_V^{(2,2)}(\hat{X})$ denotes an integral basis of $H_V^{(2,2)}(\hat{X})$ and m^r are the flux-quanta with integrality fixed by the quantization condition (4.1).¹⁷ Such a basis can be constructed explicitly, as demonstrated in sections 3.2 and 6.2, for concrete examples.

¹⁵We are grateful to Thomas W. Grimm for explanations and comments on the importance of Θ_{00} .

 $^{^{16}}$ We are working here in the carefully checked conventions of [71], where also comparison with other, inconsistent sign choices in the literature can be found.

¹⁷In general, $c_2(\hat{X})$ has to be decomposed into the integral basis of $H_V^{(2,2)}(\hat{X})$. However, the determination of this basis is very involved and requires more sophisticated techniques that would exceed the scope of this work. We refer to [58] for the application of mirror symmetry to fix the integral basis. See also [72] for a discussion of potential conflicts between the split of $H^4(\hat{X})$ into vertical and horizontal subspace and the choice of an integral basis.

4.2 Deriving conditions on G_4 -flux in F-theory

The additional constraints on the G_4 -flux in F-theory compactifications are most conveniently formulated in the three-dimensional theory obtained after compactification of the 4D $\mathcal{N} = 1$ effective action of F-theory on S^1 . Then we can use the basic duality between three-dimensional F-theory on $\hat{X} \times S^1$ and M-theory on \hat{X} . Consideration of the resolved fourfold \hat{X} means in terms of the 3D $\mathcal{N} = 2$ effective theory obtained in the circle reduction to go to the 3D Coulomb. The corresponding fields acquiring a VEV are the adjoint valued scalars ζ^A , $A = 0, \ldots, h^{(1,1)}(B) + \operatorname{rk}(G) + n_{\mathrm{U}(1)}$, along the Cartan directions of the 4D gauge group $G \times \operatorname{U}(1)^{n_{\mathrm{U}(1)}}$. The zeroth component denotes the scalar in the multiplet of the KK-vector. Then the 3D gauge group is broken in an ordinary Higgs effect to the maximal torus $\operatorname{U}(1)^{h^{(1,1)}(B)+\operatorname{rk}(G)+n_{\mathrm{U}(1)}+1}$ and in addition the charged fermions, in particular those from the 4D massless matter multiplets, obtain a mass (shift) as $m = q_A \cdot \zeta^A$, where q_A denotes the full 3D charge vector.

This Coulomb branch then describes the IR dynamics of the dual M-theory compactification on the fourfold \hat{X} in the supergravity approximation. It is the key point of the following analysis that the matching of the two dual descriptions works only on the level of the quantum effective action after massive degrees of freedom have been integrated out on the F-theory side. As we emphasize in the following, also corrections due to KK-states have to be considered. In fact, consistent conditions on the G_4 -flux in the presence of a non-holomorphic zero section are only obtained if new CS-terms for the KK-vector are taken into account.

The general approach of matching F- and M-theory in 3D initiated in [73] has been exploited recently in [42, 44, 45, 74] to study various aspects of the F-theory effective action. We refer to these references for the background of the following discussion.

4.2.1 F-theory conditions from KK-states corrected CS-terms

First we recall that G_4 -flux in M-theory induces CS-terms for the U(1)-gauge fields A^A on the 3D Coulomb branch that read

$$S_{CS}^{(3)} = -\frac{1}{2} \int \Theta_{AB}^{M} A^{A} \wedge F^{B} \qquad \Theta_{AB}^{M} = \frac{1}{2} \int_{\hat{X}} G_{4} \wedge \omega_{A} \wedge \omega_{B} \,. \tag{4.4}$$

This can be shown by reducing the M-theory three-form C_3 along (1, 1)-forms ω_A on the fourfold \hat{X} that are dual to the basis of divisors D_A , $A = 0, 1, \ldots, h^{(1,1)}(\hat{X}) - 1$. We recall, cf. (2.22) for the case $n_{\mathrm{U}(1)} = 2$, that this basis is given by $h^{(1,1)}(B)$ vertical divisors D_{α} , the divisor $\tilde{S}_P = S_P + \frac{1}{2}[K_B^{-1}]$, see (2.21), associated to the zero section, the Shioda maps $\sigma(\hat{s}_m)$ for a rank $n_{\mathrm{U}(1)}$ Mordell-Weil group and $\mathrm{rk}(G)$ Cartan divisors D_i in the presence of a four-dimensional non-Abelian gauge group G.

Now the strategy to formulate conditions on the G_4 -flux for a valid F-theory compactification is as follows. We require that the CS-levels on the F-theory side, denoted Θ_{AB}^F , have to agree with the M-theory CS-levels, denoted Θ_{AB}^M and given by the flux integral in (4.4),

$$\Theta_{AB}^F \stackrel{!}{=} \Theta_{AB}^M \,. \tag{4.5}$$

This means, whenever a non-vanishing CS-level $\Theta_{AB}^{F'}$ is there on the F-theory side, as a classical CS-level or generated by loops of massive matter on the 3D Coulomb branch, then the same CS-level has to be there on the M-theory side, i.e. it has to be generated by the G_4 -flux. In contrast, when a CS-term is not there on the F-theory effective field theory side, the corresponding flux integral in (4.4) for the same CS-term in M-theory has to vanish. However, the critical point is to allow in the corresponding F-theory loop-computation also for loops with an infinite tower of KK-states. If KK-states are not included, the G_4 -flux is in general over-constrained, in particular in the presence of a non-holomorphic zero section. In addition, certain CS-levels get shifted by KK-states and a consistent match of CS-terms in F- and M-theory is only possible if these corrections are included.

The general form for the correction to the classical CS-level on the F-theory side, denoted by $\Theta_{cl,AB}^{F}$, has been worked out in [75–77]. The correction is one-loop exact with all 3D massive fermions contributing in the loop. Assuming a fermion with charge vector q_A , the loop corrected CS-term takes the simple form

$$\Theta_{AB}^{F} = \Theta_{cl,AB}^{F} + \frac{1}{2} \sum_{\underline{q}} n(\underline{q}) q_{A} q_{B} \operatorname{sign}(q_{A} \zeta^{A}), \qquad (4.6)$$

where $n(\underline{q})$ is the number of fermions with charge vector \underline{q} and the sum runs over all these charge vectors. We note that since real masses can be negative in 3D, the sign-function is non-trivial. We can rewrite this expression further by noting the general form of thee charge vector

$$q_A = (n, q_\alpha, q_i, q_m), \qquad (4.7)$$

where we recall the 3D gauge group $U(1)^{h^{(1,1)}(B)+rk(G)+n_{U(1)}+1}$ on the Coulomb branch and that the charge under the KK-vector A^0 is just the KK-label of KK-states, $q_0 \equiv q_{KK} = n$.

Next, we assume that in a theory obtained by circle reduction from 4D, there are no states with charge under the A^{α} , since these do not correspond to a 4D gauge symmetry, i.e. we assume $q_{\alpha} = 0.^{18}$ We consider in the following the massive charged states obtained from the reduction of four-dimensional massless chiral matter to 3D, along with their KK-states. Denoting their charge vectors as $q_A = (n, 0, q_i, q_m)$ with q_i the Dynkin labels of their non-Abelian representations **R** under *G*, we write the loop-correction (4.6) as

$$\Theta_{00}^{F} = \frac{1}{2} \sum_{\mathbf{R}_{q_m}} \sum_{\underline{q} \in \mathbf{R}_{q_m}} \chi(\mathbf{R}_{q_m}) \sum_{n=-\infty}^{\infty} n^2 \operatorname{sign}(m_{CB} + n \cdot m_{KK}),$$

$$\Theta_{0\Lambda}^{F} = \frac{1}{2} \sum_{\mathbf{R}_{q_m}} \sum_{\underline{q} \in \mathbf{R}_{q_m}} q_{\Lambda} \chi(\mathbf{R}_{q_m}) \sum_{n=-\infty}^{\infty} n \operatorname{sign}(m_{CB} + n \cdot m_{KK}),$$

$$\Theta_{\Sigma\Lambda}^{F} = \frac{1}{2} \sum_{\mathbf{R}_{q_m}} \sum_{\underline{q} \in \mathbf{R}_{q_m}} q_{\Sigma} q_{\Lambda} \chi(\mathbf{R}_{q_m}) \sum_{n=-\infty}^{\infty} \operatorname{sign}(m_{CB} + n \cdot m_{KK}), \qquad (4.8)$$

where we invoked the absence of classical CS-terms. Here we have suppressed a labeling of the KK-charges n by the weights **w** of the representation \mathbf{R}_{q_m} and unified the labels i

¹⁸One might wonder whether these states correspond, on the M-theory side, to M2-branes wrapping curves in the base B, e.g. generated from resolved conifolds in B.

and m as $\Lambda = (i, m)$ and $\chi(\mathbf{R}_{q_m})$ denote the 4D chiralities of chiral matter fields. We have also defined the Coulomb branch mass $m_{CB} = q_{\Sigma} \zeta^{\Sigma}$ and the KK-mass $m_{KK} = \frac{1}{R_{KK}}$ with R_{KK} the radius of the S^1 . The terms (4.8) are, bearing our assumptions on the spectrum of massive fermions in mind, the only CS-terms receiving loop-corrections via (4.6). Other CS-terms are classically generated, either in 4D or in the 3D reduction.

We summarize all CS-terms, the presence of classical terms, the potential correction via loop-effects, their physical interpretation and a related reference in the following,

Θ^F_{AB}	$\Theta^F_{\mathrm{cl},AB}$	loop-corr.	G_4 -condition	interpretation	
Θ_{00}^F	-	yes	-	4D chiralities	
$\Theta_{0\alpha}^F$	yes	-	$\stackrel{!}{=} 0$	S^1 -circle fluxes [60]	
Θ_{0i}^F	-	yes	-	4D chiralities	
Θ_{0m}^F	-	yes	-	4D anomaly cancellation [44]	(
				(for holomorphic \hat{s}_P)	(4.9)
$\Theta^F_{\alpha\beta}$	- (?)	-	$\stackrel{!}{=} 0$	non-geometric flux?	
$\Theta^F_{\alpha i}$	yes	-	$\stackrel{!}{=} 0$	4D gaugings by GUT Cartans	
$\Theta^F_{\alpha m}$	yes	-	-	4D gaugings by $U(1)_m$	
				(4D GS-mechanism)	
$\Theta^F_{\Sigma\Lambda}$	-	yes	-	4D chiralities	

Here we also mention in one column whether the corresponding CS-term is used to impose a condition on the G_4 -flux. Indeed, the effect of CS-terms $\Theta_{0\alpha}^F$ is the induction of circle fluxes along the S^1 in compactification from 4D and, thus, not a physical effect in 4D. Therefore, we impose these CS-terms to vanish. Then, the CS-terms $\Theta_{\alpha\beta}^F$ obstruct the lift back to 4D [73], potentially by non-geometric effects, and are required to vanish. Finally, the CS-terms $\Theta_{\alpha i}$ correspond to 4D gaugings of axions via the maximal torus of the GUT and are thus set to zero. Thus, using the M-/F-theory duality relation (4.5) we formulate the F-theory conditions on the G_4 -flux in terms of 3D CS-levels Θ_{AB} as

G₄-flux conditions:
$$\Theta_{0\alpha} = \Theta_{\alpha\beta} = \Theta_{i\alpha} = 0$$
 (4.10)

We note that these conditions on the G_4 -flux look much weaker than the ones considered in the literature before. We claim that these conditions are the appropriate ones, in particular in cases with a non-holomorphic the zero section \hat{s}_P . In contrast, however, in compactifications with a holomorphic zero section, the vanishing of the CS-levels in (4.10) implies the vanishing of other, dependent CS-levels. We discuss this in the following and contrast it to the situation with a non-holomorphic \hat{s}_P .

4.2.2 KK-corrected 3D CS-terms: field theory computations

In certain cases, the loop-corrections (4.8) can vanish, leading to additional vanishing CS-terms. In particular, for a holomorphic zero-section, the CS-term Θ_{00}^F in field theory

vanishes, and consistently also the geometric CS-term Θ_{00}^M in M-theory. The latter can be seen easily using the conditions (4.10) in the relation

$$\Theta_{00} = \frac{1}{4} K^{\alpha} K^{\beta} \Theta_{\alpha\beta} , \qquad (4.11)$$

which is derived employing the definition (2.21), the intersection property (2.16), and K^{α} introduced in (2.24). In order to see the same from the field theory side and, in general, to compute the loop-corrections to the CS-terms, we first have to evaluate the sign-function in (4.8).

We calculate the sign-function geometrically, recalling from the discussion of section 3.1 that to every weight \mathbf{w} of a representation \mathbf{R}_{q_m} realized in F-theory there is a corresponding curve $c_{\mathbf{w}}$, cf. (3.1). The sign-function in (4.8) is then determined by testing whether the curve $c_{\mathbf{w}}$ associated to a given weight \mathbf{w} with Dynkin labels q_{Λ} and KK-charge n is in the Mori cone $M(\hat{X})$ of effective curves on \hat{X} . We define

$$\operatorname{sign}(q_A \zeta^A) = \begin{cases} 1, & c_{\mathbf{w}} \in M(\hat{X}), \\ -1, & \text{otherwise}. \end{cases}$$
(4.12)

Here the KK-charge $q_0 = n$ of a curve $c_{\mathbf{w}}$ is obtained geometrically as

$$n = c_{\mathbf{w}} \cdot \tilde{S}_P \,. \tag{4.13}$$

In general, a curve is in the Mori cone if it is described by holomorphic equations and an analysis of the geometry allows in general to find all holomorphic curves corresponding to matter, cf. sections 3.1 and 6.1 as well as [51].¹⁹ In the toric context, the relevant parts of the Mori cone can be constructed systematically as recently demonstrated in [55].

Now, in the presence of a holomorphic zero section the sign-function is centered around 0 because no curve $c_{\mathbf{w}}$ has KK-charge. This follows from the simple geometric fact that by definition any rational curve $c_{\mathbf{w}}$ does not intersect S_P , which always goes through the original singular curve, cf. figure 3. This implies that the loop-corrections in (4.8) that are odd in the KK-level n vanish since KK-states with charge -|n| cancel those with charge |n|. In particular, $\Theta_{00}^F = 0$, confirming the geometric result (4.11). In addition, the sum over KK-states in $\Theta_{\Lambda\Sigma}^F$ reduces to

$$\Theta_{\Lambda\Sigma}^{F} = \frac{1}{2} \sum_{\mathbf{R}_{q_m}} \sum_{\underline{q} \in \mathbf{R}_{q_m}} q_{\Sigma} q_{\Lambda} \chi(\mathbf{R}_{q_m}) \operatorname{sign}(m_{CB}), \qquad (4.14)$$

which has been used in [42, 44].

In contrast, the CS-levels $\Theta_{0\Lambda}$ receive an infinite loop-correction that has to be regularized by zeta-function regularization. In [44] this zeta function regularization has been

¹⁹In general, the values of the sign-function on a given representation \mathbf{R}_{q_m} depend on the phase of \hat{X} , respectively, of the 3D gauge theory. See [78] for a detailed discussion of phases structure of Calabi-Yau fourfolds and 3D SU(5) gauge theories. However, it can be shown by a similar argument as in [79] that 4D observables like the chiralities $\chi(\mathbf{R}_{q_m})$ are not expected to depend on the phase.

performed and it was shown that the field theory result for Θ_{0m} agrees with the 4D mixed Abelian-gravitational anomaly,

$$\Theta_{0m}^F = -\frac{1}{12} \sum_{\underline{q}} n(\underline{q}) q_m \,, \tag{4.15}$$

with $n(\underline{q})$ denoting the number of fermions in 4D with U(1)-charge vector \underline{q} . In particular, anomaly cancellation follows then using the M-/F-theory relation (4.5) and the geometric result

$$\Theta_{0\Lambda}^{M} = \frac{1}{2} K^{\alpha} \Theta_{\alpha\Lambda}^{M} , \qquad (4.16)$$

where the right side is immediately identified with the 4D Green-Schwarz term for $\Lambda = m$, cf. section 5.3. Here we used the general result $S_P \cdot D_{\Lambda} = 0$ for a holomorphic zero section. We also infer from (4.16) that the CS-terms Θ_{0i} are set to zero recalling (4.10).

In the presence of a non-holomorphic section \hat{s}_P the only thing that changes on the field theory side is a shift of the sign function in (4.12). It is no longer centered symmetrically around the origin of the sum over KK-labels n. This is geometrically clear because there are now rational curves $c_{\mathbf{w}}$ in the Mori cone that have non-zero intersection with the rational zero section \hat{s}_P , i.e. that have non-zero KK-charge (4.13). These curves have to be located precisely at the loci where the rational zero section is ill-defined and wraps a whole \mathbb{P}^1 in the fiber, which is the original singular curve. Everywhere else in the base B the zero section is only a point on the original singular fiber and does not intersect curves $c_{\mathbf{w}}$.

The effect of the shift of the sign-function is dramatic because now KK-states with positive and negative KK-label |n|, respectively, -|n| do no longer cancels in corrections (4.8) that are odd in n. The infinite parts of the sums over KK-states still cancel, but with a non-zero remainder. Thus, the CS-term Θ_{00}^F that was zero for a holomorphic zero section is now generated by a contribution of a finite number of KK-states. For example, assuming, with k denoting an integer, a sign-function of the form

$$\operatorname{sign}(m_{CB} + n \cdot m_{KK}) = \begin{cases} 1, & \text{for } n \ge -k, \\ -1, & \text{for } n < -k, \end{cases}$$
(4.17)

for only one weight of a single matter multiplet \mathbf{R}_{q_m} the loop induced CS-term in (4.8) reads

$$\Theta_{00}^F = \frac{k(k+1)(2k+1)}{6}\chi(\mathbf{R}_{q_m}).$$
(4.18)

Integers	-k - 1	-k	-k + 1	 0	1	$\sum_{n} n^2 \text{sign}$	
$n^2 \operatorname{sign}(n)$	$-(k+1)^2$	$-k^2$	$-(k-1)^2$	 0	1	0	(4.19)
$n^2 \operatorname{sign}(n+k)$	$-(k+1)^2$	k^2	$(k-1)^2$	 0	1	$\boxed{\frac{k(k+1)(2k+1)}{3}}$	

See e.g. [55] for a formal derivation or the following for an intuitive understanding,

Here the unshifted sum was normalized to zero, then the shifted sum differs only by the amount obtained as the finite sum over the differences between the first and second row in (4.19) for each column.

Thus we see that Θ_{00}^F in (4.18) is directly proportional to one chirality. Consequently, imposing $\Theta_{00}^M = 0$, as done in the literature with holomorphic zero sections, also in the non-holomorphic case would unnecessarily set this chirality to zero. We will see that the loop-correction to Θ_{00}^F precisely takes the form (4.18) for dP_2 -elliptic fibrations, both without and with an additional SU(5) gauge group, cf. sections 5.2 and 6.1, since only one singlet has a non-trivial KK-charge n = 2 and, thus, a shifted sign-function (4.17) with k = -2.

We conclude by mentioning that the CS-terms $\Theta_{\Lambda\Sigma}$ are shifted in a similar way, where as in (4.19) the shift originates from a finite number of KK-states. We note that this shift has to be taken into account when we determine certain 4D chiralitites via 3D CS-terms in sections 5.2 and 7.2. In addition, also the relation (4.15) is modified by a finite correction due to KK-states. Assuming again that only one weight in the representation \mathbf{R}_{q_m} has a non-trivial KK-charge with shifted sign-function (4.17), we obtain the corrected expression

$$\Theta_{0m}^F = -\frac{1}{12} \sum_{\underline{q}} n(\underline{q}) q_m - \frac{k(k+1)}{2} \chi(\mathbf{R}_{q_m}) q_m \,. \tag{4.20}$$

Thus, we see that these CS-levels are no longer directly related to the 4D mixed Abeliangravitational anomaly as in (4.15) with a holomorphic zero section. Geometrically this is also clear since in general $S_P \cdot \sigma(\hat{s}_m) \neq 0$ in compactifications with rational zero sections. Thus, (4.16) does not hold and the cancellation of 4D mixed Abelian-gravitational anomaly is not geometrically implied. However, after subtracting the contribution of KK-states in (4.20), 4D anomaly cancellation requires the remaining piece to again equal $\frac{1}{2}K^{\alpha}\Theta_{\alpha m}$. In other words, we can formulate the relation

$$\frac{1}{4} \int_{\hat{X}} S_P \cdot \sigma(\hat{s}_m) \cdot G_4 = \frac{k(k+1)}{2} q_m \chi(\mathbf{R}_{q_m})$$
(4.21)

on the fourfold \hat{X} as a necessary and sufficient condition for 4D anomaly cancellation. It would be nice to proof this relation purely geometrically. Finally, we note that also the CS-terms Θ_{0i}^F need no longer vanish, since geometrically $\Theta_{0i}^M \neq \frac{1}{2}K^{\alpha}\Theta_{\alpha i}^M$. However, in the geometry \hat{X} at hand, cf. section 6.1, this relation still holds since the singularity of the zero section happens away from the resolved SU(5) singularity and, thus, no SU(5)-matter has non-trivial intersections with S_P . Moreover, we obtain $\tilde{S}_P \cdot D_i = 0$ for all SU(5)-Cartan divisors D_i in section 6.2.

5 G_4 -flux & chiralities on fourfolds with two U(1)s

In this section we analyze chirality-inducing G_4 -flux in F-theory on the fourfolds \hat{X} with dP_2 -elliptic fiber and a non-holomorphic zero section. In section 5.1 we first construct the general G_4 -flux for the fourfold $\pi : \hat{X} \to \mathbb{P}^3$, where we also comment on the general D3brane tadpole. Then in section 5.2 we outline first our general strategy to obtain chiralities on Calabi-Yau fourfolds with higher rank Mordell-Weil group and a rational zero section, before applying it again to the fourfold \hat{X} with $B = \mathbb{P}^3$. Then in section 5.3 we first review anomaly cancellation conditions in general 4D effective field theories. Then we show that the general spectrum obtained for the Calabi-Yau fourfold \hat{X} is anomaly-free for the entire allowed region of figure 2. We conclude in section 5.4 with a concrete toric example to exemplify the general findings.

The case of an un-Higgsed SU(5) GUT-group is considered separately in section 6.

5.1 G_4 -flux on fourfolds with two rational sections

In this section we first show that on a general fourfold \hat{X} with dP_2 -elliptic fiber the D3-brane tadpole can always be solved by adding a sufficient amount of n_{D3} of integral D3-brane charge. Then we construct viable G_4 -flux for F-theory on a general fourfold \hat{X} with base $B = \mathbb{P}^3$. We show that for different elliptic fibrations, i.e. for different values of the integers n_7 , n_9 , the number of independent G_4 -flux quanta changes.

Integral D3-tadpole on general fourfolds with $U(1) \times U(1)$

In the following we prove the necessary condition for D3-tadpole cancellation on \hat{X} , namely that the induced D3-brane charge from the combination of the Euler number and the quantized G_4 -flux in the tadpole equation (4.2) is always integral. The following discussion is an application of the arguments in [57, 64, 69], that immediately carry over to elliptic fibrations \hat{X} with dP_2 -elliptic curve.

To this end, we use the relation (3.28) between the arithmetic genus $\chi_0(\hat{X})$ and the Euler number $\chi(\hat{X})$ to rewrite the tadpole (4.2) as

$$n_{D3} = \frac{\chi(\hat{X})}{24} - \frac{1}{2} \int_{\hat{X}} G_4^2 = -60 + \frac{1}{2} \int_{\hat{X}} \left(\frac{1}{4} c_2(\hat{X})^2 - G_4^2 \right).$$
(5.1)

Here we have also employed that $\chi_0(\hat{X}) = 2$, cf. (3.29). Using the flux quantization condition (4.1) this can be written as

$$n_{D3} = -60 - \frac{1}{2} \int_{\hat{X}} \left(x^2 - x \wedge c_2(\hat{X}) \right) \,, \tag{5.2}$$

where we used $x = G_4 + \frac{1}{2}c_2(\hat{X})$. By flux quantization (4.1) we know that x is integral, i.e. an element in $H^4(\hat{X}, \mathbb{Z})$. This implies by Wu's theorem that $x^2 \cong c_2(\hat{X}) \wedge x \mod 2$ [69], so that the integrand in (5.2) is divisible by two. Thus, the number n_{D3} of D3-branes is integral for every elliptically Calabi-Yau fourfold \hat{X} with general elliptic fiber in dP_2 .

The G_4 -flux on \hat{X} with $B = \mathbb{P}^3$

Next we explicitly determine the G_4 -flux on \hat{X} for a general elliptic fibration over the base $B = \mathbb{P}^3$, i.e. for all integers n_7 , n_9 in the allowed region in figure 2.

We begin by expanding the G_4 -flux according to (4.3) into the basis of $H_V^{(2,2)}(\hat{X})$ determined in (3.34),

$$G_4 = a_1 H_B^2 + a_2 H_B \cdot S_P + a_3 H_B \cdot \sigma(\hat{s}_Q) + a_4 H_B \cdot \sigma(\hat{s}_R) + a_5 S_P^2 , \qquad (5.3)$$

for general coefficients a_i , where as before the application of Poincaré duality is understood. Then we calculate the CS-levels (4.4) employing the intersection ring (B.4) in the basis of


Figure 4. The region of allowed values for (n_7, n_9) from figure 2. On the entire region, there are two conditions on the flux. In the interior of this region, (5.4) holds. On the red and the blue boundary, there are only four independent (2, 2)-forms in the expansion (5.3). On the blue boundary, \hat{s}_P is holomorphic.

divisors (3.31), but with $\tilde{S}_P = S_P + 2H_B$ as defined in (2.21) replacing the zero section S_P . The generic solution is a three-parameter family of G_4 -flux given by

$$G_4 = a_5 n_9 \left(4 - n_7 + n_9\right) H_B^2 + 4a_5 H_B \cdot S_P + a_3 H_B \cdot \sigma(\hat{s}_Q) + a_4 H_B \cdot \sigma(\hat{s}_R) + a_5 S_P^2, \quad (5.4)$$

which is valid for all values of n_7 and n_9 in the allowed region figure 2.

This generic three-parameter solution for the G_4 -flux is expected since there are generically five different surfaces in (5.3) and two independent conditions (4.10), namely $\Theta_{0\alpha} = \Theta_{\alpha\beta} = 0$ with α , $\beta = 1$. However, the situation becomes more interesting at special values for (n_7, n_9) . First, we recall that for (n_7, n_9) on the boundary of the region in figure 2, the dimensionality of $H_V^{(2,2)}(\hat{X})$ decreases to 4. The surface S_P^2 becomes linearly dependent in homology on the four other surfaces, as noted below (3.36). At the same time, the number of independent conditions on the G_4 -flux remains two. Thus, we find two independent G_4 -fluxes on the boundary. We have depicted this situation in figure 3

In all these cases we can obtain the expression for the G_4 -flux by specializing (5.4). This ensures that all quantities, in particular the chiralities of 4D charged matter, that are calculated from the most general G_4 -flux specialize correctly for non-generic values of (n_7, n_9) . We discuss this specialization at the end of this subsection, but note that the reader may want to skip these details on a first read and proceed with the chirality formulas in section 5.2.

Before delving into the details of this analysis, we evaluate the D3-brane tadpole (4.2) for the Calabi-Yau fourfold $\hat{X} \to \mathbb{P}^3$ and the G_4 -flux (5.4). First, we calculate the individual terms in the D3-brane tadpole. We obtain the Euler number for \hat{X} with $B = \mathbb{P}^3$ from the general formula (3.24) as

$$\chi(\hat{X}) = 4896 + 3\left[-256(n_7 + n_9) + 4(8n_7^2 + n_7n_9 + 8n_9^2) - n_7n_9(n_7 + n_9)\right], \quad (5.5)$$

where we employed $c_1(\mathbb{P}^3) = 4H_B$, $c_2(\mathbb{P}^3) = 6H_B^2$. We note that this expression is manifestly positive due to the bounds on n_7 and n_9 in (2.27), respectively, figure 4. Then we

calculate the contribution of the G_4 -flux (5.4) to the D3-tadpole as

$$\int_{\hat{X}} G_4^2 = -8a_3^2 + 2a_3a_4(n_7 - n_9 - 4) + a_5^2n_9(2n_9 - n_7)(n_7 - n_9 - 4) - 2a_4^2(4 + n_9) + 2a_5(a_3 + 2a_4)n_9(4 - n_7 + n_9).$$
(5.6)

Finally, for the purpose of G_4 -flux quantization via (4.1) we note the following expression for the second Chern-class of \hat{X} ,

$$c_2(\hat{X}) = (182 + 3n_9(n_7 - n_9 - 4))H_B^2 + 28H_B \cdot S_P + 2(8 - n_9)H_B \cdot \sigma(\hat{s}_Q) + (16 + 2n_7 - 3n_9)H_B \cdot \sigma(\hat{s}_R) - 5S_P^2, \qquad (5.7)$$

where we expanded the general expression (3.22) for a better comparison with the G_4 -flux (5.4) in the basis (3.34) of the vertical cohomology. Also for completeness we calculate the square of the second Chern-class from the general formula (A.15) as

$$\int_{\hat{X}} c_2(\hat{X})^2 = 2112 - 256(n_7 + n_9) + 4(8n_7^2 + n_7n_9 + 8n_9^2) - n_7n_9(n_7 + n_9).$$
(5.8)

Comparing this with the Euler number (5.5) we reconfirm the general relation (3.28) between the arithmetic genus and the Euler number of a Calabi-Yau fourfold.

Finally, we conclude by a discussion of (5.4) for non-generic values of (n_7, n_9) . Along each component of the boundary we have to use the homology relations between S_P^2 and the four remaining basis elements (3.34) of $H_V^{(2,2)}(\hat{X})$. The relevant relations are (B.8), (B.10), (B.11) and (B.12) worked out in the appendix. In all these cases the general formula (5.4) for the G_4 -flux can be shown to reduce to the two-parameter

$$G_4 = \tilde{a}_3 H_B \cdot \sigma(\hat{s}_Q) + \tilde{a}_4 H_B \cdot \sigma(\hat{s}_R) \,. \tag{5.9}$$

Here the coefficients \tilde{a}_3 and \tilde{a}_4 are the following linear combination of a_3 , a_4 and a_5 on the three boundary components,

$$\{n_7 = 0\}: \ \tilde{a}_3 = a_3, \ \tilde{a}_4 = a_4 - n_9 a_5, \quad \{n_9 = n_7 + 4\}: \ \tilde{a}_3 = a_3 - n_7 a_5, \ \tilde{a}_4 = a_4 - 4a_5, \{n_9 = 12 - n_7\}: \ \tilde{a}_3 = a_3 + (n_7 - 8)a_5, \ \tilde{a}_4 = a_4 + (n_7 - 8)a_5, \{n_9 = n_7 - 4\} \cup \{n_9 = 0\}: \ \tilde{a}_3 = a_3, \ \tilde{a}_4 = a_4.$$
 (5.10)

It is satisfying to see that the parameters \tilde{a}_3 , \tilde{a}_4 are continuous at the intersection points of boundary components. We note that the blue boundaries in figure 4, $n_9 = 0$ and $n_9 = n_7 - 4$, are special because \hat{s}_P is holomorphic.

We conclude with one remark on the form of the G_4 -flux obtained here. In some cases, Type IIB seven-brane gauge fluxes $F^{(1)}$, $F^{(2)}$ of two single D7-branes can be lifted into F-theory by considering a G_4 -flux of the type

$$G_4 = F^{(1)} \cdot \sigma(\hat{s}_Q) + F^{(2)} \cdot \sigma(\hat{s}_R) \,. \tag{5.11}$$

We note that the flux (5.9) is precisely of this form. In contrast, the interpretation of the general G_4 -flux in (5.4) in terms of Type IIB quantities, if possible, is less clear and would be very interesting to investigate.

5.2 4D chiralities from matter surfaces & 3D CS-terms

Finally we are prepared to calculate the chirality of matter in four-dimensional F-theory compactifications on \hat{X} with U(1)×U(1) gauge group. We demonstrate that the chirality of all six matter representations discussed in section 3 can be determined uniquely. Half of the chiralities are determined by integration of the G_4 -flux over matter surfaces $C_{\mathbf{R}}^{\mathbf{w}}$, the other half from the 3D CS-terms $\Theta_{\Sigma\Lambda}$. As we see explicitly, the 3D CS-terms $\Theta_{\Lambda\Sigma}$ are not sufficient to fix all chiralities, in contrast to earlier works with a holomorphic zero section. However, supplemented by new CS-terms present only for non-holomorphic zero sections, see section 4.2, the chiralities can be obtained also exclusively via 3D CS-terms. We emphasize that this analysis requires the inclusion of KK-charges for all curves $c_{\mathbf{w}}$. In the fourfold \hat{X} only the representation $\mathbf{1}_{(-1,-2)}$ has non-trivial KK-charge $q_{KK} = 2$.

We outline the general strategy to obtain 4D chiralities in section 5.2.1 before we determine them explicitly in section 5.2.2 The concrete calculations are performed for the Calabi-Yau fourfold \hat{X} with base $B = \mathbb{P}^3$ for the entire allowed region in figure 2.

5.2.1 General strategy to determine 4D chiralities

We begin our discussion by recalling how to extract the chiral index of 4D matter in a representation **R** under a gauge group G in a global F-theory compactification on an arbitrary resolved elliptically fibered Calabi-Yau fourfold \hat{X} .

As explained in section 3, each matter representation **R** possesses an associated ruled surface, the matter surface $C_{\mathbf{R}}^{\mathbf{w}}$, where **w** denotes a weight of **R**. Then, the chiral index of charged matter in this representation **R** is given by the flux integral²⁰ [1, 21, 40, 80]

$$\chi(\mathbf{R}) = -\frac{1}{4} \int_{\mathcal{C}_{\mathbf{R}}^{\mathbf{w}}} G_4 \,, \tag{5.12}$$

where the G_4 -flux is a quantized M-theory flux subject to the conditions (4.10). It is important to note that these conditions, more precisely the conditions $\Theta_{i\beta} = 0$, imply that the integral (5.12) is independent on the choice of a particular weight \mathbf{w} , since different weights \mathbf{w} , \mathbf{w}' of the same representation \mathbf{R} are related by a root α_i , $\mathbf{w} - \mathbf{w}' = \alpha_i$.

In cases where all the matter surfaces $\mathcal{C}_{\mathbb{R}}^{\mathbf{w}}$ are known, the integral (5.12) is the most direct and geometric way of calculating the chirality $\chi(\mathbf{R})$. However, for the fourfold \hat{X} at hand, but also for the fourfold $\hat{X}_{SU(5)}$ with an additional resolved SU(5) singularity constructed in section 6, the homology class of all matter surfaces $\mathcal{C}_{\mathbb{R}}^{\mathbf{w}}$ is not known. Fortunately, since we know the expected matter spectrum from the geometric analysis there is to determine the missing chiralities $\chi(\mathbf{R})$ indirectly via 3D CS-terms.

For this purpose we recall the three-dimensional M-/F-duality explained in section 4.2. There we have seen that the 3D CS-terms (4.4) for the U(1)-vector fields $A^{\Lambda} = (A^i, A^m)$ on the Coulomb branch can be calculated in two independent ways. On the one hand, after solving the G_4 -flux condition (4.10), one can calculate all CS-terms $\Theta^M_{\Lambda\Sigma}$ on the M-theory side by evaluating the classical flux integrals in (4.4). On the other hand, the CS-levels

²⁰The factor $-\frac{1}{4}$ has been introduced to be consistency with the conventions of [44]. It can be reabsorbed into the G_4 -flux.

 Θ_{AB}^{F} on the F-theory side are generated and corrected at one-loop from integrating out charged matter and, thus, contain, as shown in section 4.2, cf. (4.8), the 4D chiral indices $\chi(\mathbf{R}_{q_m})$ of all 4D matter. Since we know which representations \mathbf{R}_{q_m} are there from our geometric analysis in section 3.1 and the later section 6.1 for the SU(5) case, we can solve the matching condition (4.5) for the chiralities $\chi(\mathbf{R}_{q_m})$.

We note that in both fourfolds \hat{X} and $\hat{X}_{SU(5)}$ we know the homology classes for a number of matter surfaces $C_{\mathbb{R}}^{\mathbf{w}}$, but not for all. For these matter surfaces, we can evaluate the index (5.12) directly. The remaining chiralities can be fixed, as demonstrated in sections 5.2 and 7.2 by the matching of the CS-terms $\Theta_{\Lambda\Sigma}$, that are given on the F-theory side by the loop-expression given in (4.8). However, we can also obtain *all* 4D chiralities by taking into account the CS-terms Θ_{00} and Θ_{0m} in (4.8), respectively, in (4.18) and (4.20). As we will see concretely, the obtained results agree with the direct computations via (5.12), confirming the validity of application of the M-/F-theory duality (4.5) to the determination of 4D chiralities.

We conclude by noting that the matching (4.5) of CS-terms $\Theta_{\Sigma\Lambda}$ has been used in [42, 44, 45] to calculate successfully the chiralities of F-theory compactifications with SU(5) and SU(5)×U(1) gauge symmetry. In this work, however, we encounter the novel situation that the conditions arising from $\Theta_{\Sigma\Lambda}$ are not sufficient to determine the chiralities of the full spectrum with both U(1)×U(1) as well as SU(5)×U(1)×U(1) gauge symmetry as outlined in the following and in section 6. The reason for this is precisely the existence of a non-holomorphic zero section. However, either supplemented by the chiralities that can be directly determined by the flux integrals (5.12) or by the CS-terms Θ_{00} and Θ_{0m} , we obtain all chiralities. For an application of the latter resolution, see also the analogous analysis of [55] in 6D.

One might wonder whether the CS-terms always provide enough conditions to solve for the chiralities of a known F-theory spectrum. By a simple counting argument assuming a rank $n_{\mathrm{U}(1)}$ Mordell-Weil group, we enumerate the number of conditions arising from the matching of these CS-terms as

$$\#(\text{CS-terms}) = \frac{(n_{\mathrm{U}(1)} + 2)(n_{\mathrm{U}(1)} + 1)}{2} \,. \tag{5.13}$$

If all these conditions remain independent, the CS-terms might indeed be sufficient to determine the chiralities in F-theory compactifications with more U(1)-symmetries. This can be seen by a similar estimate on the growth of the number of different representations as a function of the rank $n_{\rm U(1)}$ of the Mordell-Weil group.

5.2.2 Chiralities on \hat{X} with $B = \mathbb{P}^3$: matter surfaces & CS-terms

In the following we calculate for the first time 4D chiralities $\chi(\mathbf{R})$ of an F-theory compactification on a general elliptically fibered Calabi-Yau fourfold \hat{X} with rank two Mordell-Weil group and with the full matter spectrum analyzed in section 3. The following is a direct extension of the six-dimensional analysis in [51] to a 4D chiral theory.

As mentioned before, the 4D chirality of a given representation **R** is computed by the flux integrals (5.12) given the G_4 -flux and the corresponding matter surfaces $\mathcal{C}_{\mathbf{R}}^{\mathbf{w}}$. The matter surfaces for the matter fields $\mathbf{1}_{(q_1,q_2)}$ with the three different $\mathrm{U}(1)^2$ -charges $(q_1,q_2) = (-1,1), (0,2), (-1,-2)$ have been determined in (3.7). Using the general G_4 -flux (5.4) on $\hat{X} \to \mathbb{P}^3$ we obtain the following chiralities

$$\chi(\mathbf{1}_{(-1,1)}) = -\frac{1}{4} \int_{\mathcal{C}_{(-1,1)}} G_4 = \frac{1}{4} (a_3 - a_4) n_7 (4 + n_7 - n_9) ,$$

$$\chi(\mathbf{1}_{(0,2)}) = -\frac{1}{4} \int_{\mathcal{C}_{(0,2)}} G_4 = \frac{1}{4} n_7 n_9 (-2a_4 + a_5(4 - n_7 + n_9)) ,$$

$$\chi(\mathbf{1}_{(-1,-2)}) = -\frac{1}{4} \int_{\mathcal{C}_{(-1,-2)}} G_4 = \frac{1}{4} n_9 (4 - n_7 + n_9) (a_3 + 2a_4 + a_5(n_7 - 2n_9)) . \quad (5.14)$$

In order to evaluate the involved intersections we have made use of the topological metric $\eta^{(2)}$ on the cohomology $H_V^{(2,2)}(\hat{X})$ calculated in (3.36).

In order to obtain the chiralities for the matter fields $\mathbf{1}_{(q_1,q_2)}$ with charges $(q_1,q_2) = (1,0), (0,1), (1,1)$, we have to use the 3D CS-levels and the matching condition (4.5). For this purpose we have to determine the KK-charges of all six matter representations on \hat{X} . As mentioned before, a non-trivial KK-charge is calculate by the intersection (4.13) of the curve $c_{\mathbf{w}}$ and the zero section \tilde{S}_P . Such a non-trivial intersection can only occur at loci, where the zero section is ill-defined and wraps fiber components. This is precisely the case at the loci $s_8 = s_9 = 0$, where $\mathbf{1}_{(-1,-2)}$ is supported. Since the fiber is an I_2 -fiber over all matter loci, we obtain a KK-charge

$$q_{KK}(\mathbf{1}_{(-1,-2)}) = c_{(-1,-2)} \cdot \tilde{S}_P = 2, \qquad (5.15)$$

and zero for all other matter representations. We note that (5.15) implies that the sign-function (4.12) is given by the shifted sign-function (4.17) with k = -2 whereas the other matter fields retain a point-symmetric sign-function.

We can immediately cross-check this result using the field theory computations of sections 4.2.2. We first calculate the CS-level Θ_{00}^F for the 3D KK-vector on the field theory side. Using the general expression (4.18) for k = -2 we obtain

$$\Theta_{00}^F = -\chi(\mathbf{1}_{(-1,-2)}), \qquad (5.16)$$

with the chirality $\chi(\mathbf{1}_{(-1,-2)})$ determined in (5.14). We readily calculate the corresponding flux integral Θ_{00}^{M} via (4.4) and immediately reproduce (5.16). Next we check the relation (4.20) for the CS-level Θ_{0m} , respectively, (4.21). Again we start with the field theory result for the right hand side of (4.21) which requires

$$\frac{1}{4} \int_{\hat{X}} S_P \cdot \sigma(\hat{s}_Q) \cdot G_4 \stackrel{!}{=} -\chi(\mathbf{1}_{(-1,-2)}), \qquad \frac{1}{4} \int_{\hat{X}} S_P \cdot \sigma(\hat{s}_R) \cdot G_4 \stackrel{!}{=} -2\chi(\mathbf{1}_{(-1,-2)}).$$
(5.17)

We confirm this relation easily by calculating the intersections on the left hand side directly from the G_4 -flux (5.4). Thus, as we have just demonstrated the results for the chiralities in (5.14) obtained from the matter surface integrals can be employed as an independently check of the field theory expressions for the CS-levels in 4.2.2 and the M-/F-theory duality relation (4.5). Next we proceed with the computation of the other CS-levels Θ_{mn} , m, n = 1, 2, for the two U(1) gauge fields A^m corresponding to the divisors $\sigma(\hat{s}_Q)$, respectively, $\sigma(\hat{s}_R)$. Beginning with the M-theory expressions, we obtain using the Shioda maps (3.35), (4.4) and the general G_4 -flux (5.4),

$$\Theta_{11}^{M} = \frac{1}{2} \left[a_{3}(96 - n_{7}(4 + n_{7}) + n_{9}(4 + n_{9})) + a_{5}(4 - n_{7} + n_{9})n_{9}(n_{7} - 12 - 2n_{9}) \right) + a_{4}(n_{7}^{2} + 2(4 + n_{9})(6 + n_{9}) - n_{7}(8 + 3n_{9})) \right] ,$$

$$\Theta_{12}^{M} = \frac{1}{2} \left[a_{3}(n_{7}^{2} + 2(4 + n_{9})(6 + n_{9}) - n_{7}(8 + 3n_{9})) + (4 - n_{7} + n_{9})(a_{5}(-12 + 3n_{7} - 5n_{9})n_{9} + a_{4}(12 + 5n_{9})) \right] ,$$

$$\Theta_{22}^{M} = \frac{1}{2} \left[96a_{4} + a_{3}(4 - n_{7} + n_{9})(12 + 5n_{9}) + 2a_{4}n_{9}(32 - 4n_{7} + 5n_{9}) - 2a_{5}n_{9}(4 - n_{7} + n_{9})(12 - 2n_{7} + 5n_{9}) \right] ,$$

(5.18)

with $\Theta_{21}^M = \Theta_{12}^M$. Here we have used the quartic intersections (B.4) in appendix B.

Then we compute the one-loop CS-terms Θ_{mn}^F in (4.8) on the F-theory side. For the matter spectrum at hand, cf. (3.3), we obtain

$$\Theta_{11}^{F} = \frac{1}{2} (\chi(\mathbf{1}_{(1,0)}) + \chi(\mathbf{1}_{(1,1)}) + \chi(\mathbf{1}_{(-1,1)}) - 3\chi(\mathbf{1}_{(-1,-2)}))),$$

$$\Theta_{12}^{F} = \frac{1}{2} (\chi(\mathbf{1}_{(1,1)}) - \chi(\mathbf{1}_{(-1,1)}) - 6\chi(\mathbf{1}_{(-1,-2)})),$$

$$\Theta_{22} = \frac{1}{2} (\chi(\mathbf{1}_{(0,1)}) + \chi(\mathbf{1}_{(1,1)}) + \chi(\mathbf{1}_{(-1,1)}) + 4(\chi(\mathbf{1}_{(0,2)}) - 3\chi(\mathbf{1}_{(-1,-2)}))). \quad (5.19)$$

We note that the factor of -3 in front of $\chi(\mathbf{1}_{(-1,-2)})$ occurs due to the shifted sign-function (4.17) with k = -2. Indeed, the relevant sum over KK-states in this case yields

$$\sum_{n} \operatorname{sign}(n+k) = -3.$$
(5.20)

We note that the matching of 3D CS-terms (4.5) only using the Θ_{mn} yields three conditions for the six a priori unknown chiralities in (5.19). Thus, it is impossible to determine the full matter spectrum from these CS-terms alone, which is in contrast to earlier studies in the literature with holomorphic zero sections. We emphasize that the same complication arises in the case of SU(5)×U(1)² gauge group considered in section 6, since the representations in the SU(5)-singlet sector will be identical to the six representations considered here. However, we can either use the results (5.14) from the integral of the G_4 flux over the matter surfaces or have to incorporate the CS-terms Θ_{00} and Θ_{0m} to obtain three further conditions and to fix all six chiralities.

Consequently, taking into account the results (5.14), we apply the matching condition (4.5) for the dual CS-terms (5.18) and (5.19) we obtain the remaining three chiralities as

$$\chi(\mathbf{1}_{(1,0)}) = \frac{1}{4} \left[a_5 n_7 n_9 \left(4 - n_7 + n_9 \right) + a_3 \left(2n_7^2 - (12 - n_9) \left(8 - n_9 \right) - n_7 \left(16 + n_9 \right) \right) \right], (5.21)$$

$$\chi(\mathbf{1}_{(0,1)}) = \frac{1}{2} \left[a_5 n_9 \left(4 - n_7 + n_9 \right) \left(12 - n_9 \right) - a_4 \left(n_7 \left(8 - n_7 \right) + (12 - n_9) \left(4 + n_9 \right) \right) \right],$$

$$\chi(\mathbf{1}_{(1,1)}) = \frac{1}{4} \left[2a_5 n_9 \left(4 - n_7 + n_9 \right) \left(12 - n_9 \right) - \left(a_3 + a_4 \right) \left(n_7^2 + n_7 \left(n_9 - 20 \right) + 2 \left(12 - n_9 \right) \left(4 + n_9 \right) \right) \right].$$

We conclude by noting that the chiralities we obtain include factors of $\frac{1}{2}$ and $\frac{1}{4}$. These factors should disappear once the G_4 -flux has been quantized appropriately according to (4.1). The precise quantization will, however, depend on the values of n_7 , n_9 and has to be done in a case by case analysis. In addition, for concrete n_7 , n_9 the factors in the numerator (5.14) and (5.21) have different divisibility properties and can cancel the denominators. Therefore, in order to not obscure these cancellation effects, we keep the normalization in (5.14), (5.21) and the mild fractions of $\frac{1}{2}$ and $\frac{1}{4}$. In concrete toric examples, they can be cancelled appropriately.

5.3 4D anomaly cancellation: F-theory with multiple U(1)s

Finally, after having calculated the matter spectrum of an F-theory compactification, we check consistency of the obtained low-energy effective physics. One check is anomaly cancellation. In the following we introduce the necessary quantities to analyze anomalies and refer to [44] for more details on anomalies in general and, in particular, in F-theory. Then we use these techniques to show that the general spectrum found in section 5.2 for F-theory compactifications on $\hat{X} \to \mathbf{P}^3$ with gauge group U(1)² is anomaly-free.

5.3.1 4D anomaly cancellation: general discussion

In the following we assume a 4D gauge group G with a single non-Abelian factors, see for example [44] for the general case, and with a number $n_{U(1)}$ of Abelian factors. Charges are summarized by a charge vector $\underline{q} = (q_m)$ and the number of left Weyl fermions in a matter representation $\mathbf{R}_{\underline{q}}$ are denoted in general by $n(\mathbf{R}_{\underline{q}})$. The number of fermions in a representation \mathbf{R} irrespective of their U(1)-charges is denoted by $n(\mathbf{R})$, whereas $n(\underline{q})$ indicates the number of fermions with charges \underline{q} regardless of their non-Abelian representation. All these numbers can be expressed in terms of the chiralities $\chi(\mathbf{R}_q)$.

The conditions for 4D anomaly cancellation via the generalized Green-Schwarz (GS) mechanism [81, 82] yield a system of linear equations involving the spectrum of the theory as well as parameters encoding the GS-counter terms and the gaugings of axions. These conditions for cancellation of 4D purely non-Abelian, purely Abelian, mixed Abelian-non-Abelian and mixed Abelian-gravitational anomalies read, in the same order,

purely non-Abelian anomaly :
$$\sum_{\mathbf{R}} n(\mathbf{R}) V(\mathbf{R}) = 0,$$
$$U(1)_k \times U(1)_l \times U(1)_m \text{-anomaly} : \frac{1}{6} \sum_{\underline{q}} n(\underline{q}) q_{(m} q_n q_k) = \frac{1}{4} b^{\alpha}_{(mn} \Theta_{k)\alpha},$$
$$U(1)_m \text{-non-Abelian anomaly} : \frac{1}{2} \sum_{\mathbf{R}} \sum_{\underline{q}} n(\mathbf{R}_{\underline{q}}) U(\mathbf{R}) q_m = \frac{1}{4\lambda} b^{\alpha} \Theta_{\alpha m},$$
$$U(1)_m \text{-gravitational anomaly} : \frac{1}{48} \sum_{q} n(\underline{q}) q_m = -\frac{1}{16} a^{\alpha} \Theta_{m\alpha}.$$
(5.22)

Here $V(\mathbf{R})$ and $U(\mathbf{R})$ denote group theoretical constants that arise when rewriting traces in the representation \mathbf{R} as traces in the fundamental representation \mathbf{f} . Letting F denote the non-Abelian field strength, we set

$$\operatorname{tr}_{\mathbf{R}}F^{3} = V(\mathbf{R})\operatorname{tr}_{\mathbf{f}}F^{3}, \qquad \operatorname{tr}_{\mathbf{R}}F^{2} = \operatorname{U}(\mathbf{R})\operatorname{tr}_{\mathbf{f}}F^{2}.$$
(5.23)

Similarly, we note that $\lambda = 2c_G/V(\mathbf{adj})$ with c_G the dual Coxeter number of G and \mathbf{adj} its adjoint representation. Note that $\lambda = 1$ for $G = \mathrm{SU}(N)$, the case of interest here.

There are some remarks in order. The left hand sides of (5.22) are the actual one-loop triangle anomalies of the field theory. These are determined entirely by the spectrum. The right hand sided of (5.22) are the contribution of the anomalous tree-level GS-counter terms. In an F-theory compactification on a Calabi-Yau fourfold \hat{X} over a base B they are identified as follows [44, 83]

$$\Theta_{m\alpha} = \Theta_{m\alpha}^M, \quad b_{mn}^{\alpha} = -\pi(\sigma(\hat{s}_m) \cdot \sigma(\hat{s}_n)) \cdot \Sigma_b^{\alpha}, \quad b^{\alpha} = S_G^b \cdot \Sigma_b^{\alpha}, \quad a^{\alpha} = K^{\alpha}.$$
(5.24)

Here the CS-terms Θ_{AB}^{M} are defined in (4.4), the Néron-Tate height pairing has been introduced in (2.18) for two sections and can be straightforwardly generalized to an arbitrary number of sections, Σ_{b}^{α} denotes a basis of curves defined in (2.19), S_{G}^{b} is the divisor in the base *B* supporting the gauge symmetry *G*, cf. 2.23, and K^{α} is the coefficient (2.24) in the expansion of K_{B} .

5.3.2 Anomaly cancellation in 4D F-theory with a U(1)²-sector: $B = \mathbb{P}^3$

The spectrum of the F-theory compactification to four dimensions on the fourfold $\hat{X} \to \mathbb{P}^3$ has been calculated in (5.14) and (5.21). The various anomalies for this spectrum read

$$\begin{aligned} A_{111}^{\mathrm{U}(1)} &: 2[a_3(n_9 - n_7 - 12) + a_5n_7 (4 - n_7 + n_9)], \\ A_{222}^{\mathrm{U}(1)} &: n_7 (a_3 + a_5 (4 - n_9)) (4 + n_9) , \\ A_{112}^{\mathrm{U}(1)} &: \frac{1}{6} \left[a_5n_7 \left(48 + n_7^2 + n_9^2 - 2n_7 (4 + n_9) \right) + a_3 \left(n_7^2 - 2n_7 (n_9 - 8) + (n_9 - 12) (4 + n_9) \right) \right], \\ A_{122}^{\mathrm{U}(1)} &: \frac{1}{6} \left(a_5n_7 (n_7 - n_9 - 4) (n_9 - 12) + a_3 \left(-2n_7^2 + n_7 (4 + n_9) + (n_9 - 12) (4 + n_9) \right) \right), \\ A_{1}^{\mathrm{U}(1)\text{-grav}} &: a_3 (n_9 - n_7 - 12) + a_5n_7 (4 - n_7 + n_9) , \\ A_{2}^{\mathrm{U}(1)\text{-grav}} &: 2n_7 (a_3 + a_5 (4 - n_9)) , \end{aligned}$$

$$(5.25)$$

where we brought all numerical factors in (5.22) to the left hand sides. We denoted by $A_{klm}^{\mathrm{U}(1)}$ the $\mathrm{U}(1)_k \times \mathrm{U}(1)_l \times \mathrm{U}(1)_m$ - and by $A_k^{\mathrm{U}(1)-\mathrm{grav}}$ the U(1)-gravitational anomalies. Clearly, there are no non-Abelian anomalies due to the absence of a non-Abelian group G, however, see section 6 for the inclusion of $G = \mathrm{SU}(5)$. Thus, since the anomalies (5.25) are all non-vanishing, a non-trivial GS-mechanism is required for consistency of the theory.

Therefore, all that is left to check cancellation to prove anomaly cancellation of these F-theory compactifications is to calculate the quantities on the right of (5.24) that encode the GS-mechanism. First, we obtain

$$\Theta_{\alpha m} = \left(\frac{1}{4}[(-12 - n_7 + n_9)a_3 + n_7(4 - n_7 + n_9)a_5], \frac{1}{2}n_7[a_3 + (4 - n_9)a_5]\right)_m,$$

$$b_{mn}^{\alpha} = \begin{pmatrix} 8 & 4 - n_7 + n_9 \\ 4 - n_7 + n_9 & 8 + 2n_9 \end{pmatrix}, \qquad a^{\alpha} = -4.$$
(5.26)

where we evaluated the CS-terms (4.4) for the flux (5.4), computed the height pairing (2.18) and (2.24) for $K_{\mathbb{P}^3} = \mathcal{O}_{\mathbb{P}^3}(-4)$. We note that the index $\alpha = 1$ since the only vertical divisor is the hyperplane H_B and m = 1, 2 for the two rational sections \hat{s}_Q , \hat{s}_R . Equipped with the coefficients in (5.26) we finally calculate the GS-terms on the right side of (5.22), which precisely yield (5.25). Thus, we see that all anomalies are cancelled by the GS-mechanism.

We conclude note that the spectrum calculated in section 5.2 is the uniquely determined anomaly-free spectrum if only one chirality $\chi(\mathbf{1}_{(q_1,q_2)})$ is calculated independently. Anomaly cancellation is not sufficient to fix the spectrum completely, since the U(1)³₁-anomaly is proportional to the U(1)₁-gravitational anomaly as is evident from (5.25).

5.4 A toric example

We conclude the analysis of F-theory compactifications with $U(1) \times U(1)$ gauge group with explicit toric constructions of the fourfolds \hat{X} with dP_2 -elliptic curve \mathcal{E} . We focus on elliptically fibered Calabi-Yau fourfolds $\hat{X} \to \mathbb{P}^3$ over \mathbb{P}^3 .

First we note that we can construct using the algorithm of appendix G toric fourfolds realizing a wide variety of values for n_7 and n_9 inside the allowed region in figure 2 of dP_2 fibrations over \mathbb{P}^3 with generic Calabi-Yau hypersurface \hat{X} . In the following, we present the case $n_7 = 4$ and $n_9 = 5$ to illustrate key ideas of the toric construction.

variable	vertices				\mathbf{C}^* -action			1	divisor class		
z_0	1	1	1	-1	-1	1	0	0	0	H_B	
z_1	-1	0	0	0	0	1	0	0	0	H_B	
z_3	0	-1	0	0	0	1	0	0	0	H_B	
z_2	0	0	-1	0	0	1	0	0	0	H_B	(5.27)
u	0	0	0	1	0	0	1	1	-1	$H - E_1 - E_2 + H_B$	
v	0	0	0	0	1	0	1	0	0	$H - E_2 + H_B$	
w	0	0	0	-1	-1	-1	0	1	0	$H-E_1$	
e_1	0	0	0	0	-1	0	0	-1	1	E_1	
e_2	0	0	0	1	1	0	-1	0	1	E_2	

The toric reflexive polytope defining the toric variety $dP_2(4,5)$ takes the form

Here the first column denoted the projective coordinates on the toric variety $dP_2(4,5)$, the next five columns are the vertices of the polytope, that are given as five-dimensional row vectors. Next we displayed the four \mathbb{C}^* -actions of $dP_2(4,5)$ along with the divisor classes in the last column.

Comparing the divisor classes in (5.27) with the general assignments in (2.10) we immediately confirm that the above polytope indeed describes a fibration with divisors $S_7 = 4H_B$ and $S_9 = 5H_B$. This can also be checked explicitly by torically calculating the intersections (2.15) in the toric variety associated to the single star triangulation of the polytope specified in (5.27). The Euler number, as well as the Hodge numbers, of the fourfold \hat{X} in $dP_2(4, 5)$ read

$$\chi(\hat{X}) = 1620, \qquad h^{(1,1)}(\hat{X}) = 4, (0), \qquad h^{(2,2)}_V(\hat{X}) = 5, \qquad h^{(3,1)}(\hat{X}) = 258, (0),$$
(5.28)

where the zeros in parenthesis indicate the absence of any non-toric divisors, respectively, complex structure moduli on \hat{X} as expected by construction. The Euler number $\chi(\hat{X})$ as well as the number $h_V^{(2,2)}(\hat{X})$ of independent surfaces in (5.28) agree with the general formula in (5.5), respectively, the cohomology calculations leading to (3.34). The topological metric $\eta^{(2)}$ of these five classes is read off from (3.36) with $n_7 = 4$, $n_9 = 5$.

The codimension two singularities of this concrete Calabi-Yau fourfold \hat{X} yield the full spectrum (3.3) of six different singlet representations $\mathbf{1}_{(q_1,q_2)}$. The cohomology classes of the three sections are given by the toric divisors $S_P = E_2$, $S_Q = E_1$ and $S_R = H - E_1 - E_2 + H_B$ in (5.27) and agree with the general expression (3.32). We obtain the (1, 1)-forms inducing the U(1)×U(1)-gauge fields by evaluating the Shioda map of the sections \hat{s}_Q , \hat{s}_R following (3.35) as

$$\sigma(\hat{s}_Q) = S_Q - S_P - 4H_B, \qquad \sigma(\hat{s}_Q) = S_R - S_P - 9H_B.$$
(5.29)

The Nero-Tate height pairing b_{mn}^{α} of these two classes follows from (5.26) with $n_7 = 4$, $n_9 = 5$.

Finally, we calculate the general G_4 -flux on \hat{X} employing the procedure of section 5.1. We precisely obtain the G_4 -flux in (5.4) in the case $n_7 = 4$, $n_9 = 5$. With this result, we readily calculate the chiralities for the matter representations, along with the CS-terms (4.4), again obtaining a perfect match with the results (5.14), (5.21) and (5.18) evaluated for $n_7 = 4$, $n_9 = 5$. Then we also calculate the GS-terms (5.26), most prominently the gaugings $\Theta_{m\alpha}$, and show, following section 5.3, that all anomalies are cancelled.

6 Calabi-Yau fourfolds with $SU(5) \times U(1) \times U(1)$

Building on the results of previous sections, the aim of this section is to develop tools for the construction of phenomenologically appealing F-theory compactifications with additional non-Abelian gauge symmetries. For concreteness we consider in this section a Calabi-Yau fourfold $\hat{X}_{SU(5)}$ with a resolved SU(5)-singularity, while maintaining the rank two Mordell-Weil group. Compactifications of F-theory on $\hat{X}_{SU(5)}$ give rise to a four-dimensional GUT with SU(5) × U(1) × U(1) gauge group.

We consider here one particular geometric way of adding and resolving the SU(5)singularity. Once this non-Abelian sector is added, it is rather straightforward to implement the techniques developed in previous sections for the rank-two Abelian symmetry. As the first step in section 6.1 we describe the resolution to $\hat{X}_{SU(5)}$, determine the matter representations as well as Yukawa points and the matter surfaces for a subset of matter multiplets. We note, that the spectrum that we obtain in this analysis has not been found via the toric classification of SU(5)-TOPs in [46]. As the next step we calculate the vertical cohomology ring $H_V^{(*,*)}(\hat{X}_{SU(5)})$ in section 6.2. For a general base B we derive expressions for the Chern classes and Euler number of $\hat{X}_{SU(5)}$, before we calculate the full vertical cohomology for the fourfold with $B = \mathbb{P}^3$ explicitly.

We emphasize one finding of our analysis. With the addition of an SU(5)-singularity we encounter at one codimension three locus a non-flat fiber (NFF), i.e. at this particular codimension three locus the fiber of $\hat{X}_{SU(5)}$ is no longer a complex one-dimensional elliptic curve, but a complex two-dimensional surface C_{NFF} . We note that this was also observed in [46]. The physics of non-flat fibers has been studied in [84], to which we refer for more details. In summary, the complex surface in the fibration can be wrapped by an M5-brane, that gives rise, in the blow-down of the resolved fibration of $\hat{X}_{SU(5)}$, to a charged tensionless string with an infinite tower of massless excitations. In addition, further light states are contributed by M2-branes wrapping holomorphic curves in the surface of the non-flat fiber.

Compactifications of F-theory with such a spectrum of additional light states are problematic for phenomenology. One obvious way out is to geometrically avoid the presence of the Yukawa couplings supporting the dimension two fiber. This can be achieved by considering a base B where the dangerous Yukawa coupling does not exist but all the other physical features of the compactification on $\hat{X}_{SU(5)}$ are retained. See also [45] for a toric analysis. Such bases are easily constructed and realize phenomenologically interesting F-theory compactifications [85].

Another way to deal with the non-flat fiber is to forbid at least a chiral excess of the additional light states induced by C_{NFF} . This requires the G_4 -flux on $\hat{X}_{SU(5)}$ to integrate to zero over C_{NFF} ,

Non-flat fiber condition:
$$\int_{\mathcal{C}_{NFF}} G_4 = 0,$$
 (6.1)

Following this approach later in section 7, we show explicitly that we can obtain a physical and consistent low-energy spectrum with all 4D field theory anomalies cancelled even in the presence of a non-flat fiber at codimension three. It would be interesting to investigate the microscopic meaning of (6.1) in more detail.

6.1 Singularities of the fibration: gauge group, matter & Yukawa Couplings

In this section we analyze the codimension one, two and three singularities and their resolutions in $\hat{X}_{SU(5)}$. We thoroughly determine all matter representations and determine some of the associated matter surfaces.

Before going into the details of this geometric analysis of $\hat{X}_{SU(5)}$ it is instructive to introduce some terminology and the following simple geometric picture. We construct $\hat{X}_{SU(5)}$ by first considering a non-generic Calabi-Yau fourfold \hat{X} with an SU(5)-singularity over codimension one in B. Then we resolve all singularities and obtain the smooth $\hat{X}_{SU(5)}$. Thus, the smooth Calabi-Yau fourfold $\hat{X}_{SU(5)}$ is schematically obtained from a singular Weierstrass model X as,

$$\hat{X}_{\mathrm{SU}(5)} \xrightarrow{\pi_{\mathrm{SU}(5)}} \hat{X} \xrightarrow{\hat{\pi}} X.$$
 (6.2)

Here $\pi_{SU(5)}$ blows down the four Cartan divisors of SU(5) and $\hat{\pi} = \pi_1 \circ \pi_2$ is the two-step blow-up in (3.2). Thus the fourfold $\hat{X}_{SU(5)}$ is constructed via a combination of a total of six resolutions starting from the singular Weierstrass model X.

6.1.1 Explicit resolution of a codimension one SU(5)-singularity

In the following we first engineer and then resolve an SU(5)-singularity over codimension one in *B*. We introduce the section $z \in \mathcal{O}(\mathcal{S}_{SU(5)})$ that vanishes along the divisor $\mathcal{S}^b_{SU(5)}$



Figure 5. Each dot represents a dP_2 -fibration over \mathbb{P}^3 with generic Calabi-Yau $\hat{X}_{SU(5)}$. The red region is the set of dP_2 -fibration with generic \hat{X} without SU(5), cf. figure 2.

in the base B of SU(5)-singularities of the elliptic fibration of \hat{X} . Here we focus on the interplay between the non-Abelian and Abelian sector and choose one particular SU(5)-singularity first constructed in [51] that has been missed in the literature before.

The SU(5)-singularity of interest is obtained by considering non-generic coefficients s_i in the Calabi-Yau equation (2.1) of the form

$$s_1 = z^3 s'_1, \qquad s_2 = z^2 s'_2, \qquad s_3 = z^2 s'_3, \qquad s_5 = z s'_5.$$
 (6.3)

Since the s_i are still required to be sections of the bundles (2.10) by the Calabi-Yau condition, this implies that the s'_i have to be sections of the following line bundles:

$$\begin{array}{c|cccc} section & bundle \\ \hline s'_{1} & \mathcal{O}(3[K_{B}^{-1}] - \mathcal{S}_{7} - \mathcal{S}_{9} - 3\mathcal{S}_{SU(5)}) \\ s'_{2} & \mathcal{O}(2[K_{B}^{-1}] - \mathcal{S}_{9} - 2\mathcal{S}_{SU(5)}) \\ s'_{3} & \mathcal{O}([K_{B}^{-1}] + \mathcal{S}_{7} - \mathcal{S}_{9} - 2\mathcal{S}_{SU(5)}) \\ s'_{5} & \mathcal{O}(2[K_{B}^{-1}] - \mathcal{S}_{7} - \mathcal{S}_{SU(5)}) \\ \end{array} \tag{6.4}$$

We note that we assume in the following that we obtain a fourfold $\hat{X}_{SU(5)}$ that is a generic besides the resolved SU(5)-singularity, i.e. that all the s_i respectively s'_i in (2.10) and (6.9) exist. As before in the U(1)²-case, this poses upper and lower bounds on the coefficients n_7^{α} , n_9^{α} in the expansion (2.25) of \mathcal{S}_7 , \mathcal{S}_9 , that depend on the base B. For the case $B = \mathbb{P}^3$ that is of most interest in this work, these bounds replace (2.27) and read, using $K_{\mathbb{P}^3}^{-1} = \mathcal{O}_{\mathbb{P}^3}(4)$ and $\mathcal{S}_{SU(5)} = H_B$,

 $0 \le n_7 \le 7$, $0 \le n_9 \le 6$, $n_7 + n_9 \le 9$, $0 \le 2 + n_7 - n_9$, $0 \le 4 + n_9 - n_7$. (6.5)

The allowed region is displayed in figure 5.

Next we confirm that the specialization 6.3 of the coefficients indeed gives rise to an SU(5)-singularity. The discriminant takes the form

$$\Delta = -z^5 \left(\beta_5^4 P + z\beta_5^2 P_2 R + \mathcal{O}(z^2)\right) \tag{6.6}$$

where we defined

$$\beta_5 = s_6, \quad P := P_1 P_2 P_3 P_4 P_5 = (s_2' s_5' - s_1' s_6) s_7 (s_2' s_7 - s_3' s_6) s_8 (s_6 s_9 - s_7 s_8), \tag{6.7}$$

and R is a polynomial in s_i with no common factors.

We note that the factor z^5 in (6.6) signals an A_4 -singularity over the divisor z = 0, as desired. In addition, we see that the singularity gets further enhanced at $\beta_5 = 0$ to D_5 and at P = 0 to A_5 , indicating appearance of matter multiplets at codimension two singularities. Yukawa points are located at the codimension three loci.

In order to resolve all singularities, we perform four blow-ups of the ambient space (2.7) that induce two blow-ups and two small resolutions on the Calabi-Yau fourfold. The blow-down map $\pi_{SU(5)}$: $\hat{X}_{SU(5)} \rightarrow \hat{X}$ of this resolution then reads [51],

$$\pi_{\rm SU(5)}: \quad w = d_1 d_2^2 d_3^3 d_4^2 \tilde{w} \,, \quad v = d_1 d_2 d_3 d_4 \tilde{v} \,, \quad u = d_3 d_4 \tilde{u} \,, \quad z = d_1 d_2 d_3 d_4 \tilde{z} \,, \tag{6.8}$$

where we have introduced new coordinates \tilde{w} , \tilde{v} , \tilde{u} and \tilde{z} as well as the d_i which are sections of the line bundles $\mathcal{O}(D_i)$ associated with the exceptional divisors D_i for i = 1...4. The latter are the Cartan divisors of $\hat{X}_{SU(5)}$ and admit the fibration structure (2.23) with $\mathcal{S}_{SU(5)} = \pi(\mathcal{S}_G^b)$. In the following it will be convenient to denote the new class of \tilde{z} as D_0 and supplement the Cartan divisors as $D_I = (D_0, D_i)_I$ with I = 0, 1, ..., 4. The divisor classes of all homogeneous coordinates in (6.8) are then given by

section
 bundle

$$\tilde{u}$$
 $\mathcal{O}(H - E_1 - E_2 + S_9 + [K_B] - D_3 - D_4)$
 \tilde{v}
 \tilde{v}
 $\mathcal{O}(H - E_2 + S_9 - S_7 - D_1 - D_2 - D_3 - D_4)$
 (6.9)

 \tilde{w}
 $\mathcal{O}(H - E_1 - D_1 - 2D_2 - 3D_3 - 2D_4)$
 \tilde{z}
 \tilde{z}
 $\mathcal{O}(D_0) \equiv \mathcal{O}(\mathcal{S}_{SU(5)} - D_1 - D_2 - D_3 - D_4)$
 \tilde{z}

The total transform of the Calabi-Yau hypersurface (2.1) then reads

$$p = s_6(e_1e_2)uvw + s_7(d_1d_2)e_2v^2w + s_8(d_2d_3^2d_4)e_1^2uw^2 + s_9(d_1d_4d_3^2d_2^2)e_1vw^2$$

$$+ z_2s_5(d_4d_3)e_1^2e_2u^2w + z_2^2s_2(d_1d_4)e_1e_2^2u^2v + z_2^2s_3(d_1^2d_2d_4)e_2^2uv^2 + z_2^3s_1(d_1d_4^2d_3)e_1^2e_2^2u^3,$$
(6.10)

where we dropped, by abuse of notation, all tildes of the coordinates in (6.8) and all primes of the sections in (6.9). The hypersurface (6.10) is a section of the anti-canonical bundle of the new ambient space, denoted $\widehat{dP}_2^B(\mathcal{S}_7, \mathcal{S}_9)$, after the blow-up (6.8). It reads

$$K_{\widehat{dP}_2^B}^{-1} = \pi_{\mathrm{SU}(5)}^* \left(K_{dP_2^B}^{-1} \right) - 2D_1 - 3D_2 - 5D_3 - 4D_4 \,, \tag{6.11}$$

where $K_{dP_2^B}^{-1}$ denotes the anti-canonical bundle (2.9) of $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$ that is pulled back via the blow-down map (6.8).

There is one caveat when working with a particular resolution such as (6.8). It has been discussed in detail in [20–22], but has also been long clear in the toric context, see e.g. [41, 42, 44, 78] for recent studies, that the resolution of an SU(5)-singularity over codimension one in *B* is not unique. The different resolutions are related by mild birational transformations. In the context of toric geometry, this is visible as different phases of the toric ambient variety²¹, that descend to different phases on the Calabi-Yau fourfold $\hat{X}_{SU(5)}$. One way to specify these different phases is by the Stanley-Reissner ideal SR, which describes which divisors in (6.9) do not intersect. However, we can always restrict ourselves to one particular resolution in order to extract physical quantities like the matter content, Yukawa points and chiralities in F-theory. This is intuitively clear since the resolved geometry $\hat{X}_{SU(5)}$ is just a tool to analyze the geometry and not physical in Ftheory. This can be argued more precisely in the dual M-theory compactification similar to the discussion in [79].

Thus, we focus here on one particular phase of the resolution (6.8) to extract the effective physics of F-theory on $\hat{X}_{SU(5)}$. Following the spirit of toric geometry, we specify the precise resolution (6.8) by stating a specific SR-ideal. Here we consider an ideal of the form

$$SR = \{ve_1, e_1e_2, uv, uw, we_2\} \cup \{zd_3, d_1d_3, d_1d_4\} \cup \{d_1e_1, ze_1, d_2e_1, d_4e_1, d_1u, d_2u, d_1e_2, d_2e_2, d_3e_2, d_4e_2, zw, d_1w, d_4w, d_3v, d_4v, zd_2v\} \cup \pi^*_{SU(5)}(SR_B).$$
(6.12)

We immediately recover the SR-ideal (2.5) of the dP_2 -fiber as the first set in (6.12), ensuring the intersections (2.4) of the fiber. The second set gives rise to the correct intersections of the Cartan divisors to obtain the intersection pattern of the A_4 Dynkin diagram. The third part encodes the intersections of the Cartan divisors with the sections \hat{s}_P , \hat{s}_m . The last part in (6.12) is the pull-back of the SR-ideal SR_B of the base B under the blow-down map $\pi_{SU(5)}$ in (6.8). We will comment on the structure of this ideal in more detail in section 6.2. For now we just state that it contains the SR-ideal of the base and additional intersections involving the D_i and the vertical divisors, that take care of the split of the divisor z into multiple components under the blow-up map (6.8).

Let us next discuss the structure of the resolution $X_{SU(5)}$ at codimension one in B. The general fiber of $\hat{X}_{SU(5)}$ is the dP_2 -elliptic curve \mathcal{E} . However, over $\mathcal{S}^b_{SU(5)}$ the fiber becomes reducible. One way to see this is by noting the following intersections of the Cartan divisors D_I ,

$$D_I \cdot D_J \cdot D_\alpha \cdot D_\beta = -C_{IJ} \mathcal{S}_{\mathrm{SU}(5)} \cdot S_P \cdot D_\alpha, \cdot D_\beta \tag{6.13}$$

where D_{α} , D_{β} are arbitrary vertical divisors and C_{IJ} is the extended Cartan matrix of SU(5) reading

$$(-C_{IJ}) = \begin{pmatrix} -2 & 1 & 0 & 0 & 1 \\ 1 & -2 & 1 & 0 & 0 \\ 0 & 1 & -2 & 1 & 0 \\ 0 & 0 & 1 & -2 & 1 \\ 1 & 0 & 0 & 1 & -2 \end{pmatrix}.$$
 (6.14)

²¹In general, there can also be phases of the toric variety and $\hat{X}_{SU(5)}$ that do not correspond to different resolutions but for example to different phases of *B* or of the general fiber \mathcal{E} . These might even change the intersections e.g. of the fiber \mathcal{E} from the ones considered in section 2.1 by going for instance into a phase corresponding to F_1 blown up at a point. We exclude these phases here and in particular in sections 5.4 and 7.3.



Figure 6. I_4 -fiber over $S^b_{SU(5)}$ marked according to the intersections of the nodes with S_P , S_Q , S_R , corresponding to a 2 – 3-split of a spectral cover.

We note that (6.13) can be calculated via the abstract presentation (3.18) of the cohomology ring of $\hat{X}_{SU(5)}$, see section 6.2 and appendix D for details. The intersections (6.13) then immediately imply that the divisors D_I resolve a singularity of type G = SU(5). Indeed, we note first that (6.13) implies a fibration structure (2.23) of the Cartan divisors π : $D_I \to S^b_{SU(5)}$ with the rational fibers $c_{-\alpha_I}$ corresponding to the simple roots $-\alpha_i$ and the extended root $-\alpha_0 = \sum_i \alpha_i$ of SU(5). The latter conclusion can be shown by representing the individual curves $c_{-\alpha_I}$ as intersections of the divisors D_i with a curve Σ_b dual to $\mathcal{S}^b_{SU(5)}$ in the base, i.e. assuming $\mathcal{S}^b_{SU(5)} \cdot \Sigma_b = 1$ we obtain

$$D_I \cdot \Sigma_b = c_{-\alpha_I} \,. \tag{6.15}$$

Then, we immediately confirm from (6.13) and (6.14) that the $c_{-\alpha_I}$ in fact intersect as the affine Dynkin diagram of SU(5), see figure 6, justifying the assignment $c_{-\alpha_I} \leftrightarrow -\alpha_I$.

Next we turn to the behavior of the rational sections of $\hat{X}_{SU(5)}$. Generically, the sections are points in the fiber and only intersect one of its irreducible components. As we have just seen the fiber over $\mathcal{S}_{SU(5)}^b$ splits into five rational curves. The component which is pierced by the three sections \hat{s}_P , \hat{s}_Q and \hat{s}_R can be determined by calculating the intersections of their divisor classes S_P , $S_{Q,R}$ with the curves (6.15). We find that S_P and S_R intersect $c_{-\alpha_0}$, and S_Q passes through $c_{-\alpha_3}$, i.e.

$$S_P \cdot c_{-\alpha_I} = S_R \cdot c_{-\alpha_I} = (1, 0, 0, 0, 0)_I, \quad S_Q \cdot c_{-\alpha_I} = (0, 0, 0, 1, 0, 0)_I.$$
(6.16)

This situation is summarized in figure 6. We note that this intersection pattern of the rational sections would correspond to a 2-3 split in the language of spectral covers. Again, the result (6.16) can be obtained via the abstract intersection computation outlined in section 6.2 and in appendix D.

Equipped with these intersections of the sections with the nodes of the Dynkin diagram, we can calculate the Shioda maps of the sections \hat{s}_Q , \hat{s}_R . We obtain

$$\sigma(\hat{s}_Q) = S_Q - S_P - [K_B^{-1}] + \frac{1}{5}(2D_1 + 4D_2 + 6D_3 + 3D_4), \qquad \sigma(\hat{s}_R) = \hat{S}_R - S_P - [K_B^{-1}] - \mathcal{S}_9.$$
(6.17)

Here we have used the general formula for the Shioda map in the presence of one set of Cartan divisors D_i due to a codimension one singularity of type G of the elliptic fibration,

$$\sigma(\hat{s}_m) := S_m - \tilde{S}_P - \pi(S_m \cdot \tilde{S}_P) + \left(\mathcal{C}^{-1}\right)^{ij} \left(S_m \cdot c_{-\alpha_i}\right) D_j.$$
(6.18)

The first two terms are identical to those in the Shioda map (2.20), whereas the last term incorporates the presence of G and involves the fibral curves $c_{-\alpha_i}$ corresponding to the simple roots of G as well as the inverse of the normalized coroot matrix C_{ij} of G. For the case of SU(5) considered here C_{ij} is identical to the Cartan matrix C_{ij} .

6.1.2 Matter: codimension two

Now we proceed to analyze the resolved codimension two singularities in $\hat{X}_{SU(5)}$. There are five loci $z = P_i = 0, i = 1, ..., 5$, in (6.7) where the order of vanishing of the discriminant enhances, indicating the presence of an I_5 -singularity and of **5** representations in F-theory. In addition, there is one codimension two loci $z = \beta_5 = 0$, where the singularity enhances to type D_5 and where matter in the **10** representation is located F-theory.

We analyze in the following the splitting of the nodes of the A_4 -fiber over all these codimension two loci, determine the U×U(1)-charges (q_1, q_2) of all the non-Abelian representations along with their KK-charges, which we find to be zero. As in the U(1)×U(1) case there will be only the SU(5)-singlet $\mathbf{1}_{(-1,-2)}$ with non-trivial KK-charge. We also determine which weights of the respective representations are realized as holomorphic curves and thus lie in the Mori cone of $\hat{X}_{SU(5)}$. This fixes the sign-function in (4.12) for each of these representation and allows for the field theoretic computations outlined in section 4.2, that will be used in section 7. We conclude with a determination of some matter surfaces $C_{\mathbf{R}}^{\mathbf{w}}$.

The analysis of the splitting of nodes at codimension two presented in this section is very similar to the ones performed earlier in the literature based on the Tate model of elliptic fibrations. Therefore, our discussion will be brief and we refer to [21, 22] for more background on the general methodology.

The 5-representations and their $U(1) \times U(1)$ -charges

In the following we study the codimension two loci $z = P_1 = 0$ in detail. The analysis of the other loci $z = P_i = 0$ of 5-representations is very similar. Therefore, we will only summarize the results of our computations at the end of this section.

At the loci $P_1 = (s_2s_5 - s_1s_6) = 0$ defined (6.7) the second node in figure of the fiber becomes reducible, as depicted figure 7. This can be observed by intersecting the locus $d_2 = P_1 = 0$ with the Calabi-Yau equation (6.10). Since we are only interested in the split of the fiber, we evaluate all coefficients s_i at generic points on B, cf. (6.15). Then, we locally obtain

$$p|_{d_2=P_1=0} = \frac{1}{s_1} \left(s_2 v + s_1 d_3 d_4 z \right) \left(s_5 w + s_1 d_1 d_4 z^2 \right).$$
(6.19)

This means that the class of the curve $c_{-\alpha_2}$ splits into two curves c_1, c_2 when restricted to the non-generic loci $P_1 = 0$. From the perspective of the ambient space $\widehat{dP}_2^B(\mathcal{S}_7, \mathcal{S}_9)$ we can write this splitting using (6.19) as

$$c_{-\alpha_2} = c_1 + c_2, \qquad [c_1] = H_B^3 \cdot [v], \quad [c_2] = H_B^3 \cdot [w], \qquad (6.20)$$



Figure 7. Splitting of the generic I_4 -fiber over $P_1 = 0$.

where we denoted the homology classes of the curve c_i as $[c_i]$ and used the shorthand notation [v] for the homology class of v, cf. (6.9). In order to arrive at this result we first have to use the SR-ideal 6.12 and the fact that $H_B^4 = 0$.

The splitted curves c_1 , c_2 correspond to a weight **w**, respectively, $-\mathbf{w}$ of the 5representation according to the general correspondence between isolated rational curves over codimension two in *B* and matter in F-theory explained before (3.1). The Dynkin label²² (q_i) of a weight **w** is calculated in general geometrically as the intersection

$$q_i = D_i \cdot c_{\mathbf{w}} \,, \tag{6.21}$$

which is in complete analogy with the formula (3.4) for the computation of U(1)-charges. We apply (6.21) for the curves in (6.20) to obtain the Dynkin labels of c_1 and c_2 as

$$D_i \cdot c_1 = (1, -1, 0, 0)_i, \qquad D_i \cdot c_2 = (0, -1, 1, 0)_i,$$
(6.22)

respectively. As an immediate consistency check we recover the Dynkin label of $-\alpha_2$ as the sum of these Dynkin labels. We also note, for completeness, that in this case, none of the curves c_1 , c_2 intersect the rational sections, as can be seen by calculating intersections as in (6.16). This is again expected since the original unsplitted curve $c_{-\alpha_2}$ does not intersect the rational sections either, see figure 7.

The first set of Dynkin labels in (6.22) is recognized as the Dynkin label of the weight $-(\mu_5 - \alpha_1)$, where $-\mu_5$ is the highest weight with Dynkin label $\Lambda_5 = (-1, 0, 0, 0)$. Since the Cartan divisors correspond to $-\alpha_i$, however, we interpret this as 5-representation The second Dynkin label corresponds to the weight $(\mu_5 - \alpha_1 - \alpha_2)$ of the 5 representation. All the other weights of 5 can then be obtained by adding appropriately one of the curves $c_{1,2}$ and a number of curves $c_{-\alpha_i}$ of the simple roots $c_{-\alpha_i}$. We exemplify this here to obtain the remaining weights of the $\overline{\mathbf{5}}$,

Dynkin label	Curves
(-1,0,0,0)	$c_1 + c_{-\alpha_1}$
(1,-1,0,0)	c_1
(0,1,-1,0)	$-(c_2)$
(0,0,1,-1)	$-(c_2+c_{-\alpha_3})$
(0,0,0,1)	$-(c_2+c_{-\alpha_3}+c_{-\alpha_4})$

²²The Dynkin label of **w** is the vector of 'U(1)-charges' of **w** under all Cartan generators.

(27)

From this table we can directly see which curves are in the Mori cone of $X_{SU(5)}$. The two curves c_1 , c_2 are effective because we explicitly know their holomorphic representation (6.19). Also by construction, the curves associated to the simple roots $c_{-\alpha_i}$ are effective and so are the sums of effective curves with positive coefficients. This implies that we can now evaluate the sign-function (4.12). It is positive for the first two representations in table 6.23, but negative for the other three curves, that are not in the Mori cone. We summarize these findings, along with the signs of the other representations in (6.27).

Next, we calculate the U(1)×U(1)-charge using the first equality in the charge formula (3.4). For the U(1) associated to the S_Q section, we intersect the Shioda map $\sigma(\hat{s}_Q)$ in (6.17) with any of the curves in (6.23) of the representation. For example using the locations in figure 7 of the sections on the fiber, as well as the intersections (6.13) and (6.22) we obtain

$$q_{1} = \sigma(\hat{s}_{Q}) \cdot (c_{1} + c_{-\alpha_{1}}) = 0 - 0 + (-1, 0, 0, 0) \frac{1}{5} \begin{pmatrix} 4 & 3 & 2 & 1 \\ 3 & 6 & 4 & 2 \\ 2 & 4 & 6 & 3 \\ 1 & 2 & 3 & 4 \end{pmatrix} \begin{pmatrix} 0 \\ 0 \\ 1 \\ 0 \end{pmatrix} = -\frac{2}{5}, \quad (6.24)$$

The second U(1)-charge related to the section S_R follows similarly as

$$q_2 = \sigma(\hat{s}_R) \cdot (c_1 + c_{-\alpha_1}) = 0. \tag{6.25}$$

In summary, we have found the matter representation $\mathbf{5}_{(-\frac{2}{5},0)}$ is realized at the codimension two loci $P_1 = z = 0$.

A similar analysis can be performed at all other loci $z = P_i = 0$, i = 2, 3, 4, 5 of 5representations. The computations are completely analogous to the ones performed above, but not very enlightening. In all cases we obtain a split of nodes similar to the one in figure 7 and obtain an I_5 -fiber. The U(1)×U(1)-charges can be straightforwardly calculated. We skip these details and just summarize all charges of 5 representations we obtain:

Locus
$$P_1 = 0$$
 $P_2 = 0$ $P_3 = 0$ $P_4 = 0$ $P_5 = 0$
 (q_1, q_e) $(-\frac{2}{5}, 0)$ $(-\frac{2}{5}, 1)$ $(\frac{3}{5}, 0)$ $(\frac{3}{5}, 1)$ $(-\frac{2}{5}, -1)$

$$(6.26)$$

We note that we are working in a normalization of charges which allows for fraction. We can rescale our charges as usually done in the literature in this case by a factor of 5 to obtain integral charges.

We also summarize which curves $c_{\mathbf{w}}$ of which weights are in the Mori cone of $\hat{X}_{SU(5)}$. As before we use this to determine the values of the sign-function in (4.12). We obtain the following results:

Weight	${f 5}_{(-rac{2}{5},0)}$	$5_{(-\frac{2}{5},1)}$	$5_{(rac{3}{5},0)}$	$5_{\left(\frac{3}{5},1\right)}$	${f 5}_{(-rac{2}{5},-1)}$	
(-1,0,0,0)	+	+	+	+	-	
(1, -1, 0, 0)	+	+	+	+	-) (6
(0, 1, -1, 0)	-	+	+	+	-	, î
(0, 0, 1, -1)	-	-	-	+	-	
(0, 0, 0, 1)	_	-	-	+	-	



Figure 8. D_5 -curve over $z = \beta_5 = 0$.

The 10-representation and its $U(1) \times U(1)$ -charges

For completeness, we briefly discuss the fiber at the loci $z = \beta_5 = 0$. In this case more curves $c_{-\alpha_i}$ become reducible. Repeating the same procedure as above, we insert $z = \beta_5 = 0$ and $d_i = \beta_5 = 0$ into the Calabi-Yau equation (6.10), we obtain the following splitting of nodes,

Node	splits into	Cartan charges		
z = 0	$c_a: d_4s_8u + d_1d_2d_4s_9v + d_1e_2s_7v^2 = 0$	(0, 1, 0, 0)	-	
	$c_b: d_2 = 0$	(1, -1, 0, 1)		
$d_1 = 0$	$c_{-\alpha_1}: \ d_2s_8 + s_5z_2 = 0$	(-2, 1, 0, 0)		
$d_2 = 0$	$c_b: \ z_2 = 0$	(1, -1, 0, -1)	(6.28)	
	$c_{-\alpha_4}: \ d_4 = 0$	(0, 0, 1, -2)		
	$c_c: \ d_1d_3d_4s_1z_2^2 + d_1s_2vz_2 + d_3s_5w = 0$	(0, -1, 0, 1)		
$d_3 = 0$	$c_{-\alpha_3}: d_2d_4s_3u + d_4e_1s_2u^2 + d_2s_7w = 0$	(0, 1, -2, 1)		
$d_4 = 0$	$c_{-\alpha_4}: \ d_4 = 0$	(0, 0, 1, -2)		

We observe that both the curves $c_{-\alpha_0}$ and $c_{-\alpha_2}$ split into two, respectively, three components, denoted c_a , c_b , respectively, c_b , $c_{-\alpha_4}$ and c_c , where we note that the curve c_a is the extended note, i.e. the original singular fiber. The fact that the curves c_b and $c_{-\alpha_4}$ appear twice means that they have multiplicity two. The intersection numbers between all the curves in (6.28) can be calculated taking into account the SR-ideal of the ambient space. The intersection pattern of the curves reproduces a D_5 -curve, see figure 8.

The curves $c_{a,b,c,d}$ carry some weights of the **10** and **10** representations. All the other weights are obtained as before by addition of curves $c_{-\alpha_i}$. The U(1)×U(1)-charges of the **10** representation are determined by intersections with the Shioda maps $\sigma(\hat{s}_m)$ of the rational sections. We obtain the representation

$$\mathbf{10}_{(q_1,q_2)} = \mathbf{10}_{(\frac{1}{r},0)}.$$
 (6.29)

Finally, we enumerate which curves $c_{\mathbf{w}}$ of this representation are in the Mori cone of $\hat{X}_{SU(5)}$.

(6.30)

Weight	Sign	Weight	Sign
(0, -1, 0, 0)	+	(-1, 0, 0, 1)	+
(-1, 1, -1, 0)	+	(0, 1, 0, -1)	-
(1, 0, -1, 0)	+	(1, -1, 0, 1)	+
(-1, 0, 1, -1)	+	(0, 1, -1, 1)	-
(1, -1, 1, -1)	+	(0, 0, 1, 0)	-

As before we indicate this by the values of the sign-function (4.12), which reads

KK-charges

We note that at all codimension two singularities inside the divisor $S_{SU(5)}$ the section S_P is holomorphic. This means for these type of matter the KK-charges vanish, $q_{KK} = 0$. The only representation with non-trivial KK-charge is, as before in section 5.2 the matter field $\mathbf{1}_{(-1,-2)}$. Its KK-charge is $q_{KK} = 2$.

Matter surfaces

We conclude the discussion of codimension two singularities by constructing the classes of certain matter surfaces $C_{\mathbf{R}}^{\mathbf{w}}$. As before in section 3.1 we are able to obtain these classes only for a subset of the representations.

We first comment on the calculation of matter curves in the base. A very convenient way to calculate all matter curves $\Sigma_{\mathbf{R}}$ is by projecting the intersections $d_i^* = P_l = 0$, $l = 1, \ldots, 5$ or $d_i^* = \beta_5 = 0$ to the base, where d_i^* is the coordinates associated to the Cartan divisor D_i that has split at that codimension two loci under consideration. The homology classes of the matter surfaces are then given as

$$\Sigma_{\mathbf{R}} \cong [B] \cdot [D_i^{\star}] \cdot [P_i] = [B] \cdot [\mathcal{S}_{\mathrm{SU}(5)}] \cdot [P_i], \qquad (6.31)$$

where the divisor D_i^{\star} has to correspond to the node that has split at the location of the matter representation **R**. Here we have used the property $\pi(D_I) = S_{SU(5)}$ in the second equality.

As in the Abelian case, we can only specify some matter surfaces completely. As before, this complication arises since we are only able to isolate the irreducible components in the elliptic fiber locally, but can not infer the total space of the fibration $C_{\mathbf{R}}^{\mathbf{w}} \to \Sigma_{\mathbf{R}}$. The subset of matter surfaces, that we can explicitly determine, along with their homology classes, reads,

Rep.	Matter curve $\Sigma_{\mathbf{R}}$	Matter surface $C_{\mathbf{R}}$
$5_{(-rac{2}{5},0)}$	$\mathcal{S}_{\mathrm{SU}(5)} \cdot ([s_2] + [s_5])$	_
$5_{(-\frac{2}{5},1)}$	$\mathcal{S}_{\mathrm{SU}(5)} \cdot \mathcal{S}_7$	$([K_B^{-1}] + S_9 - S_7 + 2H - 2D_1 - 3D_2 - 4D_3 - 3D_4) \cdot D_3 \cdot S_7$
$5_{(rac{3}{5},0)}$	$\mathcal{S}_{\mathrm{SU}(5)} \cdot \left([s_3] + [K_B^{-1}] \right)$	-
$5_{(rac{3}{5},1)}$	$\mathcal{S}_{\mathrm{SU}(5)} \cdot [s_8]$	$(\mathcal{S}_9 + 2H - E_1 - D_1 - 2D_2 - 4D_3 - 3D_4) \cdot D_0 \cdot ([K_B^{-1}] + \mathcal{S}_9 - \mathcal{S}_7)$
$5_{(-\frac{2}{5},-1)}$	$\mathcal{S}_{\mathrm{SU}(5)} \cdot (\mathcal{S}_7 + \mathcal{S}_9)$	-
$10_{(\frac{1}{5},0)}$	$\mathcal{S}_{\mathrm{SU}(5)} \cdot [K_B^{-1}]$	$D_0 \cdot D_2 \cdot [K_B^{-1}]$
		(6.32)

Here we made use of the SR-ideal (6.12), the assignments of sections given in (6.4), (6.9) and have abbreviated the divisors class of the s_i as $[s_i]$. In addition, we have suppressed the intersection with the class of the base B when denoting the matter curves $\Sigma_{\mathbf{R}}$ and by abuse of notation used same symbol for vertical divisors and divisor in the base.

The matter surfaces for the singlets $\mathbf{1}_{(-1,1)}$, $\mathbf{1}_{(0,2)}$ and $\mathbf{1}_{(-1,-2)}$ have been analyzed in detail in section 3.1 in the case of a U(1)×U(1) gauge group only. In the presence of an SU(5) at codimension one, the matter surfaces determined in (3.7) change. Taking into account the new classes (6.4), (6.9) and the new equation of the Calabi-Yau fourfold (6.10) we obtain the matter surfaces as

Matter surface	Homology class
$\mathcal{C}_{1_{(-1,1)}}$	$([K_B^{-1}] + \mathcal{S}_7 - \mathcal{S}_9 - 2\mathcal{S}_{\mathrm{SU}(5)}) \cdot \mathcal{S}_7 \cdot E_1$
$\mathcal{C}_{1_{(0,2)}}$	$S_7 \cdot S_9 \cdot ([K_B^{-1}] + S_9 - S_7 + 2H - 2D_1 - 3D_2 - 4D_3 - 3D_4)$
$\mathcal{C}_{1_{(-1,-2)}}$	$\left([K_B^{-1}] + \mathcal{S}_9 - \mathcal{S}_7) \cdot \mathcal{S}_9 \cdot \left(3H - E_1 - 2E_2 + 2\mathcal{S}_9 - \mathcal{S}_7 - 2D_1 - 3D_2 - 5D_3 - 4D_4\right)\right)$
	(6.22)

This can be seen by specializing (6.10) to the matter curves $\Sigma_{\mathbf{1}_{(-1,1)}} = \{s'_3 = s_7 = 0\}$, $\Sigma_{\mathbf{1}_{(0,2)}} = \{s_7 = s_9 = 0\}$, respectively, $\Sigma_{\mathbf{1}_{(-1,-2)}} = \{s_8 = s_9 = 0\}$ and, then, by multiplying with the class of the isolated rational curves over each matter curves. We recall that the latter are given by $c_{(-1,1)} = \{e_1\}$ as well as the total transforms of (3.9) and (3.10).

6.1.3 Yukawa couplings: codimension three

We focus in this section on the determination of those Yukawa points that involve at least one the non-trivial representations under SU(5). For an analysis of the Yukawa points of the singlets, we refer to the discussion in section 3.1.2, that applies without any changes.

Further enhancement of singularity type of the elliptic fibration can be read directly from the discriminant (6.6) of $\hat{X}_{SU(5)}$. Our strategy in determining these loci is to look for two polynomials, typically describing two matter curves, such that at their common vanishing locus the discriminant vanishes with order n greater than five, i.e. $\Delta \sim z^n$ for n > 5. Then we check explicitly that the fiber of the resolution $\hat{X}_{SU(5)}$ splits further. We perform this analysis in the following for all present matter representations of $\hat{X}_{SU(5)}$.

We begin with the loci of the single 10 representation. We note that here, since $\beta_5 = s_6 = 0$, the discriminant already vanishes to degree seven and takes the form

$$\Delta = -16z^7 (s_5 s_7)^3 (s_2 s_8 s_7)^2 + \mathcal{O}(z^8).$$
(6.34)

Vanishing of any of these prefactors will enhance the zero of the discriminant to degree eight or higher. First, we analyze the vanishing $s_6 = s_5 = 0$ on the divisor z = 0. The polynomial P_1 vanishes but no other P_i does, so this point belongs both to the matter curve of $\mathbf{5}_{2/5,0}$ and of **10**. Although the fiber does not degenerated to an E_6 -fiber the Yukawa coupling exists. The gauge invariant coupling we read off is

$$s_5 = s_6 = z = 0 : \qquad \mathbf{10}_{\left(\frac{1}{5},0\right)} \times \mathbf{10}_{\left(\frac{1}{5},0\right)} \times \mathbf{5}_{\left(-\frac{2}{5},0\right)}, \tag{6.35}$$

Next, we study the vanishings of s_2 or s_8 with $s_6 = z = 0$. In this case we obtain couplings of the type $5 \times 5 \times \overline{10}$. We note that the fiber does not enhance to a D_6 -fiber, but the Yukawa is still realized geometrically.

Finally, we consider the last enhancement of the vanishing of (6.34) with $s_7 = s_6 = 0$. In this case the intersection of the Cartan divisor D_4 with the Calabi Yau hypersurface is automatically zero. Since we are now only specifying three conditions $d_4 = s_6 = s_7 = 0$ in a five-dimensional ambient space, the fiber has to be complex two-dimensional, namely the whole del Pezzo surface dP_2 . The elliptic fibration of $\hat{X}_{SU(5)}$ is non-flat, with a complex two-dimensional fiber over codimension three. We note that this effect occurs also for other SU(5)-embeddings [46].

The physics of this a non-flat fibration has been discussed in [84]. As mentioned before, an M5-brane can wrap the surface in the non-flat fiber, and give rise to a string in a threedimensional M-theory compactification. In addition, M2-branes can wrap holomorphic curves in dP_2 . In the blow-down of $\hat{X}_{SU(5)}$, and in particular in the F-theory limit, the states associated to these degrees of freedom become massless. For phenomenology these additional light states are usually undesired and can render the compactification unphysical, since we have to deal with an infinite tower of charged excitations of the string, which cannot easily be represented by a finite number of fields.

As discussed in the introduction of the section, one attempt to deal with the presence of the extra degrees of freedom from the non-flat fiber is to forbid a chiral excess of states. As we will demonstrate in section 7 this can be achieved by tuning the G_4 -flux as in (6.1), so that it integrates to zero over the non-flat fiber. Alternatively, we can simply forbid the Yukawa point with the non-flat fiber geometrically. Recalling $[s_6] = [K_B^{-1}]$, cf. (2.10), and $\pi(D_4) = S_{SU(5)}$, all we have to demand is that

$$\mathcal{S}_{7} \cdot [K_B^{-1}] \cdot \mathcal{S}_{\mathrm{SU}(5)} = 0 \tag{6.36}$$

One obvious but drastic solution is $S_7 = 0$ since it deprives us from some singlets and nonsinglets representations. A more sophisticated solution to (6.36) is obtained as follows. We recall the basis expansion (2.25) of the divisor S_7 and in addition expand

$$\mathcal{S}_{\mathrm{SU}(5)} = n^{\alpha}_{\mathrm{SU}(5)} D_{\alpha} \,. \tag{6.37}$$

Then, it is possible over an appropriate base B to tune the integers n_7^{α} and $n_{SU(5)}^{\alpha}$ in such a way to only forbid the dangerous Yukawa point (6.36) without restricting the spectrum of $\hat{X}_{SU(5)}$ and the other Yukawa points. As a concrete simple example one can choose B = $Bl\mathbb{P}^3$, the blow-up of \mathbb{P}^3 along the curve $x_i = x_j = 0$. Then (6.36) yields $n_{SU(5)}^1(4n_7^1 + 3n_7^2) +$ $3n_{SU(5)}^2n_7^1 = 0$ for $D_{\alpha} = \{H, H - E\}$ with H the hyperplane and E the exceptional divisor and $K_{Bl_1\mathbb{P}^3}^{-1} = 4H - E$. Clearly, this can be solved without removing other codimension two or three singularities. We summarize all Yukawa couplings involving the **10**-representation at $s_6 = z = 0$ in the following table,

Loci	Yukawa coupling	
$s_{5} = 0$	${f 5}_{(-rac{2}{5},0)} imes {f 10}_{(rac{1}{5},0)} imes {f 10}_{(rac{1}{5},0)}$	
$s_2 = 0$	$5_{(-rac{2}{5},0)} imes 5_{(rac{3}{5},0)} imes \overline{10}_{(-rac{1}{5},0)}$	(6.3)
$s_8 = 0$	$5_{(rac{3}{5},1)} imes 5_{(-rac{2}{5},-1)} imes \overline{10}_{(-rac{1}{5},0)}$	
$s_7 = 0$	Non-flat fiber	

Besides these Yukawa couplings there are additional Yukawa couplings involving singlets and the 5-representations. All these Yukawa couplings are localized at $z = s_7 = 0$. Evaluating the discriminant at this locus we obtain the first non-vanishing terms as

$$\Delta \cong -s_3^2 s_6^5 (s_2 s_5 - s_1 s_6) s_8 s_9^2. \tag{6.39}$$

Setting each prefactor to zero we obtain Yukawa couplings, that we summarize in the following table,

Loci	Yukawa coupling	
$s_3 = 0$	$\overline{5}_{(rac{2}{5},-1)} imes 5_{(rac{3}{5},0)} imes 1_{(-1,1)}$	
$(s_2s_5 - s_1s_6) = 0$	$5_{(-rac{2}{5},0)} imes\overline{5}_{(rac{2}{5},-1)} imes1_{(0,1)}$	(6.40)
$s_8 = 0$	$\overline{5}_{(rac{2}{5},-1)} imes 5_{(rac{3}{5},1)} imes \overline{1}_{(-1,0)}$	
$s_9 = 0$	$\overline{5}_{(\frac{2}{5},-1)} \times 5_{(-\frac{2}{5},-1)} \times 1_{(0,2)}$	

6.2 The cohomology ring and the Chern classes of $\hat{X}_{SU(5)}$

In this section we apply the techniques described in section 3.2 to perform computations in the vertical cohomology $H_V^{(*,*)}(\hat{X}_{SU(5)})$ of the Calabi-Yau fourfold $\hat{X}_{SU(5)}$ with a resolved SU(5)-singularity over $\mathcal{S}_{SU(5)}$. The computations are completely analogous to the ones performed to obtain the cohomology of \hat{X} . All we have to do is to add the Cartan divisors $D_i, i = 1, \ldots, 4$, as additional variables to the polynomial ring R in (3.18). We note that the extended node $D_0 = \mathcal{S}_{SU(5)} - \sum_i D_i$ corresponding to z is then automatically included. We also have to replace the ideal SR in (3.18) by the ideal (6.12) arising after the resolution process (6.8). Finally, we have to employ the anti-canonical bundle (6.11), which is the class of $\hat{X}_{SU(5)}$ in the blown-up ambient space $\widehat{dP}_2^B(\mathcal{S}_7, \mathcal{S}_9)$. We summarize the presentation of the vertical cohomology ring of $\hat{X}_{SU(5)}$ as

$$H_{V}^{(*,*)}(\hat{X}_{SU(5)}) \cong \frac{\mathbb{C}[D_{\alpha}, \mathcal{S}_{SU(5)}, S_{P}, S_{Q}, S_{R}, D_{i}] \cdot \left[\hat{X}_{SU(5)}\right]}{SR}, \qquad (6.41)$$

with the ideal SR given in (6.12). Here we have singled out the vertical divisor $S_{SU(5)}$ of the original SU(5)-singularity.

We begin by using these techniques to calculate the Euler number and Chern classes of $\hat{X}_{SU(5)}$ for a general base B. Then we construct the full cohomology in the case of $B = \mathbb{P}^3$. In the following discussion we will quote the main results from this analysis and refer to appendix D for more details on the calculation, in particular the full quartic intersections of $\hat{X}_{SU(5)}$ from which all the following is derived.

6.2.1 Second Chern classes and Euler number of $\hat{X}_{SU(5)}$: general formulas

The strategy for the computation of the Euler number is is a mixture of the strategy in section 3.2 and the approach of [21] to compute Chern classes of fourfolds with a resolved SU(5)-singularity. The key point is to find a SR-ideal that contains sufficiently fine information to do concrete calculations, but that is sufficiently coarse in order to not depend on the details of B. The starting point is the SR-ideal (6.12). The only degree of freedom here is the pullback $\pi_{SU(5)}(SR_B)$ of the SR-ideal of the base. Instead of using the full ideal that depends on the details of B we again work with a simplified version denoted SR'_B that is based on some universal geometric properties of the fibration of $\hat{X}_{SU(5)}$.

As in (3.20) we assume that SR'_B contains all quartic intersections of the vertical divisors D_{α} . In addition we assume as in [21] that three vertical D_{α} never intersect the Cartan divisors D_i due to a too large codimension in the base. Thus we use the ideal

$$SR'_{B} = \{ D_{\alpha} \cdot D_{\beta} \cdot D_{\gamma} \cdot D_{\delta}, D_{\alpha} \cdot D_{\beta} \cdot D_{\gamma} \cdot D_{i} \}$$

$$(6.42)$$

instead of $\pi_{SU(5)}(SR_B)$ in (6.12). As we demonstrate next this is sufficient to calculate the Chern classes and Euler number as well as to check basic intersections of $\hat{X}_{SU(5)}$, as for the fourfold \hat{X} considered in section 3.2.

Next we calculate the total Chern class of $\hat{X}_{SU(5)}$. All we need is the Chern class of the blown-up ambient space $\widehat{dP}_2^B(\mathcal{S}_7, \mathcal{S}_9)$ and to apply adjunction. The computation of the Chern class of the ambient space is straightforward, but lengthy and little illuminating. We simply use the result (C.2) and refer to appendix C for details. Then we use adjunction as in (3.21) to obtain the total Chern class of $\hat{X}_{SU(5)}$. We present here the results of the second Chern class $c_2(\hat{X}_{SU(5)})$ that reads

$$c_{2}(\hat{X}_{SU(5)}) = c_{2}(\hat{X}) - c_{1}(4D_{1} + 6D_{2} + 2D_{3} + D_{4}) + 2D_{1}D_{2} + D_{2}^{2} + 2D_{3}D_{4}$$

$$+ \mathcal{S}_{SU(5)}(D_{1} + D_{2} + D_{4} - 2S_{Q}) + \mathcal{S}_{7}(D_{1} + 2D_{2} - D_{4}) + \mathcal{S}_{9}(D_{1} + D_{2} + D_{4}),$$
(6.43)

where $c_2(\hat{X})$ denotes the second Chern class (3.22) on \hat{X} . As a sanity check we take the blow-down limit by formally setting $D_i \to 0$ and $\mathcal{S}_{SU(5)} \to 0$ in which we precisely recover the Chern class (3.22) before resolution.

Next we calculate the Euler number on $\hat{X}_{SU(5)}$. For this purpose we can either bootstrap ourselves and use (6.43) in combination with the relation (3.28) or calculate the fourth Chern class on $\hat{X}_{SU(5)}$ directly by expanding (C.2) to higher order in the adjunction formula expression for the Chern class for $\hat{X}_{SU(5)}$. In either case we obtain

$$\chi(\hat{X}_{SU(5)}) = \chi(\hat{X}) - 3 \int_{B} S_{SU(5)} \left(52c_{1}^{2} - c_{1}(41S_{SU(5)} + 17S_{7} + 31S_{9}) + 10S_{SU(5)}^{2} + S_{SU(5)}(5S_{7} + 13S_{9}) + 5S_{7}^{2} + 2S_{7}S_{9} + 5S_{9}^{2} \right),$$
(6.44)

where $\chi(\hat{X})$ denotes the Euler number (3.24) of \hat{X} before the blow-up, to which our result (6.44) specializes correctly in the limit $D_i, \mathcal{S}_{SU(5)} \to 0$.



Figure 9. Number of independent surfaces $h_V^{(2,2)}(\hat{X}_{SU(5)})$ for the various values of allowed n_7 , n_9 . On the blue boundary and at $n_7 = n_9 = 0$ we have a holomorphic zero section.

The full cohomology ring of $\hat{X}_{SU(5)}$: $B = \mathbb{P}^3$

Next we compute the full vertical cohomology of $\hat{X}_{SU(5)}$ with base $B = \mathbb{P}^3$. In this case, there is only one divisor in the base, the hyperplane H_B , thus, we have $D_{\alpha} = H_B$. Furthermore, the resolved SU(5)-singularity of $\hat{X}_{SU(5)}$ is located over $\mathcal{S}^b_{SU(5)} = H_B$.

The Hodge numbers are calculated to be

$$h^{(1,1)}(\hat{X}_{SU(5)}) = h^{(3,3)}(\hat{X}_{SU(5)}) = 8, \qquad h_V^{(2,2)}(\hat{X}_{SU(5)}) \in [10,13],$$
 (6.45)

where the number of independent surfaces as in the Abelian case depends on the values of n_7 , $n_9 = 0$, see figure 9. As the basis of $H^{(1,1)}$, we use the following divisors

$$H^{(1,1)}(\hat{X}) = \langle H_B, S_P, S_Q, S_R, D_i \rangle, \qquad (6.46)$$

that fixes by Poincaré duality the basis of $H_V^{(3,3)}(\hat{X}_{SU(5)})$. All quartic intersections of the divisors (6.46) can be found in (D.1).

Then we compute the grade two piece $R^{(2)}$ of the ring (6.41). We obtain 64 different monomials in two divisors, but due to equivalence relations in (6.41) at most 13 of them are independent, with jumps along the boundary of the allowed region of n_7 , n_9 in figure 9. A carefully chosen basis of $H_V^{(2,2)}(\hat{X}_{SU(5)})$, thus, has to have always the correct number of surfaces for the entire allowed region. The analysis of appendix E proves that one convenient basis is given by

$$H_{V}^{(2,2)}(\hat{X}_{SU(5)}) = \langle H_{B}^{2}, H_{B} \cdot S_{P}, H_{B} \cdot \sigma(\hat{s}_{Q}), H_{B} \cdot \sigma(\hat{s}_{R}), S_{P} \cdot \sigma(\hat{s}_{R}), \sigma(\hat{s}_{Q}) \cdot \sigma(\hat{s}_{R}), \\ \sigma(\hat{s}_{Q})^{2}, H_{B} \cdot D_{1}, H_{B} \cdot D_{2}, H_{B} \cdot D_{3}, H_{B} \cdot D_{4}, D_{2} \cdot D_{4}, D_{1}^{2} \rangle.$$
(6.47)

At the boundaries of figure 9, one or more elements in $H_V^{(2,2)}(\hat{X}_{SU(5)})$ become linearly dependent. Thus, they can be expressed in terms of as smaller basis that one obtains

by dropping the linearly dependent elements from (6.47). The expansions of the linearly dependent elements in terms of this reduced basis for the various boundary components of figure 9 can be found in appendix E. The result of this analysis is that we can obtain a basis on all boundaries from the basis 6.47 by successively dropping the elements $S_P \cdot \sigma(\hat{s}_R)$, $H_B \cdot \sigma(\hat{s}_Q)$ and $D_2 \cdot D_4$. This way we obtain, as required, in steps twelve, eleven and ten independent surfaces on the relevant boundary components.

7 G_4 -flux & chiralities on fourfolds with $SU(5) \times U(1)^2$

With the cohomology ring at hand we finally calculate the general G_4 -flux, see section 7.1, and 4D chiralities in this section. As before, the lack of knowledge of all the matter surfaces forces us to employ the 3D Chern-Simons terms and M-/F-theory duality to determine the remaining chiralities, cf. section 7.2. It turns out again, that the Chern-Simons terms by themselves are sufficient to determine all the chiralities. The chiralities determined by geometric techniques for a subset of matter therefore provide an independent check. We also check that for the obtained spectrum all four-dimensional anomalies are cancelled. Full explicit results are again presented for the base $B = \mathbb{P}^3$. We also present one concrete toric examples in section 7.3.

There is one accompanying appendix F with further details on the G_4 -flux and the full expressions for the 4D chiralities.

We recall that there can be non-flat fibers at codimension three in $\hat{X}_{SU(5)}$. As mentioned before, these can be avoided for general bases B, but are generically present for $B = \mathbb{P}^{3}$.²³ We deal with these complications here by forbidding a chiral excess of the additional light degrees of freedom associated to the non-flat fiber. This is achieved by requiring one additional condition on the G_4 -flux, namely the vanishing of the integral (6.1). As we demonstrate here, this ensures that the 4D chiralities can successfully be determined both via geometric techniques and M-/F-theory Chern-Simons terms, and that the resulting chiral spectrum cancels all anomalies of the four-dimensional quantum field theory.

7.1 G_4 -flux on fourfolds with two rational sections & SU(5)

The construction of G_4 -flux presented in this section is very similar to the one of section 5. Therefore, we will keep the following discussion as brief as possible and choose, as before, the base $B = \mathbb{P}^3$ to demonstrate our techniques.

We begin by expanding the G_4 -flux in the generically 13-dimensional basis (6.47) of the cohomology group $H_V^{(2,2)}(\hat{X}_{SU(5)})$, see (4.3). We note that the dimensionality of $H_V^{(2,2)}(\hat{X}_{SU(5)})$ jumps at the boundaries of figure 9, however, our general formulas for the G_4 -flux derived in this section will remain valid. Then we enforce the generically six independent flux-conditions (4.10). Again we emphasize, that unlike in compactifications with a holomorphic zero section, the CS-terms Θ_{00}^M , Θ_{0i}^M must not be required to vanish, nor vanish automatically.

²³As mentioned above, we can forbid these points by setting $S_7 = 0$, however, at the cost of losing many representations of the general spectrum obtained from $\hat{X}_{SU(5)}$, too.

Since we are imposing generically six independent conditions 4.10 on the 13-parameter G_4 -flux, we obtain a seven-dimensional G_4 -flux,

$$G_4 = a_3 G_4^{(3)} + a_4 G_4^{(4)} + a_5 G_4^{(5)} + a_6 G_4^{(6)} + a_7 G_4^{(7)} + a_{12} G_4^{(12)} + a_{13} G_4^{(13)},$$
(7.1)

where the a_i denote the free parameters of the G_4 -flux. The independent fluxes $G_4^{(i)}$ read

$$\begin{aligned} G_{4}^{(3)} &= H_{B} \cdot \sigma(\hat{s}_{Q}), \qquad G_{4}^{(4)} = H_{B} \cdot \sigma(\hat{s}_{R}), \qquad G_{4}^{(12)} = D_{4} \cdot (D_{2} - 4H_{B}), \\ G_{4}^{(5)} &= -2H_{B}^{2}n_{9}(4 - n_{7} + n_{9}) + S_{P} \cdot \sigma(\hat{s}_{R}), \\ G_{4}^{(6)} &= \frac{1}{25} \Big[4n_{7}D_{1} \cdot H_{B} + 8n_{7}D_{2} \cdot H_{B} + 12n_{7}D_{3} \cdot H_{B} + 6D_{4}n_{7} \cdot H_{B} \\ &+ 50 \cdot H_{B}^{2}(4 - n_{7} + n_{9})(2 + n_{9}) + 25H_{B} \cdot S_{P}(4 - n_{7} + n_{9}) + 25\sigma(\hat{s}_{Q}) \cdot \sigma(\hat{s}_{R}) \Big], \\ G_{4}^{(7)} &= \frac{1}{125} \Big[-2D_{3} \cdot H_{B}(282 + 31n_{7} - 35n_{9}) - 2D_{1} \cdot H_{B}(144 + 7n_{7} - 15n_{9}) \\ &- 4D_{2} \cdot H_{B}(144 + 7n_{7} - 15n_{9}) + D_{4} \cdot H_{B}(-282 - 31n_{7} + 35n_{9}) \\ &+ 125 \cdot H_{B}^{2} \left(\frac{136}{5} + n_{9}(4 - n_{7} + n_{9}) \right) + 850 \cdot H_{B} \cdot S_{P} + 125\sigma(\hat{s}_{Q})^{2} \Big], \\ G_{4}^{(13)} &= \frac{1}{5} \Big[5D_{1}^{2} + 18D_{1} \cdot H_{B} - 24D_{2} \cdot H_{B} - 16D_{3} \cdot H_{B} - 8D_{4} \cdot H_{B} + 40 \cdot H_{B}^{2} \\ &+ 3n_{9}D_{1} \cdot H_{B} + 6n_{9}D_{2} \cdot H_{B} + 4n_{9}D_{3} \cdot H_{B} + 2n_{9}D_{4} \cdot H_{B} + 10H_{B} \cdot S_{P} \Big] \end{aligned}$$

The general G_4 -flux (7.1) specializes correctly at the boundaries of figure 9 and the concrete expressions for it can be obtained at every boundary component, completely analogous to the discussion of section 5.1. The necessary analysis of the behavior of the basis (6.47) along with the homology relations between linearly dependent elements on the respective boundaries can be found in appendix E.

7.2 4D chiralities from matter surfaces & 3D CS-terms

In this section we finally calculate the chiralities of the total matter content of the Ftheory compactification on $\hat{X}_{SU(5)}$, that we have determined in section 6.1. We recall that there are six singlets, five different 5-representations and one 10, cf. section 6.1. After having determined these chiralities, we check cancellation of 4D anomalies at the end of this section. We find that all anomalies are cancelled in general.

As mentioned before, all chiralities can in principle be calculated from the G_4 -flux in (7.1) and the matter surfaces by evaluating the index (5.12). However, as in the discussion of section (5.2), we only have the explicit homology classes of the six matter surfaces (3.7) and (6.32). Again, we can obtain all chiralities by combining this information with the matching of 3D Chern-Simons terms as outline in section 5.2.1. We first calculate the chiralities from the six matter surfaces for the general G_4 -flux (7.1) and then impose the non-flat fiber condition (6.1). Only then it is possible to obtain the remaining chiralities from the matching of 3D CS-terms. The chiralities for the six representations whose surfaces $C_{\mathbf{R}}^{\mathbf{w}}$ we know, cf. (6.33) and (6.32), are calculated for the general G_4 -flux (7.1) employing (5.12) as

$$\begin{split} \chi(\mathbf{5}_{(-\frac{2}{5},1)}) &= -\frac{1}{500} n_7 \left[-50a_3 + 125a_4 + 5(21n_7 - 5(8 + 3n_9))a_6 \\ &+ (158 - 111n_7 + 115n_9)a_7 + 500a_{12} + 50(n_9 - 4)a_{13} \right], \\ \chi(\mathbf{5}_{(\frac{3}{5},0)}) &= -\frac{1}{500} (4 - n_7 + n_9) \left[75a_3 + 125a_4 + 5(21n_7 - 65n_9 - 160)a_6 \\ &+ (14n_7 - 135n_9 - 742)a_7 + 50(n_9 - 24)a_{13} \right], \\ \chi(\mathbf{10}_{(\frac{1}{5},0)}) &= \frac{1}{125} \left[-25a_3 - 5(2n_7 - 5(4 + n_9))a_6 + (134 + 7n_7 + 20n_9)a_7 \\ &+ 125n_7a_{12} + 50(-27 + 5n_7 - 2n_9)a_{13} \right], \\ \chi(\mathbf{1}_{(-1,1)}) &= -\frac{1}{20}n_7(2 + n_7 - n_9) \left[-5a_3 + 5a_4 + (5n_7 - 4)a_6 + (58 - 10n_7 + 5n_9)a_7 \right], \\ \chi(\mathbf{1}_{(0,2)}) &= \frac{1}{100}n_7n_9 \left[-50a_4 + 50(n_7 - n_9 - 4)a_5 + 5(64 - 15n_7 + 15n_9)a_6 + 162a_7 + 50a_{13} \right], \\ \chi(\mathbf{1}_{(-1,-2)}) &= -\frac{1}{20}(4 - n_7 + n_9)n_9 \left[-5a_3 - 10a_4 + (10n_7 - 20n_9)a_5 \\ &+ (60 - 15n_7a_6 + 25n_9)a_6 + (54 - 5n_7 + 10n_9)a_7 + 10a_{13} \right], \end{split}$$

We then obtain the other chiralities by using the identification (4.5) of 3D CS-levels for $\Theta_{\Sigma\Lambda}$. Both the CS-levels $\Theta_{\Sigma\Lambda}^{M}$ on the M-theory side and the loop-generated CS-levels $\Theta_{\Sigma\Lambda}^{F}$ on the F-theory side are readily computed. All we have to know for the computation of the CS-terms $\Theta_{\Sigma\Lambda}^{F}$ is the KK-charge of the matter representations. We recall from section 6.1 that only the singlet $\mathbf{1}_{(-1,-2)}$ has a non-trivial KK-charge $q_{KK} = 2$ and thus a shifted sign-function (4.17) with k = -2. All the other matter representations have $q_{KK} = 0$ and a symmetric sign-function. Then, all the computations are straightforward but lengthy. Thus, we refer the interested reader to appendix F, where the results of these calculations are presented.

We make a short stop to show some interesting results/checks of the CS-levels and the discussion in section 4.2.2. First, we calculate the CS-term Θ_{00} for the KK-vector. On the F- theory side, we have to evaluate the loop-corrections of KK-states as in (4.18). Since only the singlet $\mathbf{1}_{(-1,-2)}$ has a non-trivial KK-charge, we again obtain the result (5.16). On the M-theory side, Θ_{00}^{M} is calculated directly from G_4 -flux and we obtain

$$\Theta_{00}^{M} = -\frac{1}{20} (4 - n_7 + n_9) n_9 \left[-5a_3 - 10a_4 + (10n_7 - 20n_9)a_5 + (60 - 15n_7a_6 + 25n_9)a_6 + (54 - 5n_7 + 10n_9)a_7 + 10a_{13} \right]$$

= $\chi(\mathbf{1}_{(-1,-2)}),$ (7.4)

where we used (7.3) in the last equality. Thus, we confirm the result (5.16).

Similarly, we check the matching of the CS-levels Θ_{0m} . As before, a matching of the M- and F-theory expression for these CS-levels requires the relation (5.17) to hold,

$$\frac{1}{4} \int_{\hat{X}} S_P \cdot \sigma(\hat{s}_Q) \cdot G_4 \stackrel{!}{=} -\chi(\mathbf{1}_{(-1,-2)}), \qquad \frac{1}{4} \int_{\hat{X}} S_P \cdot \sigma(\hat{s}_R) \cdot G_4 \stackrel{!}{=} -2\chi(\mathbf{1}_{(-1,-2)}).$$
(7.5)

Indeed, we evaluate the flux integral on the left to confirm this equality.

Finally, we comment on the CS-terms Θ_{0i} . From the direct computation of the relevant integrals of the G_4 -flux, cf. (4.4), we obtain $\Theta_{0i}^M = 0$, since $S_P \cdot D_i = 0$ in the homology of $\hat{X}_{SU(5)}$, cf. (D.5) in appendix D. This immediately agrees with the field theory result Θ_{0i}^F in (4.8) since there the only charged matter states in 3D with non-trivial KK-charge is the singlet $\mathbf{1}_{(-1,-2)}$, which however has $q_i = 0$ for all *i*.

Coming back to the calculation of chiralities, we recall that for $n_7 > 0$ there is a nonflat fiber, since the codimension three point $s_6 = s_7 = d_4 = 0$ exists, cf. section 6.1.3. If we ignore this fact and try to solve for the chiralities using the matching condition 4.5 we find

$$\Theta_{i=2,j=4}^{F} = \Theta_{i=2,j=4}^{M}, \qquad \Theta_{i=4,m=1}^{F} = \Theta_{i=4,m=1}^{M}, \qquad \Theta_{i=4,m=2}^{F} = \Theta_{i=4,m=2}^{M}, \\
0 = \frac{2}{5}n_{7}a_{12}, \qquad 0 = 2n_{7}a_{12}, \qquad 0 = n_{7}a_{12},$$
(7.6)

where the left hand sides are functions of chiralities. These equations are obviously inconsistent on the F-theory side. However, this is not surprising since the existence of a non-flat fiber implies the presence of additional light states, that we have not taken into account on the field theory side. Since these states are probably unwanted in setups aimed at the construction of phenomenologically appealing F-theory compactifications, we require the absence of a *chiral excess* of these light states. As we see from (7.6) the appropriate condition to impose is to demand the vanishing of the second line.

It is satisfying to see that the vanishing of this second line is precisely the non-flat fiber condition (6.1) on the G_4 -flux. Indeed, using the general G_4 -flux (7.1) and the homology class of the loci of the non-flat fiber on the left of equation (6.36), we obtain

$$\int_{X_5} G_4 D_4 \cdot \mathcal{S}_7 \cdot [s_6] = -8n_7 a_{12} \stackrel{!}{=} 0, \qquad (7.7)$$

which implies exactly a vanishing of the second line in (7.6). Thus, we see that we obtain a consistent set of equations in (7.6) if we impose the non-flat fiber condition (6.1), as claimed.

In summary, for $n_7 = 0$ the vanishing of (7.7) does not impose any additional conditions on the G_4 -flux, whereas for for $n_7 > 0$ its vanishing implies the additional condition

$$a_{12} \stackrel{!}{=} 0.$$
 (7.8)

Thus, the number of free parameters in the G_4 -flux in (7.1) is reduced to six. Furthermore, as we demonstrate in the following imposing the condition (7.7) will yield a consistent 4D chiral spectrum with all anomalies cancelled. For a better overview over the conditions we impose, we summarize the number of independent elements in the G_4 -flux graphically in figure 10.

Imposing now the condition (7.8) we solve the first line of (7.6). The solutions are straightforwardly obtained, but lengthy and can be found in appendix F. However, instead of parametrizing the chiralities in the parameters a_i of the G_4 -flux in (7.1), we can alternatively express the chiralities in terms of the chiralities of the 5- representations and one singlet. This will, on the one hand, reduce the length of the expressions we obtain and, on



Figure 10. There are six independent constraints on the G_4 -flux from the CS-terms. In order to obtain an anomaly free spectrum, for $n_7 > 0$, we force the vanishing condition (7.7), respectively, (7.8). The small number in orange next to each point is the final number of parameters in G_4 -flux. The maximal number is six in the bulk.

the other hand, guarantee that all chiralities are parametrized in terms of integers.²⁴ We then obtain the following chiralities,

$$\chi(\mathbf{5}_{(-\frac{2}{5},0)}) = c_1, \qquad \chi(\mathbf{5}_{(-\frac{2}{5},1)}) = c_2, \qquad \chi(\mathbf{5}_{(\frac{3}{5},0)}) = c_3, \chi(\mathbf{5}_{(\frac{3}{5},1)}) = c_4, \qquad \chi(\mathbf{5}_{(-\frac{2}{5},-1)}) = c_5, \qquad \chi(\mathbf{1}_{(-1,-2)}) = c_6, \qquad (7.9)$$

with integral parameters c_i in terms of which the other chiralities are expressed as

$$\chi(\mathbf{10}_{(\frac{1}{5},0)}) = -c_1 - c_2 - c_3 - c_4 - c_5, \qquad \chi(\mathbf{1}_{(0,2)}) = c_6 + (c_2 + c_4 - c_5)n_9, \chi(\mathbf{1}_{(1,0)}) = -22c_2 - c_3 - 3(2(c_4 + c_5) + 3c_6) + 2(c_2 + c_4 - c_5)n_7 + 2c_1(n_9 - 6) - 2(c_4 - 2c_5)n_9, \chi(\mathbf{1}_{(1,1)}) = -14c_5 + 3c_6 + 2c_5n_7 - (c_2 + 3c_5)n_9 + c_4(5 - n_7 + n_9) + c_1(n_7 - 2(4 + n_9)), \chi(\mathbf{1}_{(0,1)}) = 14c_4 - 5c_5 - 2c_6 + c_2(21 - 2n_7) - 2c_5n_7 - 2c_4n_9 + 4c_5n_9 + c_1(8 - 2n_7 + 2n_9), \chi(\mathbf{1}_{(-1,1)}) = -2c_2 - c_6 + c_1n_7 + 2c_2n_7 + c_4n_7 - (c_2 + c_4 - c_5)n_9.$$
(7.10)

We note that the chirality of the representation $\mathbf{10}_{(\frac{1}{5},0)}$ has the opposite sign of the c_i . For positive c_i this would yield a negative chirality of left-handed Weyl fermions in $\mathbf{10}_{(\frac{1}{5},0)}$, which has to be understood as left-handed Weyl fermions in the representation $\overline{\mathbf{10}}_{(-\frac{1}{5},0)}$. We adapt this convention in the following.

7.2.1 Anomaly cancellation

We conclude this general discussion of the spectrum by checking the cancellation of 4D anomalies. For a short review, see section 5.3, or see [44] for more details.

²⁴As mentioned before, the determination of the integral basis of $H_V^{(2,2)}(\hat{X}_{SU(5)})$ is a subtle and difficult problem, that requires the application of more advanced techniques like mirror symmetry.

Having derived the spectrum in the previous paragraph, all we have to calculate to check anomaly cancellation are the gaugings $\Theta_{\alpha m}$ for m = 1, m = 2, cf. (5.22). The results of this computations in terms of the parameters a_i of the G_4 -flux can be found in appendix F. As mentioned before, we can alternatively use the chiralities (7.9) as parameters. Then we obtain

$$\Theta_{\alpha,m=1} = -2(c_1 + c_2 + c_5), \qquad \Theta_{\alpha,m=2} = 2(c_2 + c_4 - c_5).$$
(7.11)

The coefficients b_{mn}^{α} of the GS-counter terms change slightly compared to the U(1)×U(1)case of section 5.3 because of the altered Shioda map (6.17), with and a^{α} remaining the same. We obtain

$$b_{mn}^{\alpha} = \begin{pmatrix} \frac{34}{5} & 4 - n_7 + n_9\\ 4 - n_7 + n_9 & 8 + 2n_9 \end{pmatrix}, \qquad a^{\alpha} = -4.$$
(7.12)

We readily check that all anomalies in (5.22) cancel beautifully. First, the purely non-Abelian anomaly vanishes trivially

$$\chi(\overline{\mathbf{10}}_{(-\frac{1}{5},0)}) = \sum_{(q_1,q_2)} \chi(\mathbf{5}_{(q_1,q_2)}), \tag{7.13}$$

where the sum is over all different 5-representations, that are labeled by their charges (q_1, q_2) , and where we used the representation $\overline{\mathbf{10}}_{(-\frac{1}{5},0)}$ as explained below (7.10). The purely Abelian anomalies are

$$A_{1,1,1}^{\mathrm{U}(1)} := -\frac{17}{5}(c_1 + c_2 + c_5),$$

$$A_{1,1,2}^{\mathrm{U}(1)} := \frac{1}{15}(17c_4 - 37c_5 + 5c_5n_7 + c_2(5n_7 - 5n_9 - 3) + 5c_1(n_7 - n_9 - 4) - 5c_5n_9),$$

$$A_{1,2,2}^{\mathrm{U}(1)} := \frac{1}{3}(-8c_5 - c_2n_7 + c_5n_7 - 2c_5n_9 - c_1(4 + n_9) + c_4(4 - n_7 + n_9)),$$

$$A_{2,2,2}^{\mathrm{U}(1)} := (c_2 + c_4 - c_5)(4 + n_9),$$
(7.14)

The mixed non-Abelian-Abelian, and Abelian-gravitational anomalies read

$$A_1^{\mathrm{U}(1)-\mathrm{SU}(5)} = A_1^{\mathrm{U}(1)-\mathrm{grav}} : \frac{1}{2}(-c_1 - c_2 - c_5), \qquad A_2^{\mathrm{U}(1)-\mathrm{SU}(5)} = A_2^{\mathrm{U}(1)-\mathrm{grav}} : \frac{1}{2}(c_2 + c_4 - c_5).$$
(7.15)

Using the gaugings (7.11) and the coefficients (7.12) of the GS-terms, we confirm that the 4D anomaly cancellation conditions (5.22) hold.

7.3 A toric example

In this concluding subsection we construct explicitly a toric model with all characteristics described in this section, i.e. an $SU(5) \times U(1)^2$ 4D gauge group, the full spectrum computed in sections 6.1, 7.2 and with a non-flat fiber.

The concrete fourfold $\ddot{X}_{SU(5)}$ we construct has $n_7 = n_9 = 4$ in figure 5 and has $S_{SU(5)} = H_B$. It is engineered following the general guideline of appendix G. The vertices

-							
	divisor class		es	ertice	v		variable
]	H_B	0	0	1	1	1	z_0
	H_B	1	0	0	0	-1	z_1
	H_B	-1	-1	0	-1	0	z_3
	$H_B - D_1 - D_2 - D_3 - D_4$	0	1	-1	0	0	z_2
]	D_1	0	0	-1	0	0	d_1
(7.16)	D_2	-1	-1	-1	0	0	d_2
(1.10)	D_3	-2	-1	-1	0	0	d_3
	D_4	-1	0	-1	0	0	d_4
]	$-3H_B + H - D_2 - D_3 - E_1 - E_2$	0	1	0	0	0	u
	$H - D_1 - D_2 - D_3 - D_4 - E_2$	1	0	0	0	0	v
	$H - D_1 - 2D_2 - 3D_3 - 2D_4 - E_1$	-1	-1	0	0	0	w
	E_1	-1	0	0	0	0	e_1
	E_2	1	1	0	0	0	e_2

specifying the reflexive polytope, along with their associated homogeneous coordinates and divisor classes, are as follows,

Here we follow the notation of section 6.1, i.e. H_B is the hyperplane of \mathbb{P}^3 , the D_i are the Cartan divisors of the SU(5) and H, E_1 , E_2 are the classes of the dP_2 -fiber.

The Euler number is calculated as

$$\chi(\ddot{X}_{\rm SU(5)}) = 1110 \tag{7.17}$$

which agrees perfectly with general formula (6.44), where we have used $\chi(\hat{X}) = 1632$ and $S_{SU(5)} = H_B$, $S_7 = S_9 = 4H_B$. For the construction of the corresponding toric variety we have to choose the right triangulation so that the toric Stanley-Reissner ideal is of the same from as the Stanley-Reissner ideal in (6.12). The toric divisors of the sections are given by (3.32) for $n_9 = 4$ and the Shioda maps (6.17) read

$$\sigma(\hat{s}_Q) = S_Q - S_P - 4H_B + \frac{1}{5}(2D_1 + 4D_2 + 6D_3 + 3D_4)$$
(7.18)

$$\sigma(\hat{s}_R) = S_R - S_P - 8H_B; \tag{7.19}$$

The full basis of $H_V^{(2,2)}$ is given by the 13 elements in equation (6.47). The G_4 -flux follows from (7.1) by inserting $n_7 = n_9 = 4$. Since we have $n_7 \neq 0$, we have to impose the non-flat fiber condition (7.8). Our main interest lies in the chiralities, which are then either computed torically or follow from the general formulas in 7.2 and appendix F as

$$\chi(\mathbf{5}_{(-\frac{2}{5},0)}) = \frac{1}{2}a_3 - \frac{2}{25}(40a_6 + 63a_7 + 75a_{13}),$$

$$\chi(\mathbf{5}_{(-\frac{2}{5},1)}) = \frac{1}{125}(50a_3 - 125a_4 + 80a_6 - 174a_7),$$

$$\chi(\mathbf{5}_{(\frac{3}{5},0)}) = \frac{1}{250}(-225a_3 + 1640a_6 + 1888a_7),$$

$$\chi(\mathbf{5}_{(\frac{3}{5},1)}) = -\frac{2}{5}(-5a_3 + 5a_4 + 16a_6 + 38a_7),$$

$$\chi(\mathbf{5}_{(-\frac{2}{5},-1)}) = \frac{4}{5}a_3 + 2a_4 + 8a_5 - \frac{468}{25}a_6 - \frac{1608}{125}a_7 + 4a_{13},$$

$$\chi(\mathbf{10}_{(\frac{1}{5},0)}) = -\frac{1}{5}a_3 + \frac{32}{25}a_6 + \frac{242}{125}a_7 - 6a_{13},$$

(7.20)

for the non-trivial non-Abelian representations and

$$\begin{split} \chi(\mathbf{1}_{(1,0)}) &= -\frac{27}{2}a_3 - 32a_5 + \frac{596}{5}a_6 + \frac{3128}{25}a_7 + 8a_{13}, \\ \chi(\mathbf{1}_{(1,1)}) &= -13a_3 - 13a_4 - 88a_5 + 200a_6 + \frac{4226}{25}a_7 + 16a_{13}, \\ \chi(\mathbf{1}_{(0,1)}) &= -29a_4 - 88a_5 + \frac{836}{5}a_6 + \frac{2006}{25}a_7 + 28a_{13}, \\ \chi(\mathbf{1}_{(-1,1)}) &= -\frac{2}{5}(-5a_3 + 5a_4 + 16a_6 + 38a_7), \\ \chi(\mathbf{1}_{(0,2)}) &= -8a_4 - 32a_5 + \frac{256}{5}a_6 + \frac{648}{25}a_7 + 8a_{13}, \\ \chi(\mathbf{1}_{(-1,-2)}) &= 4a_3 + 8a_4 + 32a_5 - 80a_6 - \frac{296}{5}a_7 - 8a_{13}, \end{split}$$
(7.21)

for the singlets. As before, it is convenient to parametrize the chiralities as in the previous subsections, in terms of the c_i . We then obtain

$$\chi(\mathbf{5}_{(-\frac{2}{5},0)}) = c_{1}, \qquad \chi(\mathbf{5}_{(-\frac{2}{5},1)}) = c_{2}, \qquad \chi(\mathbf{5}_{(\frac{3}{5},0)}) = c_{3}, \qquad (7.22)$$

$$\chi(\mathbf{5}_{(\frac{3}{5},1)}) = c_{4}, \qquad \chi(\mathbf{5}_{(-\frac{2}{5},-1)}) = c_{5}, \qquad \chi(\mathbf{1}_{(-1,-2)}) = c_{6}, \qquad (7.22)$$

$$\chi(\mathbf{10}_{(\frac{1}{5},0)}) = -c_{1}-c_{2}-c_{3}-c_{4}-c_{5}, \qquad \chi(\mathbf{1}_{(1,0)}) = -4c_{1}-14c_{2}-c_{3}-6c_{4}+2c_{5}-3c_{6}, \qquad (1_{(0,1)}) = 8c_{1}+13c_{2}+6c_{4}+3c_{5}-2c_{6}, \qquad \chi(\mathbf{1}_{(-1,1)}) = 4c_{1}+2c_{2}+4c_{5}-c_{6}, \qquad \chi(\mathbf{1}_{(0,2)}) = 4c_{2}+4c_{4}-4c_{5}+c_{6}, \qquad \chi(\mathbf{1}_{(1,1)}) = -12c_{1}-4c_{2}+5c_{4}-18c_{5}+3c_{6}.$$

It follows that all 4D anomalies are cancelled.

8 Conclusions and future directions

In this paper we have advanced the program on F-theory compactifications on elliptic Calabi-Yau manifolds with rank two Mordell-Weil group to four-dimensional chiral compactifications.

The analysis of resolved Calabi-Yau elliptic fibrations with dP_2 -elliptic fiber has now been performed for Calabi-Yau fourfolds over a general three dimensional base B, extending earlier results in six dimensions [46, 51]. We study general compactifications with U(1) × U(1) and a specific SU(5) × U(1) × U(1) gauge symmetry. We determined the general matter representations associated with the codimension two singularities of the fibration as well as its Yukawa points. However, at present the geometric techniques, relying on the use prime ideals, that we employed allow us to determine all matter curves and only a subset of matter surfaces. As a next step, we determine explicitly the general G_4 -flux of these resolved Calabi-Yau fourfolds. Since we considered Calabi-Yau fourfolds with a non-holomorphic zero section, we encountered the novel problem of having to define G_4 -flux over these manifolds. For this purpose we had to derive the general F-theory conditions on G_4 -flux on Calabi-Yau manifolds with a non-holomorphic zero section. We have formulated these conditions in terms of Chern-Simons terms on the Coulomb branch of the effective theory obtained by compactifying on a circle. The relevance of Kaluza-Klein states generating new Chern-Simons terms in order to ensure a consistent M-/F-theory duality mapping in 3D has been pointed out

After this interlude on G_4 -fluxes in F-theory we constructed the most general G_4 -flux. We presented explicit calculations for the three dimensional base $B = \mathbb{P}^3$. In both the $U(1) \times U(1)$ and $SU(5) \times U(1) \times U(1)$ case we could determine certain chiralities directly by evaluating the chiral index (5.12) directly while the remaining matter chiralities were determined by a subset of the 3D Chern-Simons terms. It turned out that Chern-Simons terms are sufficient to determine chiralities of all the matter multiplets if also the new, Kaluza-Klein generated Chern-Simons terms are taken into account. The geometric techniques allowed an important, independent consistency check. The complications of non-flat fibers could be successfully circumvented by imposing one additional condition on the G_4 flux, namely the vanishing of the integral of the G_4 -flux over the non-flat fiber. We also presented concrete explicit examples using toric geometry. We emphasize that our techniques are in line, but not restricted to toric geometry. In particular, we could calculate the most general G_4 -flux and chiralities, along with the proof of anomaly cancellation, for the entire class of all elliptically fibered Calabi-Yau fourfolds, cf. figure 2 and figure 5, with dP_2 -elliptic fiber, an additional SU(5) GUT-sector and a fixed base B, here chosen to be $B = \mathbb{P}^3$ for simplicity.

There paper leaves room for a number of further studies and improvements:

- The geometric techniques developed here apply to general elliptic Calabi-Yau fourfolds, and they are *not* restricted to toric examples. In particular, we could derived closed formulas for e.g. the Euler number, the G_4 -flux and the 4D chiralities that explicitly depend on the divisors S_7 , S_9 , introduced in section 2.2, which label the members in the family of all inequivalent Calabi-Yau fourfolds obtained by varying the topologically data specifying their elliptic fibration. It would be desirable to obtain these formula for an arbitrary base B. Our results are independent of the existence of toric realizations of the Calabi-Yau fourfolds. We emphasize that employing toric geometry techniques one can typically study only one polytope at a time, which obscures the dependence on S_7 , S_9 .
- The finite number of choices for the divisors S_7 , S_9 on the F-theory side fix the topology of the fibration of dP_2 over the base B. It would be interesting to understand these new degrees of freedom from the point of view of the heterotic string. In particular it would be important to understand how the parameters in S_7 , S_9 enter the heterotic vector bundle and potentially modify the spectral cover construction [85].
- Although all 4D chiralities could still be derived by a combination of geometric and field theoretic techniques, we were at this point unable to determine explicitly the matter surfaces for a subset of representations. It would be desirable to derive these missing homology classes purely geometrically by further extending the geometric techniques of section 3.1.1. For this analysis important lessons might be learned from a geometrical interpretation of the field theoretically derived G_4 -flux conditions in section 4.2.

- While explicit results were presented for a specific three-dimensional base $B = \mathbb{P}^3$, it would be important to present similar computations for systematically classified three-dimensional bases, extending a similar analysis in 6D [86, 87]. These studies are also motivated by a search for SU(5) GUT models with flat fibers.
- In this paper we have not systematically performed a classification of SU(5) GUT models. We presented models for a specific SU(5) construction with a specific base B = P³, only. The example does not have a flat fibration at codimension three, which may lead to an infinite tower of massless states (a so-called tensionless string spectrum). We removed chiral states in the tower by further constraining the G₄-flux. It would be important to classify SU(5) constructions by carrying over the Tate classification to dP₂-elliptic fibrations, and to construct a base B where the fibration can be engineered to be flat. For efforts in these directions, primarily employing toric techniques, see [46, 47]. We note that the classification of SU(n)-singularities for general n should work similarly to the SU(5) case and thus will be facilitated by techniques developed in this work.
- The techniques developed in this paper pave the way to phenomenological studies of chiral four-dimensional models with rank-two Abelian sectors. However in order to achieve these goals a number of further details have to be addressed. First, the quantization of G_4 -flux is not well understood in general. However, as pointed out in section 4 we can choose the chiralities of our matter multiplets to parameterize the spectrum which should in turn also yield integer values for the flux-quanta in the G_4 -flux. Second, the constraints imposed by the self-duality condition on the G_4 -flux and by D3-brane tadpole cancellation have to be investigated. Those are important topics for future research.

Acknowledgments

We would like to thank Jim Halverson, Albrecht Klemm, Wolfgang Lerche, Song Peng and in particular Thomas W. Grimm for discussions and comments. M.C. and D.K. are grateful to the Theory Division of CERN for hospitality during completion of the project. H.P is grateful to TASI and the University of Colorado, Boulder for hospitality. This research is supported in part by the DOE grant DE-SC0007901 (M.C., H.P., D.K.), the NSF String Vacuum Project Grant No. NSF PHY05-51164 (H.P.), Dean's Funds for Faculty Working Group (M.C. and D.K.), the Fay R. and Eugene L.Langberg Endowed Chair (M.C.) and the Slovenian Research Agency (ARRS) (M.C.).

A Chern classes of dP_2 -elliptic fibrations

In this appendix we analyze the total Chern-class of the total space of dP_2 -fibrations $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$ over a general base B. The following discussions hold in any complex dimension, although we specialize to Calabi-Yau two-, three- and fourfolds in the following.

The final goal of this section is the derivation of general formulas for the Euler number of smooth Calabi-Yau manifolds \hat{X} of complex dimension two, three and four in $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$. For the reader only interested in these formulas, we first state the results of the analysis and refer to the remainder of this section for a detailed derivation. The Euler numbers for Calabi-Yau two-, three- and fourfolds are expressed as the following integrals over the one-, two- respectively, three-dimensional base B of the Calabi-Yau manifold \hat{X} ,

$$d = 2: \quad \chi(X) = 12 \int_{B} c_{1}$$

$$d = 3: \quad \chi(X) = \int_{B} (-24c_{1}^{2} + 8c_{1}\mathcal{S}_{7} - 4\mathcal{S}_{7}^{2} + 8c_{1}\mathcal{S}_{9} + 2\mathcal{S}_{7}\mathcal{S}_{9} - 4\mathcal{S}_{9}^{2})$$

$$d = 4: \quad \chi(X) = 3 \int_{B} \left[24c_{1}^{3} + 4c_{1}c_{2} - 16c_{1}^{2}(\mathcal{S}_{7} + \mathcal{S}_{9}) + c_{1}(8\mathcal{S}_{7}^{2} + \mathcal{S}_{7}\mathcal{S}_{9} + 8\mathcal{S}_{9}^{2}) - \mathcal{S}_{7}\mathcal{S}_{9}(\mathcal{S}_{7} + \mathcal{S}_{9}) \right]$$
(A.1)

Here we denoted the Poincaré dual (1, 1)-forms by abuse of notation by the same symbol as their corresponding divisors S_7 , S_9 in the base B. Furthermore, wedge-products have been omitted for brevity of our notation. We emphasize that these formulas for $\chi(\hat{X})$ are a direct generalization of the formulas in [57] for fibrations by the E_6 elliptic curve, that are obtained as the special case $S_7 = S_9 \equiv 0$.

We prepare the derivation of the Euler numbers (A.1) by a computation of the total Chern class of the ambient space $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$. The total space of the dP_2 -fibration $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$ in (2.7) has the structure of a generalized projective bundle: the two-dimensional fiber is dP_2 and it can be understood as the projectivization of a rank five vector bundle over B by three \mathbb{C}^* -actions. Consequently, the Chern class of the $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$ is calculated analogous to the Chern-class of an ordinary projective bundle by adjunction. Denoting the projective coordinates of the general dP_2 fiber by $[u: v: w: e_1: e_2]$ we follow the assignments (2.2) of line bundles over the base B from the main text. Thus we obtain for the total Chern-class of $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$,

$$c(dP_2^B) = c(B)(1+H-E_1-E_2+\mathcal{S}_9-[K_B^{-1}])(1+H-E_2+\mathcal{S}_9-\mathcal{S}_7)(1+H-E_1)(1+E_1)(1+E_2).$$
(A.2)

Here we suppressed the dependence on S_7 , S_9 for brevity of our expression. By abuse of notation we denote the divisors E_i and H and the first Chern-classes of their associated divisor bundles by the same symbol. In addition, we used $c(B) = 1 + c_1(B) + c_2(B) + c_3(B) + \ldots$ to denote the total Chern class of B. We readily expand (A.2) order by order to obtain the following Chern classes of $dP_2^B(S_7, S_9)$,

$$c_1(dP_2^B) = 3H - E_1 - E_2 + 2S_9 - S_7,$$

$$c_2(dP_2^B) = c_2 - c_1^2 - 2E_1^2 - 3E_2^2 + 2E_1S_7 - 2E_2S_7 - E_1S_9 + 4E_2S_9 - HS_9 + S_7S_9 - S_9^2$$
(A.3)

$$+c_1(4H - 2E_1 - 3E_2 - 2\mathcal{S}_7 + 3\mathcal{S}_9), \qquad (A.4)$$

$$c_{3}(dP_{2}^{B}) = c_{3} - c_{2}(E_{1} + E_{2} - 3H + S_{7} - 2S_{9}) + c_{1}^{2}(H - E_{1} - 2E_{2} - S_{7} + S_{9}) -c_{1}(c_{2} + 2E_{1}^{2} + 3E_{2}^{2} + 2S_{7}(E_{2} - E_{1}) + S_{9}(H + E_{1} - 4E_{2} - S_{7} + S_{9})),$$
(A.5)

$$c_4(dP_2^B) = c_3(3H - E_1 - E_2) + c_2(\mathcal{S}_7(2E_1 - 2E_2) + \mathcal{S}_9(4E_2 - E_1 - H) - 2E_1^2 - 3E_2^2) + c_1c_2(H - E_1 - 2E_2),$$
(A.6)

$$c_5(dP_2^B) = -c_3(2E_1^2 + 3E_2^2) = 5c_3|_B.$$
(A.7)
Here we have used the notation $c_i \equiv c_i(B)$. All divisors in the above expressions are understood as their Poincaré dual (1, 1)-forms, that are given as the first Chern class of the associated divisor line bundle, e.g. $c_1(\mathcal{O}(E_1))$ for $\mathcal{O}(E_1)$. The expressions for the first to third Chern classes are completely general, whereas we have assumed $\dim_{\mathbb{C}}(B) = 3$ for the fourth and fifth Chern class to be able to drop terms containing more than three vertical divisors. In the last line we have used the toric intersections $E_1^2 = E_2^2 = -1$ in (2.4) since $[c_3] \cong (\int_B c_3)pt$ with pt denoting a point on B. All expressions in (A.3) have been reduced modulo the ideal SR' in (3.20).

Now we are in the position to calculate the total Chern class of the Calabi-Yau fourfold \hat{X} . For this purpose we apply adjunction to obtain

$$c(\hat{X}) = \frac{c(dP_2^B)}{1 + 3H - E_1 - E_2 + 2S_9 - S_7}$$
(A.8)

where the numerator is to be computed using (A.2) and the denominator is the Chern class of the anti-canonical bundle of $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$, cf. the first line in (A.3), employing that \hat{X} is the anti-canonical divisor. Expanding out this expression for the total Chern class of the fourfold \hat{X} yields the following individual Chern classes $c_i(\hat{X})$,

$$\begin{split} c_1(\hat{X}) &= 0, \\ c_2(\hat{X}) &= 3c_1^2 + c_2 - 2S_Q^2 - 3S_P^2 + \mathcal{S}_7(2S_Q - 2S_P) + \mathcal{S}_9(3S_P - 2S_Q - S_R + \mathcal{S}_7) \\ &+ c_1(2S_Q + S_P - 2(\mathcal{S}_7 + \mathcal{S}_9 - 2S_R)), \\ c_3(\hat{X}) &= c_3 - 8c_1^3 - c_1c_2 + \mathcal{S}_7(2S_Q^2 - 2S_P^2 - 2S_Q\mathcal{S}_7 - 2S_P\mathcal{S}_7 + 2S_P\mathcal{S}_9 + \mathcal{S}_7\mathcal{S}_9 + \mathcal{S}_9^2) \\ &+ c_1(7S_P^2 + 2S_Q\mathcal{S}_7 + 6S_P\mathcal{S}_7 - 2\mathcal{S}_7^2 + 8S_Q\mathcal{S}_9 - 7S_P\mathcal{S}_9 - 6\mathcal{S}_7\mathcal{S}_9 - 2\mathcal{S}_9^2 + 7\mathcal{S}_9\mathcal{S}_R) \\ &- c_1^2(8S_Q + S_P - 8(\mathcal{S}_7 + \mathcal{S}_9 - S_R)) - 2S_Q\mathcal{S}_9^2 - 2S_R\mathcal{S}_9^2, \\ c_4(\hat{X}) &= c_1^2(8S_Q^2 - 32S_P^2 - \mathcal{S}_7(22S_Q - 18S_P) + \mathcal{S}_9(41S_P - 41S_Q - 40S_R)) - 2S_Q^2\mathcal{S}_7^2 \\ &- 2S_P^2\mathcal{S}_7^2 + 2S_Q\mathcal{S}_7^3 - 2S_P\mathcal{S}_7^3 + 3S_Q^2\mathcal{S}_7\mathcal{S}_9 - S_P^2\mathcal{S}_7\mathcal{S}_9 - 5S_Q\mathcal{S}_7^2\mathcal{S}_9 + 2S_P\mathcal{S}_7^2\mathcal{S}_9 + 2S_Q^2\mathcal{S}_9^2 \\ &- 6S_P^2\mathcal{S}_9^2 - 2S_Q\mathcal{S}_7\mathcal{S}_9^2 - S_P\mathcal{S}_7\mathcal{S}_9^2 - 2S_Q\mathcal{S}_9^3 + 6S_P\mathcal{S}_9^3 - 4\mathcal{S}_9^3S_R + 8c_1^3(4S_Q - S_P + 3S_R) \\ &- c_2(2S_Q^2 + 3S_P^2 - 2S_Q\mathcal{S}_7 + 2S_P\mathcal{S}_7 + 2S_Q\mathcal{S}_9 - 3S_P\mathcal{S}_9 + \mathcal{S}_9S_R) \\ &+ c_1(\mathcal{S}_7(14S_P^2 + 12S_P\mathcal{S}_7 - 11S_P\mathcal{S}_9) + 17S_P^2\mathcal{S}_9 - 19S_P\mathcal{S}_9^2 - 3S_Q^2(2\mathcal{S}_7 + 3\mathcal{S}_9) \\ &+ 8S_Q(\mathcal{S}_7^2 + 2\mathcal{S}_7\mathcal{S}_9 + 2\mathcal{S}_9^2) + 23\mathcal{S}_9^2S_R + c_2(2S_Q + S_P + 4S_R)). \end{split}$$

Here we have dropped terms quartic in divisors on the base B and have reduced all expressions in the quotient ring (3.18). For the evaluation of (A.9) it proves convenient to express all the elements in terms of the divisor classes $S_P = E_2$, $S_Q = E_1$ and $S_R = H - E_1 - E_2 + S_9 + [K_B]$ of the sections of \hat{X} . Then, we can make use of the intersection relations for the sections

$$(S_m^2 + [K_B^{-1}] \cdot S_m) \cdot D_\alpha \cdot D_\beta = 0, \qquad (A.10)$$

where we collectively denote the divisor classes of the sections by S_m . We note that this relation is a direct consequence of the property of the S_m being the classes of sections, as noted before in (2.14), but it is satisfying that it can also be proven directly in the intersection ring (3.18).

In the following we assume that \hat{X} is either complex two-, three- and four-dimensional and for each case employ (A.10) and its lower dimensional analogs to simplify the Chern classes in (A.9) even further. We obtain for the top Chern class $c_d(\hat{X})$ on a *d*-dimensional Calabi-Yau manifold \hat{X} the following results,

$$\begin{split} d &= 2: \quad c_2(\hat{X}) = 4c_1(S_Q + S_P + S_R) + 2\mathcal{S}_7(S_Q - S_P) + \mathcal{S}_9(3S_P - S_R - 2S_Q) \,, \\ d &= 3: \quad c_3(\hat{X}) = -2\mathcal{S}_7^2(S_Q + S_P) + 2\mathcal{S}_7\mathcal{S}_9S_P - 2\mathcal{S}_9^2(S_Q + S_R) - 8c_1^2(S_Q + S_P + S_R) \\ &\quad + c_1(8S_P\mathcal{S}_7 + 8S_Q\mathcal{S}_9 - 7S_P\mathcal{S}_9 + 7\mathcal{S}_9S_R) \,, \\ d &= 4: \quad c_4(\hat{X}) = 2\mathcal{S}_7^3(S_Q - S_P) + \mathcal{S}_7^2\mathcal{S}_9(2S_P - 5S_Q) - \mathcal{S}_7\mathcal{S}_9^2(2S_Q + S_P) - 2\mathcal{S}_9^3(S_Q - 3S_P + 2S_R) \\ &\quad + c_1(14S_P\mathcal{S}_7^2 - 10S_P\mathcal{S}_7\mathcal{S}_9 - 13S_P\mathcal{S}_9^2 + S_Q(10\mathcal{S}_7^2 + 13\mathcal{S}_7\mathcal{S}_9 + 14\mathcal{S}_9^2) + 23\mathcal{S}_9^2S_R) \\ &\quad + 4c_1c_2(S_Q + S_P + S_R) - 8c_1^2(4S_P\mathcal{S}_7 - 3S_P\mathcal{S}_9 + 2S_Q(\mathcal{S}_7 + 2\mathcal{S}_9) + 5\mathcal{S}_9S_R) \end{split}$$

$$+24c_1^3(S_Q+S_P+S_R)+c_2(2S_Q(\mathcal{S}_7-\mathcal{S}_9)-2S_P\mathcal{S}_7+3S_P\mathcal{S}_9-\mathcal{S}_9S_R)$$
(A.11)

where we now in addition also dropped intersections of more than d-1 vertical divisors.

Finally, we are in the position to calculate the Euler number for \hat{X} . We obtain the Euler number for \hat{X} being a twofold, i.e. K3, a Calabi-Yau threefold and fourfold by integrating the above appropriate Chern classes in (A.11) over \hat{X} . Using that $S_m \cdot D_\alpha \cdot D_\beta \cdot D_\gamma = (D_\alpha \cdot D_\beta \cdot D_\gamma)|_B$ for a fourfold \hat{X} , respectively, analogous relations for the two- and threefold case, we obtain that all integrals reduce to integrals over the base B. By some algebra we immediately reproduces the results anticipated in (A.1).

As another application of the explicit formulas (A.9) for the fourfold \hat{X} we calculate the Todd class $\operatorname{Td}_4(\hat{X})$ of \hat{X} . Upon integrating the Todd class over \hat{X} we obtain the arithmetic genus $\chi_0(\hat{X})$ by means of an Hirzebruch-Riemann-Roch index theorem as

$$\chi_0(\hat{X}) := \sum_p (-1)^p h^{(p,0)}(\hat{X}) = \int_{\hat{X}} \mathrm{Td}_4(\hat{X}) \,. \tag{A.12}$$

In the case of a Calabi-Yau fourfold, see cf. [57] for more details, the Todd class is given as

$$\mathrm{Td}_4(\hat{X}) = \frac{1}{720} \left(3c_2(\hat{X})^2 - c_4(\hat{X}) \right)$$
(A.13)

in terms of the Chern classes $c_2(\hat{X})$ and $c_4(\hat{X})$. Thus, the arithmetic genus reads

$$\chi_0(\hat{X}) = \int_{\hat{X}} \operatorname{Td}_4(\hat{X}) = \frac{1}{12} \int_B c_1 c_2 = 2\chi_0(B) \,. \tag{A.14}$$

Here we have evaluated the second equality equality using the expression (A.9) for the fourth Chern class of \hat{X} and computed the square of the second Chern class $c_2(\hat{X})^2$ as

$$c_{2}(\hat{X})^{2} = S_{P} \left[4 \left(2c_{1} - \mathcal{S}_{7} \right) \left(c_{1}^{2} + c_{2} - c_{1}\mathcal{S}_{7} \right) + \left(6c_{2} + \left(2c_{1} - \mathcal{S}_{7} \right) \left(5c_{1} + \mathcal{S}_{7} \right) \right) \mathcal{S}_{9} + \left(\mathcal{S}_{7} - 3c_{1} \right) \mathcal{S}_{9}^{2} + \mathcal{S}_{9}^{3} \right] + 2S_{Q} \left[4c_{1}^{3} + 2c_{2} \left(\mathcal{S}_{7} - \mathcal{S}_{9} \right) - \mathcal{S}_{7}S_{9}^{2} - 2c_{1}^{2} \left(\mathcal{S}_{7} + 3\mathcal{S}_{9} \right) + 2c_{1} \left(2c_{2} + \mathcal{S}_{7}^{2} + \mathcal{S}_{7}\mathcal{S}_{9} + \mathcal{S}_{9}^{2} \right) \right] + S_{R} \left[\left(4c_{1} - \mathcal{S}_{9} \right) \left(2 \left(c_{1}^{2} + c_{2} \right) - 3c_{1}\mathcal{S}_{9} + \mathcal{S}_{9}^{2} \right) \right],$$
(A.15)

after some algebra in the cohomology ring (3.18). In the last equality in (A.14) we employed the relation

$$\chi_0(B) = \frac{1}{24} \int_B c_1 c_2 \tag{A.16}$$

for the arithmetic genus of the base B. We observe a surprising cancellation of all terms involving S_7 , S_9 in the fourth Chern class in (A.9) and in $c_2(\hat{X})^2$ in (A.15) so that the final result in (A.14) only depends on the Chern classes of the base B.

We conclude that for a suitable base B of a simply-connected Calabi-Yau fourfold with $\chi_0(\hat{X}) = 2$, which follows from $h^{(0,0)}(\hat{X}) = h^{(4,0)}(\hat{X}) = 1$ and $h^{(p,0)}(\hat{X}) = 0$ otherwise, we have to require $\chi_0(B) = 1$ or, using (A.16),

$$\int_{B} c_1 c_2 = 24.$$
 (A.17)

We even require the stronger conditions $h^{(1,0)}(B) = h^{(2,0)}(B) = h^{(3,0)}(B) = 0$ for B, since every non-trivial element in these Hodge cohomology groups would give rise via the pullback under $\pi : \hat{X} \to B$ to a corresponding element in the Hodge cohomology of \hat{X} . However, this is excluded by assumption of a simply-connected fourfold.

B Intersection ring on \hat{X} with $B = \mathbb{P}^3$

In this section we work out the intersection ring $H_V^{(*,*)}(\hat{X})$ of the fourfold \hat{X} over the base $B = \mathbb{P}^3$. To this end, we relate the intersections on \hat{X} to intersections on the ambient space $dP_2(n_7, n_9)$. On the ambient space $dP_2(n_7, n_9)$ the intersections are determined completely by the intersections of the fiber dP_2 , the base \mathbb{P}^3 and basic properties of the fibration. This will allow us, as demonstrated below and in the main text, to work out the basis of $H_V^{(k,k)}(\hat{X})$ for all $k \leq 4$ on the one hand, and to calculate the quartic intersections on \hat{X} .

We recall from the main text, cf. (3.31) and (3.32), that the basis D_A of divisors on π : $\hat{X} \to \mathbb{P}^3$ reads

$$H^{(1,1)}(\hat{X}) = \langle H_B, S_P, S_Q, S_R \rangle, \qquad (B.1)$$

where S_P , S_Q , S_R denote the homology classes of the sections \hat{s}_P , \hat{s}_Q and \hat{s}_R , respectively.

Intersection relations and the quartic intersections

Next we begin by calculating the intersections of two divisors D_A , where we denote the intersection pairing on \hat{X} by a subscript. We employ the representation (3.18) for $H^{(*,*)}(\hat{X})$ to relate these to intersections on the ambient space $dP_2(n_7, n_9)$, that we denote as '.'. First we note that there are $\frac{5!}{2!3!} = 10$ different quadratic combinations $D_A \cdot_{\hat{X}} D_B$. We evaluate these as

$$S_P^2 = S_P(S_P[K_B^{-1}] - (\mathcal{S}_7 - \mathcal{S}_9)(\mathcal{S}_9 - [K_B^{-1}])), \quad S_PS_Q = 0,$$

$$S_PS_R = -S_P\mathcal{S}_9(S_P - \mathcal{S}_9 + [K_B^{-1}]), \quad S_Q^2 = S_Q(S_Q[K_B^{-1}] + (\mathcal{S}_7 - [K_B^{-1}])(\mathcal{S}_7 - \mathcal{S}_9)),$$

$$S_R^2 = S_R((\mathcal{S}_7 - [K_B^{-1}]) (\mathcal{S}_9 - [K_B^{-1}]) + S_R[K_B^{-1}]), \quad S_QS_R = -S_Q\mathcal{S}_7(S_Q - \mathcal{S}_7 + [K_B^{-1}]),$$

$$S_PH_B = -S_PH_B(S_P + \mathcal{S}_7 - 2\mathcal{S}_9), \quad S_QH_B = -S_QH_B(S_Q + \mathcal{S}_9 - 2\mathcal{S}_7),$$

$$S_RH_B = S_RH_B (\mathcal{S}_7 + \mathcal{S}_9 - [K_B^{-1}] - S_R),$$

$$H_B^2 = H_B^2 (3[K_B^{-1}] - \mathcal{S}_7 - \mathcal{S}_9 + 2S_P + 2S_Q + 3S_R),$$

(B.2)

where we used $S_7 = n_7 H_B$, $S_9 = n_9 H_B$ and $[K_B^{-1}] = 4H_B$ for $B = \mathbb{P}^3$. Here we suppressed the intersection product '.' for brevity of our notation. The intersections on the left side of the equations are carried out in the Calabi-Yau fourfold \hat{X} and related to the intersections on the ambient space $dP_2(n_7, n_9)$ on the right side of the equations.

Next, we calculate the triple intersections of divisors on \hat{X} . There are $\frac{6!}{3!3!} = 20$ different cubic combinations $D_A \cdot_{\hat{X}} D_B \cdot_{\hat{X}} D_C$ with intersections

$$\begin{split} H_B^3 &= H_B^3 \left(2S_P + 2S_Q + 3S_R \right), & H_B^2 S_P = S_P H_B^2 \left(2S_9 - S_7 - S_P \right), \\ H_B^2 S_Q &= S_Q H_B^2 \left(2S_7 - S_9 - S_Q \right), & H_B^2 S_R = S_R H_B^2 \left(S_7 + S_9 - [K_B^{-1}] - S_R \right), \\ H_B S_P S_R &= -S_P H_B S_9 \left(S_P + [K_B^{-1}] - S_9 \right), & H_B S_Q S_R = -S_Q H_B S_7 \left(S_Q - S_7 + [K_B^{-1}] \right), \\ H_B S_P^2 &= S_P H_B \left(S_P [K_B^{-1}] + \left(S_9 - S_7 \right) \left(S_9 - [K_B^{-1}] \right) \right), \\ H_B S_Q^2 &= S_Q H_B \left(S_Q [K_B^{-1}] + \left(S_7 - [K_B^{-1}] \right) \left(S_7 - S_9 \right) \right), \\ H_B S_R^2 &= S_R H_B \left(\left(S_7 - [K_B^{-1}] \right) \left(S_9 - [K_B^{-1}] \right) + S_R [K_B^{-1}] \right), \\ S_P^3 &= -S_P \left([K_B^{-1}] \left(S_9 - cS_7 \right) \left(S_9 - [K_B^{-1}] \right) + S_P \left([K_B^{-1}]^2 + \left(S_7 - [K_B^{-1}] \right) \right) \left(S_7 - S_9 \right) \right), \\ S_R^3 &= -S_R \left([K_B^{-1}] \left(S_7 - [K_B^{-1}] \right) \left(S_9 - [K_B^{-1}] \right) \right) - [K_B^{-1}] \left(S_7 - [K_B^{-1}] \right) \left(S_7 - S_9 \right), \\ S_R^2 S_R &= S_P \left(S_P - S_9 + [K_B^{-1}] \right) \left(S_9 - [K_B^{-1}] \right) + \left([K_B^{-1}]^2 - S_7 S_9 \right) S_R \right), \\ S_Q^2 S_R^2 &= -S_Q \left(S_Q - S_7 + [K_B^{-1}] \right) \left(S_7 - S_9 \right), \\ S_Q S_R^2 &= -S_Q \left(S_Q - S_7 + [K_B^{-1}] \right) \left(S_7 - S_9 \right), \\ S_Q S_R^2 &= -S_Q \left(S_Q - S_7 + [K_B^{-1}] \right) \right) \left(S_7 - S_9 \right), \\ S_Q S_R^2 &= -S_Q \left(S_Q - S_7 + [K_B^{-1}] \right) \left(S_7 - S_9 \right), \\ S_Q S_R^2 &= -S_Q \left(S_Q - S_7 + [K_B^{-1}] \right) \left(S_7 - S_9 \right), \\ S_Q S_R^2 &= -S_Q \left(S_Q - S_7 + [K_B^{-1}] \right) \right) \left(S_7 - S_9 \right), \\ S_Q S_R^2 &= -S_Q \left(S_Q - S_7 + [K_B^{-1}] \right) \left(S_7 - S_9 \right), \\ S_Q S_R^2 &= -S_Q \left(S_Q - S_7 + [K_B^{-1}] \right) \left(S_7 - S_9 \right), \\ S_Q S_R^2 &= -S_Q \left(S_Q - S_7 + [K_B^{-1}] \right) \left(S_7 - S_9 \right), \\ S_Q S_R^2 &= -S_Q \left(S_Q - S_7 + [K_B^{-1}] \right) \left(S_7 - S_9 \right), \\ S_Q S_R^2 &= -S_Q \left(S_Q - S_7 + [K_B^{-1}] \right) \left(S_7 - S_9 \right), \\ S_Q S_R^2 &= -S_Q \left(S_Q - S_7 + [K_B^{-1}] \right) \left(S_7 - S_9 \right), \\ S_Q S_R^2 &= -S_Q \left(S_Q - S_7 + [K_B^{-1}] \right) \left(S_7 - S_9 \right), \\ S_Q S_R^2 &= -S_Q \left(S_Q - S_7 + [K_B^{-1}] \right) \left(S_7 - S_9 \right), \\ S_Q S_R^2 &= -S_Q \left(S_Q - S_7 + [K_B^{-1}] \right) \left(S_7 - S_9 \right), \\ S_Q S_R^2 &= -S_Q \left(S_Q - S_7 + [K_B^{-1}] \right) \left(S_7 - S_9 \right), \\ S_Q S_R^2 &=$$

We note that all other intersections vanish, since $S_P \cdot S_Q = 0$ and as before we suppressed the intersection product "." of \hat{X} , respectively, $dP_2(n_7, n_9)$ on the left, respectively, right side of the equations.

Finally we calculate the quartic intersections as

$$\begin{aligned} \mathcal{C}_{0} &= H_{B}^{3}S_{P} - 4H_{B}^{2}S_{P}^{2} + \left(16 + (n_{7} - 4)n_{9} - n_{9}^{2}\right)H_{B}S_{P}^{3} + H_{B}^{3}S_{Q} - 4H_{B}^{2}S_{Q}^{2} \\ &+ \left(\left(32 - 4n_{7} - n_{7}^{2}\right)n_{9} + 3n_{7}n_{9}^{2} - 2n_{9}^{3} - 64\right)S_{P}^{4} + \left(16 - n_{7}^{2} + n_{7}\left(n_{9} - 4\right)\right)H_{B}S_{Q}^{3} \\ &- \left(64 + 2n_{7}^{3} - 3n_{7}^{2}n_{9} - n_{7}\left(32 - 4n_{9} - n_{9}^{2}\right)\right)S_{Q}^{4} + H_{B}^{3}S_{R} + n_{9}H_{B}^{2}S_{P}S_{R} \\ &+ n_{9}\left(n_{9} - n_{7}\right)H_{B}S_{P}^{2}S_{R} + \left(n_{7} - n_{9}\right)^{2}n_{9}S_{P}^{3}S_{R} + n_{7}H_{B}^{2}S_{Q}S_{R} + n_{7}\left(n_{7} - n_{9}\right)H_{B}S_{Q}^{2}S_{R} \\ &+ n_{7}\left(n_{7} - n_{9}\right)^{2}S_{Q}^{3}S_{R} - 4H_{B}^{2}S_{R}^{2} + \left(n_{7} - 4\right)n_{9}H_{B}S_{P}S_{R}^{2} + \left(4 - n_{7}\right)\left(n_{7} - n_{9}\right)n_{9}S_{P}^{2}S_{R}^{2} \\ &+ n_{7}\left(n_{9} - 4\right)H_{B}S_{Q}S_{R}^{2} + n_{7}\left(n_{7} - n_{9}\right)\left(n_{9} - 4\right)S_{Q}^{2}S_{R}^{2} + \left(16 - n_{7}n_{9}\right)H_{B}S_{R}^{3} \\ &+ \left(n_{7} - 4\right)^{2}n_{9}S_{P}S_{R}^{3} + n_{7}\left(n_{9} - 4\right)^{2}S_{Q}S_{R}^{3} - \left(64 + n_{7}^{2}n_{9} + n_{7}\left(n_{9} - 12\right)n_{9}\right)S_{R}^{4}, \end{aligned} \tag{B.4}$$

where the coefficient of the polynomial C_0 of $D_A \cdot_{\hat{X}} D_B \cdot_{\hat{X}} D_C \cdot_{\hat{X}} D_D$ is the corresponding intersection number in \hat{X} . In this computation we have used that the intersections on dP_2 in (2.4) are embedded into the intersections of $dP_2(n_7, n_9)$ as the intersections containing $H_B^3 \cdot D_A \cdot D_B$, in particular $H_B^3 \cdot S_R^2 = H_B^3 \cdot S_P^2 = H_B^3 \cdot S_Q^2 = -1$.

Since X is complex four-dimensional, intersections of more than four divisors on X vanish, which is confirmed by the concrete presentation (3.18) of the intersection ring with $B = \mathbb{P}^3$.

The cohomology basis of $H_V^{(*,*)}(\hat{X})$

The basis of $H_V^{(k,k)}(\hat{X})$ for each k = 0, ..., 4 is now determined as follows. First, we note that there is a canonical basis both for k = 0, 1. Namely, the one-dimensional space $H^{(0,0)}(\hat{X})$ is spanned by the generated by 1, $H^{(0,0)}(\hat{X}) = \langle 1 \rangle$, and four-dimensional $H^{(1,1)}(\hat{X})$ is generated by the D_A as indicated in (B.1). Using the quartic intersections (B.4) we can now calculate a basis of $H_V^{(2,2)}(\hat{X})$. For generic n_7 , n_9 this space is five-dimensional and a basis is given by

$$H_V^{(2,2)}(\hat{X}) = \langle H_B^2, H_B \cdot S_P, H_B \cdot \sigma(\hat{s}_Q), H_B \cdot \sigma(\hat{s}_R), S_P^2 \rangle.$$
(B.5)

Here $\sigma(\hat{s}_Q)$, $\sigma(\hat{s}_R)$ denotes the Shioda map (2.17) of the rational sections, that takes in the case at hand the form

$$\sigma(\hat{s}_Q) = S_Q - S_P - 4H_B, \quad \sigma(\hat{s}_R) = S_R - S_P - (4+n_9)H_B.$$
(B.6)

The intersection matrix $\eta^{(2)}$ in this basis is readily calculated from (B.4) as

$$\eta^{(2)} = \begin{pmatrix} 0 & 1 & 0 & 0 & -4 \\ 1 & -4 & 0 & 0 & 16 + (-4 + n_7 - n_9) n_9 \\ 0 & 0 & -8 & n_7 - n_9 - 4 & n_9 (4 - n_7 + n_9) \\ 0 & 0 & \eta_{34}^{(2)} & -2 (4 + n_9) & 2n_9 (4 - n_7 + n_9) \\ -4 & \eta_{25}^{(2)} & \eta_{35}^{(2)} & \eta_{45}^{(2)} & -64 - (8 + n_7 - 2n_9) (-4 + n_7 - n_9) n_9 \end{pmatrix}, \quad (B.7)$$

where lengthy entries $\eta_{rs}^{(2)}$ that are determined by symmetry of $\eta^{(2)}$ are omitted and denoted by $\eta_{sr}^{(2)}$.

From this matrix it is apparent that if (n_7, n_9) are on the boundary of the allowed region in figure 2, the rank is reduced to four. In this case, the five surfaces in (B.5) become homologous. A choice of basis in this case is given by the first four surfaces in (B.5) with the surface S_P^2 dropped. Consequently, the intersection matrix $\eta^{(2)}$ in this case is the 4 × 4-submatrix of (B.7) obtained by deleting the fifth row and column. Let us discus these non-generic cases in more detail, to illustrate the decrease from five to four basis elements.

 $n_7 = 0$: we see from (2.10) that $s_7 \neq 0$, which implies that \hat{s}_Q and \hat{s}_R are holomorphic, cf. the discussion below (2.12). we obtain the cohomology relation

$$S_P^2 \cong -n_9(4+n_9)H_B^2 - 4H_B \cdot S_P - n_9H_B \cdot \sigma(\hat{s}_R)$$
(B.8)

in $H_V^{(2,2)}(\hat{X})$ which follows from the relations

$$H_B S_P = H_B(S_P([K_B^{-1}] + \mathcal{S}_9) - S_R(S_R + [K_B^{-1}] - \mathcal{S}_9)),$$

$$H_B S_R = H_B S_R(\mathcal{S}_9 - [K_B^{-1}] - S_R)$$

$$S_P^2 = \left([K_B^{-1}] + \mathcal{S}_9\right) \left(\mathcal{S}_9 - [K_B^{-1}]\right) S_P + [K_B^{-1}] \left([K_B^{-1}] - \mathcal{S}_9 + S_R\right) S_R.$$
 (B.9)

These relations can be proven for $n_7 = 0$ using the presentation (3.18).

 $n_9 = 4 + n_7$: in this situation, the section \hat{s}_Q is holomorphic. Analogously, as before, we compute the homology relation on \hat{X} reading

$$S_P^2 \cong -8(4+n_7)H_B^2 - 4H_B \cdot S_P - n_7H_B \cdot \sigma(\hat{s}_Q) - 4H_B \cdot \sigma(\hat{s}_R).$$
(B.10)

 $n_9 = 12 - n_7$: in this case all sections are rational, but $s_1 \neq 0$ in (2.10). As before, we compute the homology relation

$$S_P^2 \cong -2(n_7 - 12)(n_7 - 8)H_B^2 - 4H_B \cdot S_P + (n_7 - 8)H_B \cdot \sigma(\hat{s}_Q) + (n_7 - 8)H_B \cdot \sigma(s_R) \,.$$
(B.11)

 $n_9 = 0$: for this value the sections \hat{s}_P and \hat{s}_R are holomorphic. Similarly, we compute for $n_9 = 0$ the following relation in homology,

$$S_P^2 + 4H_B \cdot S_P = 0. (B.12)$$

This is immediately clear since for a holomorphic zero section the relation (2.16) holds in the homology of \hat{X} , which is precisely the relation (B.12). It is satisfying that this relation is reproduced by the intersection calculation (3.18) by noting the relations

$$S_P^2 = [K_B^{-1}] \left(\left(S_7 - [K_B^{-1}] \right) S_P + S_R \left([K_B^{-1}] + S_R \right) \right) ,$$

$$S_P H_B = -H_B \left(\left(S_7 - [K_B^{-1}] \right) S_P + S_R \left([K_B^{-1}] + S_R \right) \right) ,$$
(B.13)

that are calculated for $n_9 = 0$. Combining these two relations precisely yields (B.12).

 $n_9 = n_7 - 4$: for this value the sections \hat{s}_P , but this time since $s_8 \neq 0$ as is clear from (2.10). Thus, the relation (B.13) still holds.

Next, we note that the basis of $H^{(3,3)}(\hat{X})$ is determined by Poincaré duality from $H^{(1,1)}(\hat{X})$. Thus, the group $H^{(3,3)}(\hat{X})$ is generated by four elements, independently of n_7 and n_9 , which can be checked by an explicit calculation. However, we will refrain from presenting the details, since the group $H^{(3,3)}(\hat{X})$ is of no immediate relevance for F-theory, and leave this as an exercise for the interested reader. Finally, $H^{(4,4)}(\hat{X})$ is generated by a single element which is the volume form normalized by the volume of \hat{X} . It is given, up to combinatorical factors, by \mathcal{C}_0 in (B.4).

C Total Chern class of $\widehat{dP}_2^B(\mathcal{S}_7, \mathcal{S}_9)$

In this brief appendix we outline the calculation of the total Chern class of the resolved space $\widehat{dP}_2^B(\mathcal{S}_7, \mathcal{S}_9)$, which is straightforward but tedious. We keep our discussion short, leaving some details of the calculations for the interested reader.

The basic strategy to obtain the Chern class of $\widehat{dP}_2^B(\mathcal{S}_7, \mathcal{S}_9)$ is to successively modify (A.2) taking into account all changes of divisor classes in the blow-up (6.8). First we have to use the new divisor classes (6.9) for homogeneous coordinates $[\tilde{u} : \tilde{v} : \tilde{w} : e_1 : e_2]$. This alters the second to fourth factors in (A.2) accordingly. Then we have to include the classes of the Cartan divisor, which is achieved by multiplying in (A.2) by $\prod_i (1 + D_i)$. Finally we have to modify the first factor in (A.2) to incorporate the split and shift of the class of z by the D_i as in (6.9) appropriately. All we have to do is to formally replace the divisor $\mathcal{S}_{SU(5)}$ in the expression for the Chern class of B as

$$S_{\mathrm{SU}(5)} \to S_{\mathrm{SU}(5)} - \sum_{i} D_{i},$$
 (C.1)

expand and then re-express the result we get in terms of the Chern class of B and the intersections of $S_{SU(5)}$. Taking all these changes into account we obtain the following expression

$$c(\widehat{dP}_{2}^{B}) = \left[1 + c_{1} + c_{2} - D_{1}^{2} - D_{1}D_{2} - D_{2}^{2} - D_{2}D_{3} - D_{3}^{2} - D_{2}D_{4} - D_{3}D_{4} - D_{4}^{2} + \left(\mathcal{S}_{SU(5)} - D_{0}\right)\mathcal{S}_{SU(5)} + c_{3} - c_{1}\left(D_{1}^{2} + D_{1}D_{2} + D_{2}^{2} + D_{3}^{2} + D_{3}D_{4} + D_{4}^{2} + D_{2}D_{3} + D_{2}D_{4}\right) \\ + \mathcal{S}_{SU(5)}\left(D_{1}^{2} + 2D_{1}D_{2} + (D_{2} + D_{3} + D_{4})^{2}\right) + \left(\mathcal{S}_{SU(5)} - D_{0}\right)\mathcal{S}_{SU(5)}\left(c_{1} - \mathcal{S}_{SU(5)}\right) \\ - D_{1}^{2}D_{2} - D_{1}D_{2}^{2} - (D_{2} + D_{3})(D_{2} + D_{4})(D_{3} + D_{4})\right] \\ \times (1 + [u])(1 + [v])(1 + [w])(1 + E_{1})(1 + E_{2}), \qquad (C.2)$$

where we used the relation $S_{SU(5)} - D_0 = D_1 + D_2 + D_3 + D_4$ between the extended node D_0 and $S_{SU(5)}$ as well as abbreviated the ambient space as \widehat{dP}_2^B . We also denote the classes (6.9) of the homogeneous coordinates by brackets [·]. We emphasize that the new factors $(1 + D_i)$ from the Cartan divisors D_i have been combined with the changes in the first factor in (A.2) containing the Chern class of B.

D Intersection ring for $\hat{X}_{SU(5)}$ with $B = \mathbb{P}^3$

In this section we present the core cohomology calculations of elliptically fibered Calabi-Yau fourfolds $\pi : \hat{X} \to \mathbb{P}^3$ over \mathbb{P}^3 with general elliptic fiber \mathcal{E} in dP_2 and an additional resolved SU(5)-singularity over codimension one in \mathbb{P}^3 . The geometry has been introduced in section 6, to which we refer for more details. The following discussion is very similar to the one in A and B. Thus, we will keep the exposition as short as possible and only present the key results necessary for the construction of G_4 -flux and 3D CS-terms in the main text.

The starting point of the calculation of the cohomology ring $H_V^{(*,*)}(\hat{X})$ of \hat{X} is the representation (6.41) as a quotient ring. We are working with a particular phase of the fourfold \hat{X} with Stanley-Reissner ideal (6.12). The eight-dimensional basis of $H^{(1,1)}(\hat{X})$ is given in (6.46) and the class of the anti-canonical bundle $K_{dP_2}^{-1}$ in (6.11). Omitting lengthy details, that are best performed using a computer algebra program, we immediately obtain the quartic intersections on \hat{X} as

$$\begin{split} \mathcal{C}_{0} &= 12D_{1}^{3}H_{B} + 4D_{2}D_{3}D_{4}H_{B} - 8D_{2}D_{4}^{2}H_{B} + 12D_{4}^{3}H_{B} - 2D_{1}^{2}H_{B}^{2} + D_{1}D_{2}H_{B}^{2} - 2D_{2}^{2}H_{B}^{2} + D_{2}D_{3}H_{B}^{2} \\ &- 2D_{3}^{2}H_{B}^{2} + D_{3}D_{4}H_{B}^{2} - 2D_{4}^{2}H_{B}^{2} + (n_{7}-7)D_{2}D_{3}^{2}H_{B} - 4(n_{7}-4)D_{2}^{2}D_{3}D_{4} + 8(n_{7}-4)D_{2}^{2}D_{4}^{2} \\ &- (n_{7}-4)D_{2}^{2}D_{3}H_{B} - (n_{7}-7)(n_{7}-4)D_{2}^{2}D_{3}^{2} + (n_{7}-4)^{2}D_{2}^{3}D_{3} - 16(n_{7}-3)D_{2}D_{4}^{3} - (n_{9}-7)D_{1}D_{2}^{2}H_{B} \\ &- ((n_{7}-5)n_{7} + (n_{9}-6)(n_{9}-5))D_{2}^{4} + (4 + (n_{7}-n_{9})^{2} - 4n_{9})D_{3}D_{4}^{3} + 4(n_{7}-n_{9}-1)D_{2}D_{3}^{2}D_{4} \\ &+ (n_{7}-n_{9}-1)D_{3}^{2}D_{4}H_{B} - (n_{7}-n_{9}-1)(2 + n_{7}-n_{9})D_{3}^{2}D_{4}^{2} + (n_{9}-10)D_{1}^{2}D_{2}H_{B} + 4(n_{9}-2)D_{2}D_{3}D_{4}^{2} \\ &+ (6 - 2n_{7}+n_{9})D_{3}^{3}H_{B} + (n_{9}-n_{7}-2)D_{3}D_{4}^{2}H_{B} + (1 - n_{7}+n_{9})^{2}D_{3}^{3}D_{4} + (n_{7}+n_{9}-9)D_{2}^{3}H_{B} \\ &+ (25 + (n_{7}-14)n_{7}+4n_{9})D_{2}D_{3}^{3} + (16n_{7}-12 + (n_{9}-20)n_{9})D_{1}^{3}D_{2} - (14 + 8n_{7}+(n_{9}-17)n_{9})D_{1}^{2}D_{2}^{2} \\ &+ (21 + 4n_{7} + (n_{9}-14)n_{9})D_{1}D_{2}^{3} - 2(16n_{7}-36 + (n_{9}-14)n_{9})D_{1}^{4} - 2(28 + n_{7}^{2}+n_{9}(2 + n_{9}) - 2n_{7}(7 + n_{9}))D_{4}^{4} \\ &- (20 + 2n_{7}^{2}+n_{9}(5 + n_{9}) - 2n_{7}(7 + n_{9}))D_{3}^{4} + H_{B}^{3}S_{P} + (16 + (n_{7}-n_{9}-4)n_{9})H_{B}S_{P}^{3} \\ &- 4H_{B}^{2}S_{P}^{2} + (-64 - (8 + n_{7}-2n_{9})(n_{7}-n_{9}-4)n_{9})S_{P}^{4} + D_{3}^{3}S_{Q} + D_{3}^{2}H_{B}S_{Q} + H_{B}^{3}S_{Q} \\ &- 4D_{3}^{2}S_{Q}^{2} - 4D_{3}H_{B}S_{Q}^{2} - 4H_{B}^{2}S_{Q}^{2} + (16 + n_{7}(n_{9}-n_{7}-2))D_{3}S_{Q}^{3} + (16 + n_{7}(n_{9}-n_{7}-2))H_{B}S_{Q}^{3} + H_{B}^{3}S_{R} \end{split}$$

$$-(64+n_{7}(2+n_{7}-n_{9})(2n_{7}-n_{9}-10))S_{Q}^{4}-4n_{7}D_{4}^{3}S_{R}-n_{7}D_{3}^{2}H_{B}S_{R}+D_{3}D_{4}H_{B}n_{7}S_{R}-2n_{7}D_{4}^{2}H_{B}S_{R}$$

$$+n_{7}(n_{7}-n_{9}-1)D_{3}^{2}D_{4}S_{R}+n_{7}(n_{9}-n_{7})D_{3}^{3}S_{R}+n_{7}(2-n_{7}+n_{9})D_{3}D_{4}^{2}S_{R}+n_{9}H_{B}^{2}S_{P}S_{R}+n_{7}D_{3}H_{B}S_{Q}S_{R}$$

$$+n_{9}(n_{9}-n_{7})H_{B}S_{P}^{2}S_{R}+(n_{7}-n_{9})^{2}n_{9}S_{P}^{3}S_{R}+n_{7}D_{3}^{2}S_{Q}S_{R}+n_{7}H_{B}^{2}S_{Q}S_{R}+n_{7}(n_{7}-n_{9}-2)D_{3}S_{Q}^{2}S_{R}$$

$$+n_{7}(n_{7}-n_{9}-2)H_{B}S_{Q}^{2}S_{R}+n_{7}(2-n_{7}+n_{9})^{2}S_{Q}^{3}S_{R}-4H_{B}^{2}S_{R}^{2}-D_{3}H_{B}n_{7}S_{R}^{2}+(n_{7}-4)n_{7}D_{3}D_{4}S_{R}^{2}$$

$$-2(n_{7}-4)n_{7}D_{4}^{2}S_{R}^{2}-(n_{7}-3)n_{7}D_{3}^{2}S_{R}^{2}+H_{B}(n_{7}-4)n_{9}S_{P}S_{R}^{2}-(n_{7}-4)(n_{7}-n_{9})n_{9}S_{P}^{2}S_{R}^{2}$$

$$+n_{7}(n_{9}-2)D_{3}S_{Q}S_{R}^{2}+n_{7}(n_{9}-2)H_{B}S_{Q}S_{R}^{2}+(n_{7}-4)^{2}n_{9}S_{P}S_{R}^{3}+n_{7}(n_{7}-n_{9}-2)(n_{9}-2)S_{Q}^{2}S_{R}^{2}$$

$$-n_{7}(n_{7}+n_{9}-6)D_{3}S_{R}^{3}+(16-n_{7}(2+n_{9}))H_{B}S_{R}^{3}+n_{7}(n_{9}-2)^{2}S_{Q}S_{R}^{3}-(64+n_{7}(2+n_{9})(n_{7}+n_{9}-10))S_{R}^{4}.$$
(D.1)

As before, all intersections are understood to be evaluated on \hat{X} and the quartic intersections $D_A \cdot_{\hat{X}} D_B \cdot_{\hat{X}} D_C \cdot_{\hat{X}} D_D$ are read off as the coefficient of the appropriate monomial in \mathcal{C}_0 .

As one immediate application we calculate the Cartan matrix C_{IJ} of the affine Lie algebra of SU(5) as

$$C_{IJ} = -D_I \cdot D_J \cdot H_B^2 = \begin{pmatrix} 2 & -1 & 0 & 0 & -1 \\ -1 & 2 & -1 & 0 & 0 \\ 0 & -1 & 2 & -1 & 0 \\ 0 & 0 & -1 & 2 & -1 \\ -1 & 0 & 0 & -1 & 2 \end{pmatrix},$$
(D.2)

where we supplemented as before the Cartan divisors D_i , i = 1, ..., 4, by the divisor $D_0 = H_B - D_1 - D_2 - D_3 - D_4$ corresponding to the extended node $-\alpha_0$ as $D_I = (D_0, D_i)$. In addition, we confirm the intersections (2.13), (2.14) and (2.15) explicitly employing (D.1). Finally, we calculate the intersections of the sections with the nodes of the Dynkin diagram of SU(5) as

$$S_P \cdot D_I \cdot H_B^2 = S_R \cdot D_I \cdot H_B^2 = (1, 0, 0, 0, 0)_I, \quad S_Q \cdot D_I \cdot H_B^2 = (0, 0, 0, 1, 0)_I$$
(D.3)

reproducing the independent findings in (6.16) and, consequently, the Shioda maps (6.17). Here we exploited (6.15) to represent the curve $c_{-\alpha_I}$ corresponding to the simple root α_I as

$$c_{-\alpha_I} = D_I \cdot H_B^2 \,, \tag{D.4}$$

employing $H_B^2 \cdot S_{SU(5)}^b = H_B^3 = 1$. We note that we can even make the stronger statement

$$S_P \cdot D_i = 0 \tag{D.5}$$

in the homology of $\hat{X}_{SU(5)}$, which will be important for the discussion of section 7.2.

The cohomology group $H_V^{(2,2)}(\hat{X}_{SU(5)})$ is readily calculated. We generically obtain a 13-dimensional vector space, with certain jumps in the cohomology for non-generic values of n_7 and n_9 , see figure 5. For the application of the construction of G_4 -flux, it proves useful to make a particular choice of basis by hand. We present a thorough analysis of this basis in the next appendix.

E Basis of $H_V^{(2,2)}(\hat{X}_{\mathrm{SU}(5)})$ with $B = \mathbb{P}^3$

The vector space $H_V^{(2,2)}$ is 13-dimensional for arbitrary values of n_7 and n_9 . A basis is (6.47), that we recall here for convenience

$$H_{(V,13)}^{(2,2)} = \langle H_B^2, H_B \cdot S_P, H_B \cdot \sigma(S_Q), H_B \cdot \sigma(S_R), S_P \cdot \sigma(S_R), \sigma(S_Q) \cdot \sigma(S_R), \\ \sigma(S_Q)^2, H_B \cdot D_1, H_B \cdot D_2, H_B \cdot D_3, H_B \cdot D_4, D_2 \cdot D_4, D_1^2 \rangle.$$
(E.1)

To obtain the smaller vector spaces of the boundaries we need to drop some elements of this basis. The vectors that become linearly depended at the boundary have to be re-expressed as a linear combination of the smaller basis on the boundary. In the next subsections, we choose bases for the different boundaries. We write the linearly dependent elements, that we can drop from the above 13-dimensional basis, as linear combinations of the rest of the vectors in the smaller basis.

12 dimensional

At the boundaries with a twelve-dimensional basis we can take

$$H_{(V,12)}^{(2,2)} = H_{(V,13)}^{(2,2)} \setminus \{S_P \cdot \sigma(S_R)\},$$
(E.2)

in which case the surface $S_P \cdot \sigma(S_R)$ is re-expressed as:

• At $n_9 = 0$

$$S_P \cdot \sigma(S_R) \cong 0, \tag{E.3}$$

• At the boundary $n_9 = n_7 + 2$

$$S_{P} \cdot \sigma(S_{R}) \cong -\frac{2}{35} \left(30H_{B}^{2}(26+3n_{7}) + 25\sigma(\hat{s}_{Q}) \cdot \sigma(\hat{s}_{R}) + H_{B} \cdot \left[4D_{1}n_{7} + 8D_{2}n_{7} + 12D_{3}n_{7} + 6D_{4}n_{7} + 150S_{P} + 150\sigma(\hat{s}_{Q}) + 15n_{7}\sigma(\hat{s}_{Q}) + 130\sigma(\hat{s}_{R}) \right] \right), \quad (E.4)$$

• At the boundary $n_9 = 9 - n_7$

$$S_{P} \cdot \sigma(S_{R}) \cong \frac{2}{475} \left(18D_{1} \cdot H_{B} + 36D_{2} \cdot H_{B} - 66D_{3} \cdot H_{B} - 33D_{4} \cdot H_{B} - 41100H_{B}^{2} + 8400n_{7}H_{B}^{2} - 400n_{7}^{2}H_{B}^{2} - 4425H_{B} \cdot S_{P} + 550n_{7}H_{B} \cdot S_{P} - 5145H_{B} \cdot \sigma(\hat{s}_{Q}) + 450n_{7}H_{B} \cdot \sigma(\hat{s}_{Q}) - 125\sigma(\hat{s}_{Q})^{2} - 2790H_{B} \cdot \sigma(\hat{s}_{R}) + 350n_{7}H_{B} \cdot \sigma(\hat{s}_{R}) - 275\sigma(\hat{s}_{Q}) \cdot \sigma(\hat{s}_{R}) \right).$$
(E.5)

11 dimensional

We obtain an eleven-dimensional basis by choosing

$$H_{(V,12)}^{(2,2)} = H_{(V,13)}^{(2,2)} \setminus \{S_P \cdot \sigma(S_R), H_B \cdot \sigma(S_Q)\},$$
(E.6)

• At $n_7 = n_9 - 4$ we obtain the homology relation

$$S_{P} \cdot \sigma(S_{R}) \cong 0,$$

$$H_{B} \cdot \sigma(S_{Q}) \cong \frac{1}{1220 + 150n_{7}} \left(25D_{1}^{2} + 684D_{3} \cdot H_{B} + 342D_{4} \cdot H_{B} - 3200H_{B}^{2} + (438 - 45n_{7})D_{1} \cdot H_{B} + 18(32 - 5n_{7})D_{2} \cdot H_{B} - 120n_{7}D_{3} \cdot H_{B} - 60n_{7}D_{4} \cdot H_{B} - 800H_{B} \cdot S_{P} - 125\sigma(\hat{s}_{Q})^{2} - 490H_{B} \cdot \sigma(\hat{s}_{R}) - 275\sigma(\hat{s}_{Q}) \cdot \sigma(\hat{s}_{R}) \right).$$
(E.7)

10 dimensional

Finally a ten-dimensional basis is given by

$$H_{(V,12)}^{(2,2)} = H_{(V,13)}^{(2,2)} \setminus \{S_P \cdot \sigma(S_R), H_B \cdot \sigma(S_Q), D_2 \cdot D_4\}.$$
 (E.8)

• At $n_7 = 0$

$$S_P \cdot \sigma(S_R) \cong 2n_9 \left[(4+n_9) H_B^2 + H_B \cdot \sigma(\hat{s}_R) \right],$$

$$H_B \cdot \sigma(S_Q) \cong -2(2+n_9) H_B^2 - 2 \left(\frac{2+n_9}{4+n_9} \right) \sigma(\hat{s}_R) \cdot H_B$$

$$- \left(\frac{1}{4+n_9} \right) \sigma(\hat{s}_Q) \cdot \sigma(\hat{s}_R) - H_B \cdot S_P,$$

$$D_2 \cdot D_4 \cong 4D_4 \cdot H_B.$$
(E.9)

F CS-levels & chiralities for $\hat{X}_{\mathrm{SU}(5)}$ with $B = \mathbb{P}^3$

In this appendix we present the 3D Chern-Simons terms on both the M- and F-theory side and the chiralities, all of which parametrized by the parameters a_i of the general G_4 -flux.

F.1 Chern-Simons levels

F-theory loop induced CS terms

The one loop Chern-Simons levels read as follows. For the Cartan generators, we obtain Θ_{ij}^F as:

$$\begin{split} \Theta_{1,3}^{F} &= \Theta_{1,4}^{F} = 0, \\ \Theta_{1,1}^{F} &= -\frac{1}{2} \Big[6\chi \big(\mathbf{10}_{\left(\frac{1}{5},0\right)} \big) + 2\chi \big(\mathbf{5}_{\left(-\frac{2}{5},0\right)} \big) + 2\chi \big(\mathbf{5}_{\left(-\frac{2}{5},1\right)} \big) + 2\chi \big(\mathbf{5}_{\left(\frac{3}{5},0\right)} \big) + 2\chi \big(\mathbf{5}_{\left(\frac{3}{5},1\right)} \big) - 2\chi \big(\mathbf{5}_{\left(-\frac{2}{5},-1\right)} \big) \Big], \\ \Theta_{1,2}^{F} &= -\frac{1}{2} \Big[-3\chi \big(\mathbf{10}_{\left(\frac{1}{5},0\right)} \big) - \chi \big(\mathbf{5}_{\left(-\frac{2}{5},0\right)} \big) - \chi \big(\mathbf{5}_{\left(-\frac{2}{5},1\right)} \big) - \chi \big(\mathbf{5}_{\left(\frac{3}{5},0\right)} \big) - \chi \big(\mathbf{5}_{\left(\frac{3}{5},1\right)} \big) + \chi \big(\mathbf{5}_{\left(-\frac{2}{5},-1\right)} \big) \Big], \\ \Theta_{2,2}^{F} &= -\frac{1}{2} \Big[2\chi \big(\mathbf{10}_{\left(\frac{1}{5},0\right)} \big) + 2\chi \big(\mathbf{5}_{\left(-\frac{2}{5},1\right)} \big) + 2\chi \big(\mathbf{5}_{\left(\frac{3}{5},0\right)} \big) + 2\chi \big(\mathbf{5}_{\left(\frac{3}{5},1\right)} \big) - 2\chi \big(\mathbf{5}_{\left(-\frac{2}{5},-1\right)} \big) \Big], \end{split}$$

$$\begin{split} \Theta_{2,3}^{F} &= -\frac{1}{2} \Big[-\chi \big(\mathbf{10}_{(\frac{1}{5},0)} \big) + \chi \big(\mathbf{5}_{(-\frac{2}{5},0)} \big) - \chi \big(\mathbf{5}_{(-\frac{2}{5},1)} \big) - \chi \big(\mathbf{5}_{(\frac{3}{5},0)} \big) - \chi \big(\mathbf{5}_{(\frac{3}{5},1)} \big) + \chi \big(\mathbf{5}_{(-\frac{2}{5},-1)} \big) \Big], \\ \Theta_{3,3}^{F} &= -\frac{1}{2} \Big[2\chi \big(\mathbf{10}_{(\frac{1}{5},0)} \big) - 2\chi \big(\mathbf{5}_{(-\frac{2}{5},0)} \big) + 2\chi \big(\mathbf{5}_{(\frac{3}{5},1)} \big) - 2\chi \big(\mathbf{5}_{(-\frac{2}{5},-1)} \big) \Big], \\ \Theta_{3,4}^{F} &= -\frac{1}{2} \Big[-\chi \big(\mathbf{10}_{(\frac{1}{5},0)} \big) + \chi \big(\mathbf{5}_{(-\frac{2}{5},0)} \big) + \chi \big(\mathbf{5}_{(-\frac{2}{5},1)} \big) + \chi \big(\mathbf{5}_{(\frac{3}{5},0)} \big) - \chi \big(\mathbf{5}_{(\frac{3}{5},1)} \big) + \chi \big(\mathbf{5}_{(-\frac{2}{5},-1)} \big) \Big], \\ \Theta_{4,4}^{F} &= -\frac{1}{2} \Big[2\chi \big(\mathbf{10}_{(\frac{1}{5},0)} \big) - 2\chi \big(\mathbf{5}_{(-\frac{2}{5},0)} \big) - 2\chi \big(\mathbf{5}_{(-\frac{2}{5},1)} \big) - 2\chi \big(\mathbf{5}_{(\frac{3}{5},0)} \big) + 2\chi \big(\mathbf{5}_{(\frac{3}{5},1)} \big) - 2\chi \big(\mathbf{5}_{(-\frac{2}{5},-1)} \big) \Big]. \\ (F.1) \end{split}$$

The mixed Abelian-non-Abelian Θ^F_{im} read

$$\begin{split} \Theta_{i=1,m=1}^{F} &= \Theta_{i=4,m=1}^{F} = \Theta_{i=1,m=2}^{F} = \Theta_{i=2,m=2}^{F} = \Theta_{i=4,m=2}^{F} = 0, \\ \Theta_{i=2,m=1}^{F} &= -\frac{1}{2} \left(-\frac{4}{5} \chi \left(\mathbf{10}_{\left(\frac{1}{5},0\right)} \right) + \frac{4}{5} \chi \left(\mathbf{5}_{\left(-\frac{2}{5},0\right)} \right) \right), \\ \Theta_{i=3,m=1}^{F} &= -\frac{1}{2} \left(\frac{4}{5} \chi \left(\mathbf{5}_{\left(-\frac{2}{5},1\right)} \right) - \frac{6}{5} \chi \left(\mathbf{5}_{\left(3/5,0\right)} \right), \\ \Theta_{i=3,m=2}^{F} &= \chi \left(\mathbf{5}_{\left(-\frac{2}{5},1\right)} \right). \end{split}$$
(F.2)

The purely Abelian Θ^F_{mn} read

$$\Theta_{m=1,m=1}^{F} = -\frac{1}{2} \left(\frac{4}{25} \chi \left(\mathbf{10}_{\left(\frac{1}{5},0\right)} \right) - \frac{4}{25} \chi \left(\mathbf{5}_{\left(-\frac{2}{5},0\right)} \right) + \frac{4}{25} \chi \left(\mathbf{5}_{\left(-\frac{2}{5},1\right)} \right) \\ + \frac{9}{25} \chi \left(\mathbf{5}_{\left(\frac{3}{5},0\right)} \right) + \frac{9}{5} \chi \left(\mathbf{5}_{\left(\frac{3}{5},1\right)} \right) - \frac{4}{5} \chi \left(\mathbf{5}_{\left(-\frac{2}{5},-1\right)} \right) + \chi \left(\mathbf{1}_{\left(1,0\right)} \right) \\ + \chi \left(\mathbf{1}_{\left(1,1\right)} \right) + \chi \left(\mathbf{1}_{\left(-1,1\right)} \right) - 3\chi \left(\mathbf{1}_{\left(-1,-2\right)} \right) \right), \\ \Theta_{m=1,m=2}^{F} = -\frac{1}{2} \left(-\frac{2}{5} \chi \left(\mathbf{5}_{\left(-\frac{2}{5},1\right)} \right) + 3\chi \left(\mathbf{5} \right)_{\left(3/5,1\right)} - 2\chi \left(\mathbf{5}_{\left(-\frac{2}{5},-1\right)} \right) \\ + \chi \left(\mathbf{1}_{\left(1,1\right)} \right) - \chi \left(\mathbf{1}_{\left(-1,1\right)} \right) - 6\chi \left(\mathbf{1}_{\left(-1,-2\right)} \right) \right), \\ \Theta_{m=2,m=2}^{F} = -\frac{1}{4} \left(\chi \left(\mathbf{5}_{\left(-\frac{2}{5},1\right)} \right) + 5\chi \left(\mathbf{5}_{\left(\frac{3}{5},1\right)} \right) - 5\chi \left(\mathbf{5} \right)_{\left(-2/5,-1\right)} + \chi \left(\mathbf{1}_{\left(1,1\right)} \right) \\ + \chi \left(\mathbf{1}_{\left(0,1\right)} \right) + \chi \left(\mathbf{1}_{\left(-1,1\right)} \right) + 4\chi \left(\mathbf{1}_{\left(0,2\right)} \right) - 12\chi \left(\mathbf{1}_{\left(-1,-2\right)} \right) \right). \quad (F.3)$$

M-theory classical CS terms

On the M-theory side we obtain for Θ_{ij}^M :

$$\begin{split} \Theta^M_{1,3} &= \Theta^M_{1,4} = 0, \qquad \Theta^M_{2,4} = 2n_7 a_{12}, \\ \Theta^M_{(1,1)} &= \frac{1}{5} (6+n_9) a_3 + \frac{1}{2} (4+n_9) a_4 + n_9 (4-n_7+n_9) a_5 \\ &\quad + \frac{1}{25} [-10(4+n_9)(8+3n_9) + 3n_7 (14+9n_9)] a_6 \\ &\quad + \frac{1}{250} [-2968 + n_7 (56+111n_9) - n_9 (1148+115n_9)] a_7 \\ &\quad - \frac{1}{5} (-184+40n_7-23n_9+n_9^2) a_{13}, \end{split}$$

$$\Theta_{1,2}^{M} = \frac{1}{500} \left(-50(6+n_{9})a_{3} - 125(4+n_{9})a_{4} + 250(-4+n_{7}-n_{9})n_{9}a_{5} + 10[10(4+n_{9})(8+3n_{9}) - 3n_{7}(14+9n_{9})]a_{6} + (2968 - n_{7}(56+111n_{9}) + n_{9}(1148+115n_{9}))a_{7} + 50(-184+40n_{7} + (-23+n_{9})n_{9})a_{13} \right),$$
(F.4)

$$\Theta_{2,2}^{M} = \frac{1}{250} \bigg(25(21 - n_7 + n_9)a_3 + 125(4 + n_9)a_4 + 250n_9(4 - n_7 + n_9)a_5 + 5[-2n_7^2 - 5(4 + n_9)(41 + 11n_9) + n_7(122 + 57n_9)]a_6 + [-5524 + 7n_7^2 - 3n_9(273 + 40n_9) + n_7(277 + 113n_9)]a_7 + 25[64 + n_7(-24 + n_9) - (-13 + n_9)n_9]a_{13} \bigg),$$

$$\Theta_{2,3}^{M} = \frac{1}{500} \bigg(50(-17+n_7)a_3 - 125(4+n_9)a_4 + 250(-4+n_7-n_9)n_9a_5 + 10[2n_7^2 - 6n_7(14+5n_9) + 5(4+n_9)(27+5n_9)]a_6 + [8616 - 14n_7^2 - 5n_7(94+23n_9) + 5n_9(114+25n_9)]a_7 - 500n_7a_{12} + 50(-n_7(-4+n_9) + 2(6+n_9))a_{13} \bigg),$$

$$\Theta_{3,3}^{M} = \frac{1}{500} \bigg(25(50 - 5n_7 + 3n_9)a_3 - 125(n_7 - 2(4 + n_9))a_4 +500n_9(4 - n_7 + n_9)a_5 + 50(-14 + n_9)n_9a_{13} +5[-25n_7^2 - 5(4 + n_9)(90 + 23n_9) + 2n_7(210 + 73n_9)]a_6 +4[-3030 - n_9(483 + 65n_9) + n_7(310 + 66n_9)]a_7 \bigg),$$

$$\Theta_{3,4}^{M} = \frac{1}{500} \bigg(25(-16+3n_{7}-3n_{9})a_{3} + 125(-4+n_{7}-n_{9})a_{4} + 250(-4+n_{7}-n_{9})n_{9}a_{5} + 5[720+21(-12+n_{7})n_{7} + 440n_{9} - 86n_{7}n_{9} + 65n_{9}^{2}]a_{6} + [14(-55+n_{7})n_{7} - 149n_{7}n_{9} + 135n_{9}^{2} + 6(584+227n_{9})]a_{7} + 500n_{7}a_{12} + 50[-12+n_{7}(-4+n_{9}) - (-12+n_{9})n_{9}]a_{13} \bigg),$$

$$\Theta_{4,4}^{M} = \frac{1}{10} (16 - 3n_7 + 3n_9)a_3 + \frac{1}{2} (4 - n_7 + n_9)a_4 + n_9 (4 - n_7 + n_9)a_5 + \frac{1}{50} [-21n_7^2 - 5(4 + n_9)(36 + 13n_9) + n_7(252 + 86n_9)]a_6 + \frac{1}{250} [-14n_7^2 + n_7(770 + 149n_9) - 3(1168 + n_9(454 + 45n_9))]a_7 - 4n_7a_{12} + \frac{1}{5} [4(3 + n_7) - (12 + n_7)n_9 + n_9^2]a_{13}.$$
(F.5)

For the mixed CS-levels Θ^M_{im} we obtain:

$$\begin{split} \Theta^M_{i=1,m=1} &= \Theta^M_{i=1,m=2} = \Theta^M_{i=2,m=2} = 0, \\ \Theta^M_{i=4,m=1} &= \frac{2}{5} n_7 a_{12}, \qquad \Theta^M_{i=4,m=2} = 2n_7 a_{12}, \\ \Theta^M_{i=2,m=1} &= \frac{1}{625} \bigg(25(-15+n_7+n_9)a_3 + 5(-15+n_7+n_9)(2n_7-5(4+n_9))a_6 \\ &\quad -[-3360+7n_7^2 + (209-5n_9)n_9 + n_7(179+2n_9)]a_7 \\ &\quad -25(-4+n_9)(-5+n_7+n_9)a_{13} \bigg), \end{split}$$

$$\Theta_{i=3,m=1}^{M} = \frac{1}{2500} \left(-25(54+13n_{7}-9n_{9})a_{3}+250n_{7}a_{4} +5[54n_{7}^{2}-45(-6+n_{9})(4+n_{9})+n_{7}(52+3n_{9})]a_{6} +[-639n_{7}^{2}-6(-6+n_{9})(426+5n_{9})+n_{7}(1690+677n_{9})]a_{7} -500n_{7}a_{12}-50(18+n_{7}-3n_{9})(-4+n_{9})a_{13} \right),$$

$$\Theta_{i=4,m=1}^{M} = \frac{1}{500} n_7 \bigg(50a_3 - 125a_4 + 5(40 - 21n_7 + 15n_9)a_6 + (-158 + 111n_7 - 115n_9)a_7 - 500a_{12} - 50(-4 + n_9)a_{13} \bigg).$$
(F.6)

The CS-levels $\Theta^{M}_{\alpha m}$ for $D_{\alpha} = H_{B}$ read

$$\Theta^{M}_{\alpha,m=1} = -\frac{17}{5}a_3 + \frac{1}{2}(-4 + n_7 - n_9)a_4 + (-4 + n_7 - n_9)n_9a_5 + \left[n_7^2/2 + \frac{1}{5}(4 + n_9)(27 + 5n_9) - \frac{1}{50}n_7(208 + 75n_9)\right]a_6 + \frac{1}{250}\left[8616 - n_7(312 + 125n_7) + 5n_9(114 + 25n_9)\right]a_7 + \frac{2}{5}(6 + n_9)a_{13},$$

$$\Theta^{M}_{\alpha,m=2} = \frac{1}{2}(-4+n_{7}-n_{9})a_{3} - (4+n_{9})a_{4} + 2(-4+n_{7}-n_{9})n_{9}a_{5} + \left[24+16n_{9} + \frac{5}{2}n_{9}^{2} - \frac{1}{10}n_{7}(56+25n_{9})\right]a_{6} + \left[\frac{1}{2}n_{7}^{2} + \frac{1}{5}(4+n_{9})(27+5n_{9}) - \frac{1}{50}n_{7}(208+75n_{9})\right]a_{7} + (4+n_{9})a_{13}.$$
(F.7)

The purely Abelian CS-levels Θ^M_{mn} read

$$\begin{split} \Theta_{m=1,m=1}^{M} &= \frac{1}{12500} \bigg(25(8616 - n_7(312 + 125n_7) + 5n_9(114 + 25n_9)) a_3 \\ &\quad +125(1080 - 208n_7 + 25n_7^2 + 470n_9 - 75n_7n_9 + 50n_9^2) a_4 \\ &\quad +1250(-54 + 5n_7 - 10n_9)(-4 + n_7 - n_9)n_9a_5 \\ &\quad +5 \bigg[625n_7^3 - 5(4 + n_9)(14016 + 5n_9(854 + 125n_9)) \\ &\quad -4n_7^2(1411 + 625n_9) + 8n_7(6834 + 5n_9(893 + 125n_9)) \bigg] a_6 \\ &\quad +2 \bigg[-1090272 - 3125n_7^3 - 25n_9(6130 + n_9(1852 + 125n_9)) \\ &\quad +n_7^2(26777 + 3125n_9) + n_7(82008 + 5n_9(2028 + 625n_9)) \bigg] a_7 \\ &\quad +50 \bigg[-2968 + n_7(56 + 111n_9) - n_9(1148 + 115n_9) \bigg] a_{13} \bigg), \end{split}$$

$$\begin{split} \Theta_{m=1,m=2}^{M} &= \frac{1}{100} (25n_{7}^{2} + 10(4 + n_{9})(27 + 5n_{9}) - n_{7}(208 + 75n_{9}))a_{3} \\ &+ [12 + 8n_{9} + \frac{5}{4}n_{9}^{2} - \frac{1}{20}n_{7}(56 + 25n_{9})]a_{4} \\ &+ \frac{1}{2}(-12 + 3n_{7} - 5n_{9})(-4 + n_{7} - n_{9})n_{9}a_{5} \\ &+ \left[n_{7}^{2} \left(-\frac{454}{125} - \frac{9}{4}n_{9}\right) - \frac{1}{10}(4 + n_{9})(228 + n_{9}(157 + 30n_{9}))\right. \\ &+ \frac{3}{100}n_{7}(1040 + n_{9}(892 + 175n_{9}))\right]a_{6} \\ &+ \frac{1}{2500} \left[625n_{7}^{3} - 5(4 + n_{9})(14016 + 5n_{9}(854 + 125n_{9}))\right. \\ &+ 8n_{7}(6834 + 5n_{9}(893 + 125n_{9})) - 4n_{7}^{2}(1411 + 625n_{9})\right]a_{7} \\ &+ \frac{1}{25}(-10(4 + n_{9})(8 + 3n_{9}) + 3n_{7}(14 + 9n_{9}))a_{13}, \end{split}$$

$$\begin{split} \Theta_{m=2,m=2}^{M} &= \left[12 + 8n_9 + \frac{5}{4}n_9^2 - \frac{1}{20}n_7(56 + 25n_9) \right] a_3 \\ &+ \frac{1}{2} \left[-n_7(1 + 4n_9) + (4 + n_9)(12 + 5n_9) \right] a_4 \\ &+ (-12 + 2n_7 - 5n_9)(-4 + n_7 - n_9)n_9 a_5 \\ &+ \frac{1}{100} \left[-25(4 + n_9)(12 + 5n_9)^2 - 6n_7^2(7 + 50n_9) \\ &+ 5n_7(688 + n_9(842 + 185n_9)) \right] a_6 \\ &+ \frac{1}{500} \left[-50(4 + n_9)(12 + 5n_9)(27 + 5n_9) \\ &- n_7^2(1278 + 625n_9) + n_7(12164 + 5n_9(2542 + 375n_9)) \right] a_7 \\ &+ \left[-\frac{1}{2}(4 + n_9)(12 + 5n_9) + \frac{1}{5}n_7(4 + 9n_9) \right] a_{13}. \end{split}$$
(F.8)

F.2 Chiralities

Finally, we present the rest of the chiralities obtained by matching the M-/F-theory CSterms. The G_4 -flux parameter supported along the non-flat fiber is assumed to be zero by setting $a_{12} = 0$. The chiralities in terms of the a_i in the G_4 -flux are:

$$\chi(\mathbf{5}_{(-\frac{2}{5},0)}) = \frac{1}{250} \Big[-25(-13+n_7+n_9)a_3 - 5(-13+n_7+n_9)(2n_7-5(4+n_9))a_6 \\ + [-3092+7n_7^2 + (249-5n_9)n_9 + n_7(193+2n_9)]a_7 \\ + 25(-88+(-17+n_9)n_9 + n_7(16+n_9))a_{13} \Big],$$

$$\chi(\mathbf{5}_{(\frac{3}{5},0)}) = \frac{1}{500} \Big[-75(6+n_7-n_9)a_3 \\ +5 \Big[4n_7^2 + 11n_7(4+n_9) - 15(-6+n_9)(4+n_9) \Big] a_6 \\ + [-139n_7^2 - 2(-6+n_9)(426+5n_9) + n_7(458+149n_9)] a_7 \\ -50(6+n_7-n_9)(-4+n_9)a_{13} \Big],$$

$$\chi(\mathbf{5}_{\left(-\frac{2}{5},-1\right)}) = \frac{1}{500} \Big[50(4+n_9)a_3 + 125(4+n_9)a_4 + 250n_9(4-n_7+n_9)a_5 + 10(-10(4+n_9)(7+3n_9) + n_7(38+27n_9))a_6 + (3n_7(28+37n_9) - (4+n_9)(608+115n_9))a_7 - 50(-76+20n_7 - 15n_9 + n_9^2)a_{13} \Big],$$
(F.9)

For the chiralities of the singlets we obtain

$$\chi(\mathbf{1}_{(1,0)}) = \frac{1}{100} \Big[25[2n_7^2 - (-9 + n_9)(-6 + n_9) - n_7(15 + n_9)]a_3 \\ +50n_7(-4 + n_7 - n_9)n_9a_5 + 200n_7a_{13} \\ +5[5(-9 + n_9)(-6 + n_9)(4 + n_9) \\ -3n_7^2(12 + 5n_9) + n_7(276 + n_9(111 + 10n_9))]a_6 \\ +[50n_7^3 + 244(-9 + n_9)(-6 + n_9) \\ -3n_7^2(309 + 25n_9) + n_7(2694 + n_9(883 + 25n_9))]a_7 \Big],$$
(F.10)

$$\begin{split} \chi(\mathbf{1}_{(1,1)}) &= -\frac{1}{4} \Big[76 + n_7^2 + n_7(-17 + n_9) + 11n_9 - 2n_9^2 \Big] (a_3 + a_4) \\ &\quad -\frac{1}{2} (-4 + n_7 - n_9)n_9 (-19 + 2n_9)a_5 + \left[12 + \frac{1}{2} (2 + n_7 - n_9)n_9 \right] a_{13} \\ &\quad +\frac{1}{4} \Big[608 - n_7 (212 + (-25 + n_7)n_7) + 316n_9 \\ &\quad + (-56 + n_7)n_7n_9 + (17 + 6n_7)n_9^2 - 6n_9^3 \Big] a_6 \\ &\quad + \frac{1}{100} \Big[n_7^2 (262 - 25n_9) + n_7 (-4454 + 3n_9 (-91 + 25n_9)) \\ &\quad - (4 + n_9) (-4918 + n_9 (39 + 50n_9)) \Big] a_7, \end{split}$$

$$\chi(\mathbf{1}_{(0,1)}) = \frac{1}{4} \Big[n_7(-15+2n_7) + (4+n_9)(-19+2n_9) \Big] a_4 \\ -\frac{1}{2}(-4+n_7-n_9)n_9(-19+2n_9)a_5 \\ + \Big[76+49n_9 + \frac{1}{20}((-12+n_7)n_7(1+10n_7) \\ -n_7(79+10n_7)n_9 + 10(7+2n_7)n_9^2 - 20n_9^3) \Big] a_6 \\ + \frac{1}{100} \Big[-50n_7^3 + n_7^2(403+50n_9) + 2n_7(541+n_9(-488+25n_9)) \\ -(4+n_9)(-212+n_9(-449+50n_9)) \Big] a_7 \\ + \frac{1}{2} \Big[52-n_9(-9+n_7+n_9) \Big] a_{13}.$$
(F.11)

G Toric tuning of S_7 and S_9 for $B = \mathbb{P}^3$

In this appendix we explain how to construct the toric polytopes of $dP_2^B(\mathcal{S}_7, \mathcal{S}_9)$ with base $B = \mathbb{P}^3$ for all the points in the allowed region of figure 2. In the following we present a list of vertices, each of which realizing one possible choice of (n_7, n_9) .

In order to find this list of vertices we follow the algorithm presented in [47] to construct all reflexive convex polytopes for a given toric base and a given toric fiber. This algorithm uses the $GL(\mathbb{Z}, 5)$ symmetry to bring all the vertices of the polytope in the following form

variable	vertices						
z_0	1	1	1	p_1	p_2		
z_1	-1	0	0	0	0		
z_3	0	-1	0	0	0		
z_2	0	0	-1	0	0		
u	0	0	0	1	0		
v	0	0	0	0	1		
w	0	0	0	-1	-1		
e_1	0	0	0	0	-1		
e_2	0	0	0	1	1		

(G.1)

In the first four lines we see the polytope of the base, in the last five lines the polytope of the fiber. The degrees of freedom of the fibration are parametrized by the two integers (p_1, p_2) . The coordinates (p_1, p_2) are fixed by the requirement of convexity of the polytope (G.1). We obtain the following list of points (p_1, p_2) each of which giving rise to a reflexive and convex polytope (G.1), along with the corresponding values of (n_7, n_9) :

				-								
p_1	p_2	n_7	n_9]	p_1	p_2	n_7	n_9	p_1	p_2	n_7	n_9
-4	-4	8	4		0	-3	4	1	2	-1	2	1
-3	-4	7	3		0	-2	4	2	2	0	2	2
-3	-3	7	4		0	-1	4	3	2	1	2	3
-3	-2	7	5		0	0	4	4	2	2	2	4
-2	-4	6	2		0	1	4	5	2	3	2	5
-2	-3	6	3		0	2	4	6	2	4	2	6
-2	-2	6	4		0	3	4	7	3	-1	1	0
-2	-1	6	5		0	4	4	8	3	0	1	1
-2	0	6	6		1	-3	3	0	3	1	1	2
-1	-4	5	1		1	-2	3	1	3	2	1	3
-1	-3	5	2		1	-1	3	2	3	3	1	4
-1	-2	5	3		1	0	3	3	3	4	1	5
-1	-1	5	4		1	1	3	4	4	0	0	0
-1	0	5	5		1	2	3	5	4	1	0	1
-1	1	5	6		1	3	3	6	4	2	0	2
-1	2	5	7		1	4	3	7	4	3	0	3
0	-4	4	0		2	-2	2	0	4	4	0	4

Open Access. This article is distributed under the terms of the Creative Commons Attribution License (CC-BY 4.0), which permits any use, distribution and reproduction in any medium, provided the original author(s) and source are credited.

References

- R. Donagi and M. Wijnholt, Model Building with F-theory, Adv. Theor. Math. Phys. 15 (2011) 1237 [arXiv:0802.2969] [INSPIRE].
- [2] C. Beasley, J.J. Heckman and C. Vafa, GUTs and Exceptional Branes in F-theory I, JHEP 01 (2009) 058 [arXiv:0802.3391] [INSPIRE].
- [3] C. Beasley, J.J. Heckman and C. Vafa, GUTs and Exceptional Branes in F-theory II: Experimental Predictions, JHEP 01 (2009) 059 [arXiv:0806.0102] [INSPIRE].
- [4] R. Donagi and M. Wijnholt, Breaking GUT Groups in F-theory, Adv. Theor. Math. Phys. 15 (2011) 1523 [arXiv:0808.2223] [INSPIRE].
- [5] R. Blumenhagen, T.W. Grimm, B. Jurke and T. Weigand, *Global F-theory GUTs*, *Nucl. Phys.* B 829 (2010) 325 [arXiv:0908.1784] [INSPIRE].
- [6] J. Marsano, N. Saulina and S. Schäfer-Nameki, Compact F-theory GUTs with U(1) (PQ), JHEP 04 (2010) 095 [arXiv:0912.0272] [INSPIRE].
- [7] C.-M. Chen, J. Knapp, M. Kreuzer and C. Mayrhofer, *Global* SO(10) *F-theory GUTs*, *JHEP* 10 (2010) 057 [arXiv:1005.5735] [INSPIRE].
- [8] T.W. Grimm, S. Krause and T. Weigand, F-Theory GUT Vacua on Compact Calabi-Yau Fourfolds, JHEP 07 (2010) 037 [arXiv:0912.3524] [INSPIRE].

(G.2)

- J. Knapp and M. Kreuzer, Toric Methods in F-theory Model Building, Adv. High Energy Phys. 2011 (2011) 513436 [arXiv:1103.3358] [INSPIRE].
- [10] J.J. Heckman, Particle Physics Implications of F-theory, Ann. Rev. Nucl. Part. Sci. 60 (2010) 237 [arXiv:1001.0577] [INSPIRE].
- T. Weigand, Lectures on F-theory compactifications and model building, Class. Quant. Grav. 27 (2010) 214004 [arXiv:1009.3497] [INSPIRE].
- [12] A. Maharana and E. Palti, Models of Particle Physics from Type IIB String Theory and F-theory: A Review, Int. J. Mod. Phys. A 28 (2013) 1330005 [arXiv:1212.0555] [INSPIRE].
- [13] C. Vafa, Evidence for F-theory, Nucl. Phys. B 469 (1996) 403 [hep-th/9602022] [INSPIRE].
- [14] D.R. Morrison and C. Vafa, Compactifications of F-theory on Calabi-Yau threefolds. 1, Nucl. Phys. B 473 (1996) 74 [hep-th/9602114] [INSPIRE].
- [15] D.R. Morrison and C. Vafa, Compactifications of F-theory on Calabi-Yau threefolds. 2., Nucl. Phys. B 476 (1996) 437 [hep-th/9603161] [INSPIRE].
- [16] K. Kodaira, On compact analytic surfaces: Ii, Annals Math. 77 (1963) 563.
- [17] J. Tate, Algorithm for determining the type of a singular fiber in an elliptic pencil, in Modular functions of one variable IV (1975) 33–52.
- M. Bershadsky, K.A. Intriligator, S. Kachru, D.R. Morrison, V. Sadov et al., Geometric singularities and enhanced gauge symmetries, Nucl. Phys. B 481 (1996) 215
 [hep-th/9605200] [INSPIRE].
- [19] S.H. Katz and C. Vafa, Matter from geometry, Nucl. Phys. B 497 (1997) 146
 [hep-th/9606086] [INSPIRE].
- [20] M. Esole and S.-T. Yau, Small resolutions of SU(5)-models in F-theory, arXiv:1107.0733 [INSPIRE].
- [21] J. Marsano and S. Schäfer-Nameki, Yukawas, G-flux and Spectral Covers from Resolved Calabi-Yau's, JHEP 11 (2011) 098 [arXiv:1108.1794] [INSPIRE].
- [22] C. Lawrie and S. Schäfer-Nameki, The Tate Form on Steroids: Resolution and Higher Codimension Fibers, JHEP 04 (2013) 061 [arXiv:1212.2949] [INSPIRE].
- [23] A. Grassi, J. Halverson and J.L. Shaneson, Matter From Geometry Without Resolution, JHEP 10 (2013) 205 [arXiv:1306.1832] [INSPIRE].
- [24] S. Krause, C. Mayrhofer and T. Weigand, Gauge Fluxes in F-theory and Type IIB Orientifolds, JHEP 08 (2012) 119 [arXiv:1202.3138] [INSPIRE].
- [25] R. Donagi and M. Wijnholt, Higgs Bundles and UV Completion in F-theory, Commun. Math. Phys. 326 (2014) 287 [arXiv:0904.1218] [INSPIRE].
- [26] J. Marsano, N. Saulina and S. Schäfer-Nameki, Monodromies, Fluxes and Compact Three-Generation F-theory GUTs, JHEP 08 (2009) 046 [arXiv:0906.4672] [INSPIRE].
- [27] E. Dudas and E. Palti, Froggatt-Nielsen models from E₈ in F-theory GUTs, JHEP 01 (2010)
 127 [arXiv:0912.0853] [INSPIRE].
- [28] M. Cvetič, I. Garcia-Etxebarria and J. Halverson, Global F-theory Models: Instantons and Gauge Dynamics, JHEP 01 (2011) 073 [arXiv:1003.5337] [INSPIRE].
- [29] E. Dudas and E. Palti, On hypercharge flux and exotics in F-theory GUTs, JHEP 09 (2010)
 013 [arXiv:1007.1297] [INSPIRE].

- [30] M.J. Dolan, J. Marsano, N. Saulina and S. Schäfer-Nameki, F-theory GUTs with U(1) Symmetries: Generalities and Survey, Phys. Rev. D 84 (2011) 066008 [arXiv:1102.0290]
 [INSPIRE].
- [31] J. Marsano, H. Clemens, T. Pantev, S. Raby and H.-H. Tseng, A Global SU(5) F-theory model with Wilson line breaking, JHEP 01 (2013) 150 [arXiv:1206.6132] [INSPIRE].
- [32] A. Néron, Modeles minimaux des variétés abéliennes sur les corps locaux et globaux, Pub. Math. L'IHÉS 21 (1964) 5.
- [33] T. Shioda, On the Mordell-Weil lattices, Comment. Math. Univ. St. Paul 39 (1990) 211.
- [34] T. Shioda, Mordell-Weil lattices and Galois representation. I, Proc. Japan Acad. 65 A (1989) 268.
- [35] R. Wazir, Arithmetic on Elliptic Threefolds, math.NT/0112259
- [36] J.H. Silverman, The arithmetic of elliptic curves, vol. 106, Springer (2009).
- [37] P.S. Aspinwall and D.R. Morrison, Nonsimply connected gauge groups and rational points on elliptic curves, JHEP 07 (1998) 012 [hep-th/9805206] [INSPIRE].
- [38] P.S. Aspinwall, S.H. Katz and D.R. Morrison, *Lie groups, Calabi-Yau threefolds and F-theory, Adv. Theor. Math. Phys.* 4 (2000) 95 [hep-th/0002012] [INSPIRE].
- [39] T.W. Grimm and T. Weigand, On Abelian Gauge Symmetries and Proton Decay in Global F-theory GUTs, Phys. Rev. D 82 (2010) 086009 [arXiv:1006.0226] [INSPIRE].
- [40] A.P. Braun, A. Collinucci and R. Valandro, G-flux in F-theory and algebraic cycles, Nucl. Phys. B 856 (2012) 129 [arXiv:1107.5337] [INSPIRE].
- [41] S. Krause, C. Mayrhofer and T. Weigand, G₄ flux, chiral matter and singularity resolution in F-theory compactifications, Nucl. Phys. B 858 (2012) 1 [arXiv:1109.3454] [INSPIRE].
- [42] T.W. Grimm and H. Hayashi, F-theory fluxes, Chirality and Chern-Simons theories, JHEP 03 (2012) 027 [arXiv:1111.1232] [INSPIRE].
- [43] D.R. Morrison and D.S. Park, F-Theory and the Mordell-Weil Group of Elliptically-Fibered Calabi-Yau Threefolds, JHEP 10 (2012) 128 [arXiv:1208.2695] [INSPIRE].
- [44] M. Cvetič, T.W. Grimm and D. Klevers, Anomaly Cancellation And Abelian Gauge Symmetries In F-theory, JHEP 02 (2013) 101 [arXiv:1210.6034] [INSPIRE].
- [45] V. Braun, T.W. Grimm and J. Keitel, New Global F-theory GUTs with U(1) symmetries, JHEP 09 (2013) 154 [arXiv:1302.1854] [INSPIRE].
- [46] J. Borchmann, C. Mayrhofer, E. Palti and T. Weigand, *Elliptic fibrations for* SU(5) × U(1) × U(1) F-theory vacua, Phys. Rev. D 88 (2013) 046005 [arXiv:1303.5054] [INSPIRE].
- [47] V. Braun, T.W. Grimm and J. Keitel, Geometric Engineering in Toric F-theory and GUTs with U(1) Gauge Factors, JHEP 12 (2013) 069 [arXiv:1306.0577] [INSPIRE].
- [48] A. Klemm, M. Kreuzer, E. Riegler and E. Scheidegger, Topological string amplitudes, complete intersection Calabi-Yau spaces and threshold corrections, JHEP 05 (2005) 023 [hep-th/0410018] [INSPIRE].
- [49] M. Cvetič, D. Klevers, H. Piragua and P. Song, Elliptic Fibrations with Rank Three Mordell-Weil Group: F-theory with U(1) × U(1) x U(1) Gauge Symmetry, arXiv:1310.0463 [INSPIRE].

- [50] A. Grassi and V. Perduca, Weierstrass models of elliptic toric K3 hypersurfaces and symplectic cuts, arXiv:1201.0930 [INSPIRE].
- [51] M. Cvetič, D. Klevers and H. Piragua, F-Theory Compactifications with Multiple U(1)-Factors: Constructing Elliptic Fibrations with Rational Sections, JHEP 06 (2013) 067
 [arXiv:1303.6970] [INSPIRE].
- [52] G. Aldazabal, A. Font, L.E. Ibáñez and A.M. Uranga, New branches of string compactifications and their F-theory duals, Nucl. Phys. B 492 (1997) 119 [hep-th/9607121]
 [INSPIRE].
- [53] A. Klemm, P. Mayr and C. Vafa, BPS states of exceptional noncritical strings, hep-th/9607139 [INSPIRE].
- [54] M. Cvetič, D. Klevers and H. Piragua, F-Theory Compactifications with Multiple U(1)-Factors: Addendum, JHEP 12 (2013) 056 [arXiv:1307.6425] [INSPIRE].
- [55] T.W. Grimm, A. Kapfer and J. Keitel, Effective action of 6D F-theory with U(1) factors: Rational sections make Chern-Simons terms jump, JHEP 07 (2013) 115 [arXiv:1305.1929]
 [INSPIRE].
- [56] P. Mayr, Mirror symmetry, N = 1 superpotentials and tensionless strings on Calabi-Yau four folds, Nucl. Phys. B 494 (1997) 489 [hep-th/9610162] [INSPIRE].
- [57] A. Klemm, B. Lian, S.S. Roan and S.-T. Yau, Calabi-Yau fourfolds for M-theory and F-theory compactifications, Nucl. Phys. B 518 (1998) 515 [hep-th/9701023] [INSPIRE].
- [58] T.W. Grimm, T.-W. Ha, A. Klemm and D. Klevers, Computing Brane and Flux Superpotentials in F-theory Compactifications, JHEP 04 (2010) 015 [arXiv:0909.2025] [INSPIRE].
- [59] D.S. Park, Anomaly Equations and Intersection Theory, JHEP 01 (2012) 093
 [arXiv:1111.2351] [INSPIRE].
- [60] T.W. Grimm and R. Savelli, Gravitational Instantons and Fluxes from M/F-theory on Calabi-Yau fourfolds, Phys. Rev. D 85 (2012) 026003 [arXiv:1109.3191] [INSPIRE].
- [61] F. Bonetti and T.W. Grimm, Six-dimensional (1,0) effective action of F-theory via M-theory on Calabi-Yau threefolds, JHEP 05 (2012) 019 [arXiv:1112.1082] [INSPIRE].
- [62] D. Klevers, Holomorphic Couplings In Non-Perturbative String Compactifications, Fortsch. Phys. 60 (2012) 3 [arXiv:1106.6259] [INSPIRE].
- [63] B.R. Greene, D.R. Morrison and M.R. Plesser, Mirror manifolds in higher dimension, Commun. Math. Phys. 173 (1995) 559 [hep-th/9402119] [INSPIRE].
- [64] S. Sethi, C. Vafa and E. Witten, Constraints on low dimensional string compactifications, Nucl. Phys. B 480 (1996) 213 [hep-th/9606122] [INSPIRE].
- [65] K. Dasgupta, G. Rajesh and S. Sethi, *M theory, orientifolds and G flux, JHEP* 08 (1999)
 023 [hep-th/9908088] [INSPIRE].
- [66] C. Closset, T.T. Dumitrescu, G. Festuccia, Z. Komargodski and N. Seiberg, Contact Terms, Unitarity and F-Maximization in Three-Dimensional Superconformal Theories, JHEP 10 (2012) 053 [arXiv:1205.4142] [INSPIRE].
- [67] C. Closset, T.T. Dumitrescu, G. Festuccia, Z. Komargodski and N. Seiberg, Comments on Chern-Simons Contact Terms in Three Dimensions, JHEP 09 (2012) 091
 [arXiv:1206.5218] [INSPIRE].

- [68] F. Bonetti, T.W. Grimm and S. Hohenegger, A Kaluza-Klein inspired action for chiral p-forms and their anomalies, Phys. Lett. B 720 (2013) 424 [arXiv:1206.1600] [INSPIRE].
- [69] E. Witten, On flux quantization in M-theory and the effective action, J. Geom. Phys. 22 (1997) 1 [hep-th/9609122] [INSPIRE].
- [70] S. Gukov, C. Vafa and E. Witten, CFT's from Calabi-Yau four folds, Nucl. Phys. B 584 (2000) 69 [Erratum ibid. B 608 (2001) 477-478] [hep-th/9906070] [INSPIRE].
- [71] A. Bilal and S. Metzger, Anomaly cancellation in M-theory: A Critical review, Nucl. Phys. B
 675 (2003) 416 [hep-th/0307152] [INSPIRE].
- [72] K. Intriligator, H. Jockers, P. Mayr, D.R. Morrison and M.R. Plesser, Conifold Transitions in M-theory on Calabi-Yau Fourfolds with Background Fluxes, arXiv:1203.6662 [INSPIRE].
- [73] T.W. Grimm, The N = 1 effective action of F-theory compactifications, Nucl. Phys. B 845 (2011) 48 [arXiv:1008.4133] [INSPIRE].
- [74] T.W. Grimm, D. Klevers and M. Poretschkin, Fluxes and Warping for Gauge Couplings in F-theory, JHEP 01 (2013) 023 [arXiv:1202.0285] [INSPIRE].
- [75] A.J. Niemi and G.W. Semenoff, Axial Anomaly Induced Fermion Fractionization and Effective Gauge Theory Actions in Odd Dimensional Space-Times, Phys. Rev. Lett. 51 (1983) 2077 [INSPIRE].
- [76] A.N. Redlich, Parity Violation and Gauge Noninvariance of the Effective Gauge Field Action in Three-Dimensions, Phys. Rev. D 29 (1984) 2366 [INSPIRE].
- [77] O. Aharony, A. Hanany, K.A. Intriligator, N. Seiberg and M.J. Strassler, Aspects of N = 2 supersymmetric gauge theories in three-dimensions, Nucl. Phys. B 499 (1997) 67
 [hep-th/9703110] [INSPIRE].
- [78] H. Hayashi, C. Lawrie and S. Schäfer-Nameki, Phases, Flops and F-theory: SU(5) Gauge Theories, JHEP 10 (2013) 046 [arXiv:1304.1678] [INSPIRE].
- [79] E. Witten, Phase transitions in M-theory and F-theory, Nucl. Phys. B 471 (1996) 195
 [hep-th/9603150] [INSPIRE].
- [80] H. Hayashi, R. Tatar, Y. Toda, T. Watari and M. Yamazaki, New Aspects of Heterotic-F Theory Duality, Nucl. Phys. B 806 (2009) 224 [arXiv:0805.1057] [INSPIRE].
- [81] M.B. Green and J.H. Schwarz, Anomaly Cancellation in Supersymmetric D = 10 Gauge Theory and Superstring Theory, Phys. Lett. B 149 (1984) 117 [INSPIRE].
- [82] A. Sagnotti, A Note on the Green-Schwarz mechanism in open string theories, Phys. Lett. B 294 (1992) 196 [hep-th/9210127] [INSPIRE].
- [83] T.W. Grimm and W. Taylor, Structure in 6D and 4D N = 1 supergravity theories from *F*-theory, JHEP 10 (2012) 105 [arXiv:1204.3092] [INSPIRE].
- [84] P. Candelas, D.-E. Diaconescu, B. Florea, D.R. Morrison and G. Rajesh, Codimension three bundle singularities in F-theory, JHEP 06 (2002) 014 [hep-th/0009228] [INSPIRE].
- [85] M. Cvetic, A. Grassi, D. Klevers, and H. Piragua, work in progress.
- [86] D.R. Morrison and W. Taylor, Toric bases for 6D F-theory models, Fortsch. Phys. 60 (2012) 1187 [arXiv:1204.0283] [INSPIRE].
- [87] D.R. Morrison and W. Taylor, Classifying bases for 6D F-theory models, Central Eur. J. Phys. 10 (2012) 1072 [arXiv:1201.1943] [INSPIRE].